Studies in African Linguistics

Editor

David Odden

Associate Editors

G. Tucker Childs (Portland State University) Bruce Connell (York University) Katherine Demuth (Brown University) Omar Ka (University of Maryland, Baltimore) Ronald P. Schaefer (Southern Illinois University) John V. Singler (New York University)

Publisher

The Department of Linguistics and the Center for African Studies Ohio State University, Columbus, Ohio

Cover Design

Tina Messmer

Journal webpage: http://ling.osu.edu/sal

Studies in African Linguistics is published twice a year. Occasional supplements are published at irregular intervals and are available to current subscribers at reduced rates. It is printed and bound in the United States by Western Newspaper Publishing Co., Inc., Indianapolis, Indiana.

Contributions to the journal should be sent to: The Editor, *Studies in African Linguistics*, Department of Linguistics, 222 Oxley Hall, Ohio State University, Columbus, OH 43210. Contributors please refer to "Guidelines for Contributors" inside the back cover.

Correspondence concerning subscriptions or other business should be sent to: Managing Editor, *Studies in African Linguistics*, Department of Linguistics, 222 Oxley Hall, Ohio State University, Columbus, OH 43210. Subscription rates are the following:

	North America	Elsewhere
Individuals	\$25.00 per year	\$33.00 per year
Institutions:	\$50.00 per year	\$60.00 per year
Single issue:	one half the yearly i	ate
Supplements:	variable price deper	ding on supplement

Make checks payable to Studies in African Linguistics.

Volume 33, Number 2 • 2004

Copyright © 2004 by Studies in African Linguistics, Ohio State University ISSN 00390-3533

Studies in African Linguistics

Volume 33, Number 2

Fall 2004

TABLE OF CONTENTS

Torben Andersen	Jumjum phonology	133
Chris Collins	The absence of the linker in doubl constructions in N uu	l e object 163
Margaret Dunham	On the verbal system in Langi, a l language of Tanzania (F.33)	3antu 199
Eileen Kilpatrick & Leoma Gilley	The importance of word order in explaining tone patterns in Avoka	235 ya verbs
Publications received		261
Upcoming meetings on Afr	ican languages/linguistics	267
Notice on supplement		268
Guidelines for contributors	s i	nside back cover

Studies in African Linguistics Volume 33, Number 2, 2004

JUMJUM PHONOLOGY^{*}

Torben Andersen Aalborg University

This article describes the basic aspects of the phonology of Jumjum, a littleknown Western Nilotic language. The treatment includes syllable structure and word shapes, vowels and vowel harmony, consonants and consonant assimilation, and tones and tonal processes.

1. Introduction

Jumjum is a little-described Western Nilotic language spoken in the southern part of Blue Nile Province in Sudan. It belongs to the group of languages which Evans-Pritchard (1932) called Southern Burun, and which also includes Mabaan and Ulu. In Köhler's (1955) internal subgrouping of the Western Nilotic languages, Southern Burun and the neighbouring Northern Burun languages constitute one of three branches of Western Nilotic, the two other branches being Dinka-Nuer and the Luo languages. There are no previously published studies of Jumjum, except for two short word lists in Evans-Pritchard (1932: 28-31) and Bender (1971: 268).¹

^{*} The data on which this article is based were collected during two trips to Khartoum in 2002 and 2004. I wish to thank the Carlsberg Foundation for grants for the fieldwork and my Jumjum informants Juzuli Fadol Lago, Ramadan Makin Bashir and Yusif Juzuli for their assistence. I also wish to thank an anonymous referee and the editor for helpful comments on an earlier version of the article.

¹ However, at the 9th Nilo-Saharan Linguistics Colloquium, Institute of African and Asian Studies, University of Khartoum, 16-19 February 2004, Abeer Mohamed Ali Beshir presented a paper on Jumjum, "Morphophonemic alternations in Jumjum number formation", based on her recent master's thesis.

The present article describes the basic aspects of the phonology of Jumjum.² Section 2 deals with syllable structure and segmental word shapes. Section 3 describes the vowel system and accounts for the distribution of the vowels in terms of three types of harmony: [ATR] harmony, rounding harmony, and total harmony. Section 4 describes the inventory and distribution of consonants and accounts for certain constraints in terms of assimilation. Finally, section 5 describes the tonal system, accounts for the variability in the manifestation of the underlying tones in terms of three general tonal processes, and illustrates the functions of tones.

Morphophonological alternations abound in Jumjum as in many other Western Nilotic languages, but because of space limitations they are not dealt with in this article.

2. Syllable Structure and Segmental Word Shapes

As illustrated by means of monosyllabic words in (1), there are four syllable types in Jumjum.³ All syllables begin with a consonant, they are open or closed, their vowels are short or long, and there are no intrasyllabic consonant clusters.

(1)	a. CV	bí	(future tense particle)
	b. CVC	náŋ	'crocodile'
	c. CVV	wôo	'away, out'
	d. CVVC	jâan	'tree'

The length of a word is from one to at least four syllables, and all four syllable types also occur as the first and last syllables of disyllabic and trisyllabic words, as seen in (2)-(3), in which, respectively, the attested disyllabic and trisyllabic word shapes are listed and exemplified.⁴ Here points indicate syllable

² There seem to be two dialects of Jumjum. The dialect described here is said to be spoken by people of Watkey.

³ Tone marks indicate underlying tones, except that the tone marking of forms enclosed in brackets indicates surface tones, as in section 5 below. Lack of tone marks indicates that the underlying tones have not been identified.

⁴ Here and in similar displays, morphologically complex words are normally followed by a morphemic translation in parentheses after the English translation. The following abbreviations are used in morphemic translations: **1DUIN** first person dual inclusive; **1PL** first person plural; **1PLEX** first person plural exclusive; **1PLIN** first person plural inclusive; **1SG** first person singular; **2PL** second person plural; **2SG** second person singular; **3** third person; **3PL**

boundaries, while hyphens indicate morpheme boundaries. The middle syllable of trisyllabic words has been attested as *CV* or *CVC*, but not as *CVV* or *CVVC*.

(2)	a. CV.CV	wàŋ-ì	'my eye' (eye-1SG)
	b. CV.CVC	kàbál	'sheep' (sg.)
	c. CV.CVV	?ìkêe	'you' (2PL)
	d. CV.CVVC	?ìkôon	'we' (1PLEX)
	e. CVC.CV	kùm-mú	'egg' (egg-SG)
	f. CVC.CVC	kóttàŋ	'tortoise'
	g. CVC.CVV	?òl-l-âa	'you pushed me' (push-PST-2SG>1SG)
	h. CVV.CV	wìil-è	'its tail' (tail-3SG)
	i. CVV.CVC	tàatáŋ	'ash'
	j. CVVC.CV	?áat-tà	'chicken' (chicken-SG)
	k. CVVC.CVC	?íın-nàn	'intestine' (intestine-SG)
	l. CVVC.CVV	nùut-k-âa	'you tell me' (tell-BEN-2SG>1SG)
	m. CVVC.CVVC	nùut-k-ôon	'you tell us' (tell-BEN-2SG>1PLEX)
(3)	a. CV.CV.CVC	pát-àkáv	'bark' (bark-SG)
(-)	b. CV.CVC.CV	ρήτ-λησή	'barks' (bark-PL)
	c. CVC.CV.CV	mín-nèní	'child' (child-SG)
	d. CVC.CVC.CV	yàntáŋ-gá	'names' (name-PL)
	e. CVV.CV.CV	?λητ-λη-λ	'I came' (go:CP-PST-1SG)
	f. CVV.CV.CVC	?ʌʌt-śn-òn	'we came' (go:CP-PST-1PLEX)
	g. CVV.CVC.CV	péetàn-ní	'garbages' (garbage-PL)
	h. CVVC.CV.CV	méeŋká-ní	'spiders' (spider-PL)
	i. CVVC.CV.CVC	сл́лу-g-òn-ón	'we have it' (be.present-BEN-PST- 3PL>1PLEX)
	j. CVVC.CV.CVV	nìɪj-g-ád្-âa	'you cook for me' (cook-BEN-PRS- 2SG>1SG)
	k. CVVC.CV.CVVC	nìij-g-ód្-ôon	'you cook for us' (cook-BEN-PRS- 2SG>1PLEX)
	l. CVVC.CVC.CV	ກລ໌ກl-dຼລ້໗໗໌າ	'pythons' (python-PL)

third person plural; **3SG** third person singular; **AG** antigenitive; **ANTIP** antipassive; **ATT** attributive; **BEN** benefactive; **CF** centrifugal; **COM** comitative; **CP** centripetal; **FUT** future tense; **M** multiplicative; **NMLZ** nominalizer; **PL** plural; **PREP** preposition; **PRO** pronoun; **PRS** present tense; **PST** past tense; **PTCP** participle; **SG** singular or, sometimes, singulative; **X>Y** X as subject and Y as object in a cross-reference suffix.

A few examples of words with four syllables are given in (4).

(4)	a. CV.CV.CVC.CV	dùgùlúŋ-ŋí	'leopards' (leopard-PL)
	b. CVC.CV.CV.CVC	kúrkù-génén	'their nails' (nail:PL-3PL)
	c. CVVC.CV.CV.CV	ní1j-g-èn-èní	'I cooked for you' (cook-BEN-
			PST-1SG>2SG)

Morphologically, a word consists of a root and zero or more non-radical morphemes. In verbs and in many nouns, the root is the first CV(V)C part of the word. Non-radical morphemes are either derivational or inflectional, and they are expressed by means of suffixes and/or changes in the root, which concern vowel quality, vowel length, final consonant, and/or tone. There are no prefixes.

3. Vowels

3.1 Inventory and distribution of vowels. Jumjum has ten vowel qualities, which are divided into two symmetric sets distinguished by the feature [ATR] (Advanced Tongue Root), as shown in Table 1.

Table 1. Vowel qualities

		[-/	ATR]				[+,	ATR]	
high	I			U		i				u
mid		3		э			[e]		[0]	
low			а					Λ		

In monosyllabic words, however, there are only eight vowel qualities, the mid [+ATR] vowels [e] and [o] not occurring in such words. Thus, [e] and [o] only occur in words in which an adjacent syllable contains /i/ or /u/, i.e. a high [+ATR] vowel, as illustrated in (5). Hence, [e] and [o] must be analysed as allophones of the [-ATR] phonemes ϵ and β , respectively.

(5)	a. wéek-ìn	'we are crying'	(cry-1PLIN)
	b. bóoc-ìn	'we are afraid'	(fear-1PLIN)
	c. yèewùk	'dry season'	
	d. dóŋ-ú	'your neck'	(neck:SG-2SG) ⁵
	e. níin-è	'you are lying down'	(lie-2PL)
	f. níin-òn	'we are lying down'	(lie-1PLEX)
	g. túw-è	'you are dying'	(die-2PL)
	h. <u>t</u> úw-òn	'we are dying'	(die-1PLEX)

There is a binary length contrast for all eight phonemic vowel qualities, as illustrated by monosyllabic nouns in Table 2.

	/1/	wíŋ	'head'	/11/	?ìŋ	'woman'
	/æ/	mèn	'person'	/88/	déeŋ	'cow'
[-ATR]	/a/	kán	'leopard'	/aa/	pâan	'moon'
	/3/	dốŋ	'neck'	/၁၁/	?ôon	'man'
	/U/	bùŋ	'arm'	/บบ/	gûuŋ	'dog'
	/i/	jín	'stomach'	/ii/	díin	'bird'
[+ATR]	/λ/	ŋʎŋ	'back'	/лл/	рл̂лт	'mountain'
	/u/	tùk	'lip'	/uu/	kùun	'thorn'

 Table 2.
 Contrastive vowel qualities and lengths

3.2 [ATR] harmony. The allophony of $|\varepsilon|$ and $|\circ|$ results from a more general rule of vowel harmony to the effect that a high [+ATR] vowel spreads its [ATR] value to a non-low [-ATR] vowel of an adjacent syllable. The [+ATR] spread may occur in either direction: from root to suffix, or from suffix to root.⁶ The effect of this rule is illustrated in Table 3, which shows possessive inflection of some body part nouns. Before the 2nd person singular possessive suffix /-u/ 'your', the stem vowels /I, u, ε , \circ / become [i, u, e, o]. Moreover, the 1st person singular possessive suffix /-I/ is manifested as [i] after stems with /i/ or /u/, and the 3rd person singular possessive suffix /- ε / is manifested as [e] in the same environment.

⁵ Singular nouns with the morphemic gloss ":SG" exhibit replacement of a root-final plosive with a homorganic nasal, see Andersen (to appear).

⁶ A similar type of vowel harmony is found in the Northern Burun language Mayak, see Andersen (1999).

		Unpossessed	1SG	2SG	3SG	
	/I/	wín	wìn-ì	wíŋ-ù	wíŋ-è	'head'
	/ɛ/	lèk	lèk-ì	lek-u	lék-è	'teeth'
[-ATR]	/a/	wáŋ	wàŋ-ì	wáŋ-ù	wáŋ-è	'eye'
	/3/	dốŋ	dòŋ-ì	dóŋ-ú	dốŋ-ế	'neck'
	/U/	bùŋ	bùŋ-ì	bùŋ-ù	bừŋ-È	'arm'
	/i/	?ìntá	?ìnt-ì	?ìnt-ú	?int-e	'hand'
[+ATR]	/Λ/	kàn	kàn-ì	kàn-ù		'stomach'
	/u/	tùk	tùk-ì	tuk-u	<u>t</u> ùk-è	ʻlip'

Table 3.Possessive inflection of body part nouns

In the verbs forms (6b) and (7b), similarly, the root vowel harmonizes with the [+ATR] cross-reference suffix /-ín/. That the root vowel is lexically [-ATR] is evidenced by the forms (6a) and (7a), whose cross-reference suffix /-ón/ is [-ATR].

(6)	a. ?ʊ̀ʊl -ɔ́n	'he pushes us'
	push:M-3SG>1PLEX b. ?ùul -ín push:M-3SG>1PLIN	'he pushes us'
(7)	a. níic -k -òt -ón cook-BEN-PRS-3PL>1PLEX	'they cook for us'
	b. níic -k -ìt -ín cook-BEN-PRS-3PL>1PLIN	'they cook for us'

3.3 Rounding harmony. In suffix position, the vowel [Λ] varies with [U] and [U]. The quality [Λ] occurs after unrounded root vowels, while [U] and [U] occur after and in [ATR] harmony with rounded root vowels. Thus, the suffix vowel harmonizes with the root vowel for rounding, and if the root vowel is [+round], the suffix vowel also harmonizes with the root vowel for [ATR].⁷ In (8) this variation is shown in a -*CV* suffix that forms verbal nouns from verbal roots, and in (9) it is shown in the plural suffix -*gV*.

⁷ [ATR] harmony restricted to rounded vowels seems not to have been attested in other Western Nilotic languages.

(8)		Root	Verbal noun	
	/1/	lık-	lìŋ-ŋʎ	'to break'
	/ɛ/	уєєр-	yὲεm-mλ	'to throw'
	/a/	kac-	kàn-n⁄i	'to bite'
	/ɔ/	?əl-	?òl-lú	'to push'
	/U/	yuk-	yບັງ-໗ບ໌	'to kick'
	/i/	piit-	pìin_n∕	'to sow'
	/Λ/	nлk-	nλŋ-ŋʎ	'to kill'
	/u/	kuu <u>t</u> -	kùun-nú	'to blow'
(9)		Singular	Plural	
	/1/	kìın	kíud-gà	'guinea-fowl'
	/ɛ/	jâan	jèn-gà	'tree'
	/a/	làŋ-ŋán	làŋ-g⁄i	'fly'
	/ɔ/	bốn-nàn	bốy-gù	'skin'
	/U/	pôon	púuj-gú	'hare'
	/i/	yìm-mán	yím-gà	'blood'
	/Λ/	?ân	?ńʌr-gà	'house'
	/u/	kùm-mú	kúm-gù	'egg'

Another example is the 1st person singular subject suffix of intransitive verbs, -V, see (19) in section 4.2 below.

Suffixal [Λ] always varies with [U] and [u], and suffixal [U] always varies with [Λ] and [u], but suffixal [u] does not always vary with [Λ] and [U]. Thus, the 2nd person singular possessive suffix /-u/ 'your' is always realized as [u], and as shown in section 3.2 above, it spreads its [+ATR] feature to the root vowel. Hence, what is neutralized in suffix position is the contrast between underlying / Λ / and underlying /U/.

3.4 Total harmony. In some word-medial suffixes the quality of the vowel is copied from the following vowel. Such vowels may be analysed as underlyingly unspecified. One example is the past tense suffix -Vn, as in (10), where it occurs between the root *daam*- and suffixes which cross-reference a 3rd person plural subject and an object.

a.	dáam-án-à	'they looked at me'
b.	dáam-ín-í	'they looked at you (2SG)'
c.	dáam-ìn-í	'they looked at us (1DUIN)'
d.	daam-in-in	'they looked at us (1PLIN)'
e.	dáam-òn-ón	'they looked at us (1PLEX)'
f.	dáam-èn-é	'they looked at you (2PL)'
	a. b. c. d. e. f.	 a. dáam-án-à b. dáam-ín-ì c. dáam-ìn-í d. dáam-ìn-ín e. dáam-òn-ón f. dáam-èn-é

4. Consonants

4.1 Inventory and distribution of consonants. The inventory of consonants is shown in Table 4.

	bilabial	inter- dental	alveo- lar	palatal	velar	glottal
voiceless plosive	р	ţ	t	c	k	?
voiced plosive	b	d	d	j	g	
nasal	m	n	n	n	ŋ	
lateral			1			
trill			r			
glide	w			у		

Table 4.Inventory of consonants⁸

The occurrence of the consonants in three different positions of a word is exemplified in (11)-(12): initial position, final position of a monosyllable or disyllable, and intervocalic position. Some consonants are excluded from some of these positions: (i) The interdental nasal [n] does not occur word-initially, intervocalically or word-finally; (ii) the glottal stop [?] occurs only word-initially; (iii) the voiced plosives and the alveolar voiceless plosive [t] do not occur word-finally.

⁸ The transcription deviates from IPA in that [j] is a palatal plosive and [y] a palatal glide. Moreover, the diacritic [_] indicates interdental rather than dental place of articulation.

(11)	Wor	d-initial		Word-final in monosyllable					
	/p/	pîık	'water'	/p/	lèp	'tongues'			
	/t/	tôon	'cock'	/t/	?Èt	'faeces' (pl.)			
	/t/	τòη	'spear'	/t/	_				
	/c/	cáay	'ten'	/c/	dúuc	'five'			
	/ k /	kàaŋ	'snake'	/ k /	yàk	fish' (pl.)			
	/?/	?áak	'milk'	/?/	-				
	/b/	bừŋ	'arm'	/b/	_				
	/d/	déen	'cow'	/d/	_				
	/d/	déel	'goat'	/d/					
	/j/	jâan	'tree'	/j/	_				
	/g/	gèel	'lion'	/g/					
	/m/	mâan	'fire'	/m/	рл̂лт	'mountain'			
	/ŋ/	-		/ <u>n</u> /					
	/n/	náan	'calf'	/n/	mèn	'person'			
	/ɲ/	náŋ	'crocodile'	/ɲ/	wín	'head'			
	/ŋ/	ŋán	'back'	/ŋ/	?ìŋ	'woman'			
	/1/	lêen	'elephant'	/1/	wìil	'tail'			
	/r/	rúnit	'rainbow'	/ r /	bàr	'mosquito'			
	$ \mathbf{w} $	wáŋ	'eye'	/w/	dàw	'monkey species'			
	/y/	yàan	'bull'	/y/	рл́у	'path'			
(12)	Inter	vocalic		Word	d-final in	disvllable			
	/p/	ς ήλρ-ί	'she is cooking'	/p/	? ⁹				
		(cook:AN	TIP-3)	1					
	/t/	núut-ù	'they teach him'	/t/	yùd-ìt	'they are standing'			
	-	(teach-3P	rL>3)	-	(stand-M	?:3)			
	/t/	péetàt	'garbage'	/t/	_	,			
	/c/	cícàm	'knife'	/c/	уєє-п-іс	'your'			
					(PRO-SC	-2PL)			

/**k**/

/?/

/b/

/d/

náalók

_

'up'

'my lip' (lip-1SG)

'sheep' (sg.)

'I teach him'

/**k**/

/?/

/b/

/d/

<u>t</u>ùk-ì

kàbál

núud-ù

(teach-1SG>3)

⁹ /p/ and /w/ have not been attested word-finally in disyllables, but that is possibly due to scarcity of data.

/d/	déed-àŋ	'it is small'	/d/	_	
	(be.small-	3)			
/j/	kллj-т	'to come in'	/j/	_	
	(enter:CP-	-NMLZ)			
/g/	dúugùk	'six'	/g/	_	
/m/	dáam-à	'I look at it'	/m/	cícàm	'knife'
	(look.at-1	SG>3)			
/ <u>n</u> /	_		/ŋ/	_	
/n/	cùnúuk	'eight'	/n/	лл́лк-càr	i 'louse' (louse-SG)
/ɲ/	rúnìt	'rainbow'	/ɲ/	làŋ-ŋán	'fly' (fly-SG)
/ŋ/	wàŋ-ì	'my eye'	/ŋ/	líyáŋ	'feather'
	(eye-1SG))			
/1/	kéelòk	'one'	/1/	kàbál	'sheep' (sg.)
/ r /	ŋíɪràn	ʻrib'	/r/	tàbúr	'dust'
/w/	yèewùk	'dry season'	$ _{\mathbf{W}} $?	
/y/	wлy-э	'my father'	/y/	nùud-èy	'he is teaching you?
	(father-1S	G)		(teach-3SC	G>2SG)

As seen in section 2 above, there are heterosyllabic consonant clusters. Many, but far from all logically possible, clusters have been attested, and some of the non-attested ones are excluded for phonotactic reasons rather than being accidental gaps. Table 5 shows, by means of "x" in the respective cells, which C_1C_2 combinations have been attested out of all logically possible ones. C_1 is shown vertically, C_2 horizontally.

The following gaps seem to be systematic:

- Plosives with different voice values cannot be adjacent.
- The sequence plosive + nasal does not occur. This is a result of assimilation, see section 4.2 below.
- Liquids and glides do not occur as C₂ in any cluster, except in geminates. The reason probably is that no suffixes underlyingly begin with any of these consonants.
- There are no heterorganic nasal clusters, except /nm/. This is a result of assimilation, see section 4.2 below.
- Labial plosives do not occur as C₂, except in geminates. This is probably an accidental gap due to the fact there are no suffixes which underlyingly begin with a labial plosive.

• The alveolar plosives /t/ and /d/ do not occur as C₁, except that /t/ occurs in a geminate. The reason for this gap is a historical sound change, see section 4.3 below.

		C_2																		
		p	ţ	t	c	k	b	d	d	j	g	m	ŋ	n	ŋ	ŋ	1	r	w	у
	р	x			x	x														
	ţ		x		x	x														
	t			x																
	c		x		x	x														
	k		x		x	x														
	b						x	x		x							Ì			
	d							x		x	x									
	d																			
	j							x		x	x									
C_1	g									x	x									
	m					x		x		x	x	x								
	n		x									x	x							
	n			x		x			x		x			x						
	n							x		x	x				x					
	ŋ					x		x		x	x					x				
	1		x		x	x		x		x	x				x		x			
	r				x	x		x			x				x	x		x		
	W				x	x		x			x			x					x	
	у				x	x		x			x			x	x					x

Table 5. Heterosyllabic consonant clusters

The large number of different consonant clusters is a feature that distinguishes Jumjum from the more well-known Western Nilotic languages, so it seems worthwhile to exemplify these clusters, as done in (13)-(17).¹⁰

¹⁰In some cases, the function of some of the morphemes contained in the words has not been identified (with certainty), as indicated by question marks.

(13) Voiceless plosives as C_1

/pp/	yèɛp-p-é	'he threw it thither' (throw:CF-PST-3SG>3)
/pc/	yéep-c-éd-é	'he throws it thither' (throw-CF-PRS-3SG>3)
/pk/	kùp-k-ì	'they opened for him' (open-BEN-PST:3PL>3)
/ <u>t</u>t/	pìit-t-ń	'he sowed' (sow:ANTIP-PST-3)
/tc/	gìt-cán	'leaf' (leaf-SG)
/ <u>t</u> k/	gìt-ká	'leaves' (leaf-PL)
/tt/	kìt-tà	'star' (star-SG)
/ct/	ŋʌʌc-t̪ʌŋ	'to smell' (smell-NMLZ?)
/cc/	wiic-c-ń	'he swept' (sweep:ANTIP-PST-3)
/ck/	mác-kλ	'fires' (fire-PL)
/k <u>t</u> /	wллk-tʌŋ	'to breathe' (breathe-NMLZ?)
/kc/	yák-càn	'fish' (fish-SG)
/kk/	láak-kń	'urine' (urine-PL)

(14) Voiced plosives as C_1

tíib-bà	'shadow' (shadow-SG)
ŋùbdón	'heel'
yìb-j-ń	'you throw them thither' (throw:M?-CF-2SG>3)
kìd-d-é	'he drives them' (drive-M-3SG>3)
núud-jón	'you teach us' (teach-2SG>1PLEX)
túd-gú	'cocks' (cock-PL)
kúj-d-ù	'I pick up' (pick:ANTIP-?-1SG)
ŋáj-j-à	'I know him' (know-CF-1SG>3)
púuj-gú	'hares' (hare-PL)
yứug-j-ếḏ-ế	'he kicks them thither' (kick:M-CF-PRS-3SG>3)
tùg-gú	'lips' (lip-PL)
	tíib-bà ŋùbdón yìb-j-á kìd-d-é núud-jón túd-gú kúj-d-ù ŋáj-j-à púuj-gú yúug-j-éd-é tùg-gú

(15) Nasals as C_1 'blood' (blood-SG) /mm/ yìm-mán /md/ kíim-dànní 'hearts' (heart-PL) 'I look thither (?)' (look.at:ANTIP-CF?-PRS-1SG) /mj/ dánm-j-ád-á $/mk/ p_{\Lambda}m-k_{\Lambda}$ 'mountains' (mountain-PL) 'eggs' (egg-PL) /mg/ kúm-gù 'work' /nm/ kánmà ?ìn-tá 'hand' (hand-SG) /nt/ 'buttock' (buttock-SG) /nn/ wán-nà nán-tá 'place' (place-SG) /nt/

	/nd/	dúndúl	'bell'
	/nn/	tín-nÀ	'breast' (breast-SG)
	/nk/	?ìn-ká	'hands' (hand-PL)
	/ng/	mλ̀лn-gλ̀	'women' (women-PL)
	/nd/	?ìŋ-d-é	'he gives him (smth)' (give-?-3SG>3)
	/ŋj/	?íŋ-já	'you give me (smth)' (give-2SG>1SG)
	/ŋŋ/	ຼຼະບົກ-ກບ້	'granary' (granary-SG)
	/ng/	yúun-gù	'they rub it' (rub-3PL:3)
	/ŋḏ/	gòŋ-ḏ-έ	'he pulls it' (pull-M-3SG:3)
	/ŋj/	dèɛŋ-j-é	'he shoots it thither' (shoot-CF-3SG>3)
	/ŋk/	méeŋká	'spider'
	/ŋg/	yàŋ-gà	'meats' (meat-PL)
	/ŋŋ/	yλŋ-ŋ 	'meat' (meat-SG)
(16)	Liqu	ids as C_1	
	/l <u>t</u> /	bìl-t-í	'it is present' (be.present?-?-3)
	/l d /	ரட்டி பிற்ற	'pythons' (python-PL)
	/11/	bèel-lá	'cane' (cane-SG)
	/lc/	pól-càn	'cloud' (cloud-SG)
	/lj/	?òl-j-έ	'he pushes him thither' (push-CF-3SG>3)
	/lɲ/	dèl-nán	'penis' (penis-SG)
	/lk/	nál-kà	'rains' (rain-PL)
	/lg/	kòlg-ón	'fat' (fat-SG)
	/rd/	tór-d-ìn	'to pull' (pull-M-NMLZ)
	/ rr /	yòor-rú	'to see' (see-NMLZ)
	/rc/	tóor-c-éd-é	'he pulls it thither' (pull-CF-PRS-3SG>3)
	/ rɲ /	dúr-nù	'young man' (young.man-SG)
	/ rk /	kúrkòn	'nail'
	/rg/	díir-gá	'birds' (bird-PL)
	/ rŋ /	búrŋù	'cloth'
(17)	Glid	es as C_1	

(17) Glides as C₁
/ww/ yśwwè 'two'
/wd/ pòw-díní 'homes' (home-PL)
/wn/ ?ńw-nà 'bone' (bone-SG)
/wc/ kàw-cán 'seed' (seed-SG)
/wk/ kàw-ká 'seeds' (seed-PL)
/wg/ ?ńw-gà 'bones' (bone-PL)

/yd/	?áay- <u>d</u> -ìn	'to climb' (climb-M-NMLZ)
/yn/	kàay-n⁄i	'to pasture' (pasture-NMLZ)
/yc/	bλλy-còndέ	'which is long' (be.long-?)
/yŋ/	kèy-nán	'calabash' (calabash-SG)
/yy/	?λy-yλ	'salt' (salt-SG)
/yk/	mùy-kú	'oxen' (ox-PL)
/yg/	dèyg-ón	'porridge' (porridge-SG)

In verbs, all consonant clusters occur across morpheme boundaries. In nouns, by contrast, there are consonant clusters within morphemes, namely in some (presumably) monomorphemic disyllabic nouns, such as $\eta \dot{v} b d \dot{o} n$ 'heel', $d \dot{v} n d \dot{v} l$ 'bell' and $m \dot{\epsilon} \epsilon \eta k \dot{a}$ 'spider'.

4.2 Assimilation in /C-n/. Assimilations take place when a consonant-final stem is followed by a suffix which underlyingly begins with the alveolar nasal /n/. Table 6 shows the underlying representation (UR) and the surface representation (SR) of the consonant sequences in question. As can be observed, (i) the nasal spreads its nasality to a preceding plosive, which in return spreads its place of articulation to the nasal; (ii) stem-final nasals also spread their place of articulation to the suffixal nasal; (iii) the nasal totally assimilates to stem-final liquids; but (iv) no assimilation takes place if the stem-final consonant is a glide.

Table 6. Surface representation of underlying /C-n/

UR	p-n	t-n	c-n	k-n	m-n	n-n	n-n	ŋ-n	l-n	r-n	y-n	w-n
SR	m-m	n-n	ր-ր	ŋ-ŋ	m-m	n-n	ր-ր	ŋ-ŋ	1-1	r-r	y-n	w-n

One suffix that behaves in this way is -nV, which forms an abstract verbal noun from a verbal root, as illustrated in (18) for each of the twelve underlying consonants that can occur finally in a monosyllabic root. The root-final consonants surface more overtly in some verb forms, as also shown in (18). The verb forms cited consist of a transitive root and the suffix $/-\epsilon/$, which cross-references a third person singular subject and a third person object, except that in (18f), the verb consists of an intransitive root and the suffix $/-\lambda/$, which cross-references a first person singular subject. In these verb forms, root-final underlying plosives are manifested as voiced plosives. It might be suggested that the nasal of the nominalizing suffix is underlyingly unspecified for place of articulation. However, since it is alveolar after the labial glide /w/ and after the palatal glide /y/, it must be alveolar underlyingly.

(18)	Root-final	Root-3SG>3	Verbal noun	
	a. /p/	γέεb-έ	yὲɛm-mʎ	'throw at'
	b. /t/	kúu <u>d</u> -é	kùun_nú	'blow'
	c. /c/	káj-έ	kàn-n⁄i	'bite'
	d. /k/	líg-έ	lìŋ-ŋʎ	'break'
	e. /m/	?ám-έ	?àm-mλ	'eat'
	f. /n/	níin-à	níin-nÀ	'lie'
	g. /ɲ/	tún-é	ຼtບັກ-ກບ໌	'light'
	h. /ŋ/	gʻon-é	gວ້ວ໗-໗ບ໌	'pull'
	i. /l/	?51-έ	?òl-lú	'push'
	j. /r/	?ér-é	?èr-rx	'cut'
	k. /y/	káay-έ	kàay-n⁄i	'pasture'
	l. /w/	kóɔw-έ	kòow-nú	'take'

Another suffix that behaves in the way shown in Table 6 is the past tense suffix /-n/. In (19) it is exemplified after simple intransitive verb stems and before the 1st person singular subject suffix $-\lambda$, $-\dot{v}$, $-\dot{u}$. The underlying nature of the root-final consonants is revealed by the corresponding present tense forms, which have no tense suffix.

(19)		Root-final	Root-1SG	Root-PST-1SG	
	a.	/c/	bốoj-ừ	bວ໌ວ ກ- ກ-ບ້	'fear'
	b.	/k/	wéeg-à	wέεŋ-ŋ-λ	'cry'
	c.	/n/	níin-à	níin-n-à	'lie'
	d.	/1/	?໌ວວl-ù	?ɔ́əl-l-ù	'sing'
	e.	/y/	сл́лу-л̀	сл́лу-п-л̀	'sit'
	f.	/w/	túw-ù	túw-n-ù	'die'

A third suffix with the same behaviour is the plural suffix /-ni/, which is illustrated in (20). Unlike the two other suffixes, this suffix also occurs after a vowel-final stem, as in (20a).

(20)		Stem-final	Singular	Plural	
	a.	/a/	méeŋká	méeŋká-ní	'spider'
	b.	/ <u>t</u> /	rúnit	rúpin-ní	'rainbow'
	c.	/ k /	yílùk	yílùŋ-ŋí	'rainy season'
	d.	/n/	túupón	túupón-ní	'dust'
	e.	/ŋ/	kóttàŋ	kóttàŋ-ŋí	'tortoise'
	f.	/r/	kárkár	kárkár-rí	'stony place'

The direction of the assimilation with respect to manner of articulation seems to be determined by the sonority of the consonants involved. Given the sonority hierarchy glide > liquid > nasal > obstruent, a less sonorous consonant assimilates to a more sonorous one. Thus, obstruents assimilate to nasals, and nasals assimilate to liquids.

The above-mentioned suffixes are productive. Thus, the nominalizing suffix -nV and the past tense suffix -n are used after all simple verb stems, and the plural suffix -ni seems to be the default productive plural suffix.

4.3 The interdental nasal [n]. As seen in section 4.1 above, the interdental nasal [n] has a very restricted distribution, and its phonemic status might be questioned. Thus, [n] only occurs word-internally in heterosyllabic clusters. It regularly occurs in a geminate [nn], as in (21a-c), and before the voiceless interdental plosive /t/, as in (21d-e), and in both cases [n] could result from assimilation, see below. But [n] has also been attested before /m/, as in the noun in (21f), and here [n] must have phonemic status, unless [nm] is derived from underlying /tm/, for which, so far, no independent evidence has been found.

(21)	a.	bốn-nàn	'skin'	(skin-SG)
	b.	?έŋ-ŋλ	'faeces'	(faeces-SG)
	c.	màan-n⁄i	'to drink'	(drink-NMLZ)
	d.	yàntáŋ	'name'	
	e.	?ìn-tá	'hand'	(hand-SG)
	f.	kánmà	'work'	

Before an interdental plosive, $[\underline{n}]$ can readily be analysed as an allophone of alveolar /n/, since [n] does not occur in that position. Moreover, geminate $[\underline{n}\underline{n}]$ is often clearly the manifestation of an underlying root-final interdental /t/ plus an underlying alveolar /n/, as shown in section 4.2 above.

However, there are other instances of [nn] which cannot synchronically be analysed as underlying /tn/. Many disyllabic singular nouns end in a singular -*CV* suffix whose consonant is manifested almost like the underlying suffixal /n/ discussed in section 4.2 above. However, unlike the other suffixes, this suffix is not productive. Some examples are given in (22) together with the corresponding plural forms.

(22)		Root-final	Singular	Plural	
	a.	/p/	lέm-mλ	lèp	'tongue'
	b.	/ <u>t</u> /	?έ n -nλ	?è <u>t</u>	'faeces'
	c.	/c/	ຼt໌ບຸກ-ກບ້	túuc-kù	'granary'
	d.	/k/	lếŋ-ŋà	lèk	'tooth'
	e.	/m/	gím-mλ	gím-gà	'cheek'
	f.	/n/	tín-nÀ	tín-gà	'breast'
	g.	/ŋ/	yλ ŋ-ŋ λ	yàŋ-gà	'meat'
	h.	/1/	wíl-là	wìl-gá	'guest'
	i.	/ r /	tìn-n⁄i	tír	'witch-doctor
	j.	/y/	wày-n⁄i	wáy-gλ	'shoe'
	k.	/w/	?ńw-nλ	?ńw-gλ	'bone'

The plural forms are either suffixless or end in a plural suffix -kV or -gV, and their roots exhibit non-assimilated final consonants. The singular forms exhibit exactly the same consonant clusters as in section 4.2 above and under almost the same circumstances. It is therefore tempting to analyse the consonant of the singular suffix as an underlying /n/ here too. However, this analysis does not work for roots that end in /r/, as in (22i), since the resulting cluster here is /nn/ rather than /rr/, which is the synchronically productive result of /r/ plus underlying /n/. Hence, the consonant of the singular suffix cannot be analysed as being an underlying /n/ for all the allomorphs of the suffix. Instead, the surface forms of the singular nouns must be taken to be underlying forms, and therefore the geminate interdental /nn/ in (22b) cannot be derived synchronically from /tn/. Consequently, [n] is a separate phoneme here.

At an earlier stage in the historical development of Jumjum, however, the consonant of the singular suffix must have been an underlying /n/ throughout. Thus, the alternation between root-final /r/ in the plural and root-final /n/ in the singular in (22i) is historically explainable as follows: The /r/ in the plural form goes back to an alveolar plosive */t/, which changed to /r/ in root-final position, and before this change took place, */t/ assimilated to the suffixal /n/ in the singu-

lar form. The change */t/ > /r/ also explains why /t/ does not occur word-finally today. In Mabaan, root-final /t/ has been retained, the cognates of Jumjum *tir* 'witch-doctors' and *cir* 'spoons' being *tiet* and *ciet*, respectively, in that language.

Geminate $[\underline{nn}]$ is also found in some singular nouns which have the singulative suffix -<u>n</u>àn, as in (23). Again, the alternants of the roots that occur in the plural forms must be taken to be the original forms, and their final consonant is /n/ or /y/. Hence, the interdental $[\underline{n}]$ of the suffix cannot have received its place of articulation from the root. So again we are forced to ascribe phonemic status to $[\underline{n}]$. The sequence $[\underline{nm}]$ in (21f) above is also left unaccounted for, unless this conclusion is accepted.

(23)		Singular	Plural	
	a.	?íīnj-nan	?ìɪn-gÀ	'intestine'
	b.	bón_nàn	bốy-gừ	'skin'

5. Tones.

5.1 Tonal inventory and tonal processes. Jumjum is a tone language with two underlying tones: high (H) and low (L). They may combine to form a falling tone (\widehat{HL}) .

The identification of tones may start by examining nouns in the tonal frame shown in (24). This frame is a possessive construction in which the empty slot is filled by a possessee. The latter is followed by a pronoun which agrees in number with the possessee (sg. y en 'that of', pl. y ek 'those of') and which is modified by the following possessor.

(24)	a. [yè-n	?ôon]	b. [yὲ-k	?ôɔn]
	PRO-SG	:AG man:SG	PRO-PL:	AG man:SG
	'the man's	,	'the man's	,

When uttered in isolation, monosyllabic nouns fall into two tonal classes: words with a low tone [`] and words with a falling tone [^]. However, as revealed when the words are used in frame (24), the low-toned class actually represents two different underlying tonal classes. This is illustrated in Table 7, which shows three monosyllabic nouns in isolation and in frame (24) as well as their posited underlying tonal representation. In isolation, *yaan* 'bull' and *deen* 'cow' have a low tone, while *toon* 'cock' has a falling tone. In frame (24) the situation is

different: Here *yaan* still has a low tone, but $d \varepsilon \varepsilon \eta$ and $t \circ on$ now both have a high tone. The latter two nouns, however, are distinguished by the pitch of the following word $y \varepsilon n$, which has a low tone after $t \circ on$, as it also has after *yaan*, while it has a downstepped (⁴) high tone after $d \varepsilon \varepsilon \eta$. The underlying tones posited for the three nouns are low for /yàan/, high for /ds $\varepsilon \eta$ /, and falling for /toon/. Arguments for this analysis are given below.

 Table 7.
 Tonal classes of monosyllabic nouns

	Surface tones		Surface tones	Underlying tones		
	in isolati	lon	in frame (24)			
a.	[yàan]	[L]	[yàan yèn ?ôon]	/yàan/	/L/	'bull'
b.	[dèen]	[L]	[d̪ɛ̃ɛŋ ⁺yɛ́n ?う͡ɔn]	/déeŋ/	/H/	'cow'
c.	[t͡ɔɔn]	[HÎL]	[tốon yèn ?ôon]	/t͡ɔɔn/	/ HÎL/	'cock'

Table 8.	Tonal classes	of disyllabic nouns
----------	---------------	---------------------

	Surface tones		Surface tones	Underlyi	Underlying tones	
	in isolatio	on	in frame (24)			
a.	[t̪ùd̪gù]	[LL]	[túdgú ⁺yɛ́k ?う̂ɔn]	/túdgú/	/HH/	'cocks'
b.	[wíl¹lÂ]	[H⁺HÎL]	[wíl ¹ lá yèn ?ôon]	/wíllà/	/HL/	'guest'
c.	[dìŋŋ\]	[LL]	[d̪ìŋŋʎ ⁺yến ?うɔn]	/d̪ìŋŋʎ/	/LH/	'pestle'
d.	[bɛ̀ɛlgʌ̀]	[LL]	[bèɛlgà yèk ?ôɔn]	/bɛ̀ɛlgλ/	/LL/	'canes'
e.	[mɛ́rkʌ]	[HL]	[mếr⁴ká ⁴yếk ?ôɔn]	/mêrká/	/HÎLH/	'children'
f.	[kúurà]	[HL]	[kúurà yèn ?ôɔn]	/kûurà/	/HLL/	'ball'

The six disyllabic nouns shown in Table 8 fall into three tonal classes when uttered in isolation: (i) words in which both syllables have a low tone, (ii) words in which the first syllable has a high tone and the second syllable a low tone, and (iii) words in which the first syllable has a high tone and the second syllable a tone that falls from a high level lower than the preceding high, a downstepped falling tone. However, these three classes represent six different underlying tonal classes, as revealed when the six nouns are inserted into frame (24) in Table 8. Together with the following word, these six nouns here exhibit six different surface tone configurations. As the posited underlying tones indicate, the six disyllable nouns utilize all six logically possible combinations of H, L and HL in the first syllable and H and L in the second syllable. This analysis works in combination with the following three realization rules:

- (i) Utterance-final Lowering: In utterance-final (or prepausal) position a sequence of one or more high tones are changed to low tones.
- (ii) Downstepped Raising: After a high tone, a low tone is changed to a downstepped high tone in non-prepausal position and to a downstepped falling tone in prepausal position.
- (iii) Fall Simplification: In non-prepausal position a falling tone \widehat{HL} is simplified to H, and a following high tone is downstepped.

The application of these tone rules depends only on the tonal context and the boundaries of the utterance. Thus, they are independent of word boundaries and phrase boundaries. Each rule is illustrated in the following subsections.

In the examples given below, forms in brackets show the surface tones, while forms without brackets, in the next line, show the underlying tones and also the morpheme boundaries involved.

5.2 Utterance-final Lowering. In utterance-final (or prepausal) position a high tone is manifested phonetically as a low pitch which is identical to the low pitch of a low tone in the same position. Thus, in isolation, the high-toned monosyllabic noun $d\hat{\epsilon}\epsilon\eta$ 'cow' has the same (slightly falling) low pitch as the low-toned monosyllabic noun $y\hat{\alpha}an$ 'bull'. Similarly, the final high tone of the disyllabic noun $d\hat{\eta}\eta\dot{\lambda}$ 'pestle' is manifested as a pitch that is identical to (or even lower than) the pitch of the preceding low tone. Thus $d\hat{\eta}\eta\dot{\lambda}$ gets the same pitch pattern as the low-toned word $b\hat{\epsilon}\epsilon lg\lambda$ 'canes'. Hence, Utterance-final Lowering changes an underlying high tone to a low tone in utterance-final position, thereby neutralizing the contrast between High and Low in that position.

Utterance-final Lowering also applies to a sequence of high tones. Thus the HH word $\underline{t}\dot{v}dg\dot{v}$ 'cocks' has the same pitch sequence as the LL word $b\hat{\epsilon}\epsilon lg\lambda$ 'canes', namely a low pitch in the first syllable and an (extra) low pitch in the second syllable.

A high-toned word immediately preceding an utterance-final high-toned word also undergoes pitch lowering. Thus, while $d_{\Lambda\Lambda}m_{\Lambda}$ surfaces with high tones before an underlyingly low-toned word in (25a), it surfaces with low tones before an underlyingly high-toned word in (25b).

(25)	a.	[?ìkì dʌʌmʌ kìŋ]	6-	You are looking at a bee'
		?ìkì dánm -á	kìn	-
		2SG look.at:ANTIP-2SG	bee:SG	
	b.	[?ìkì dʌʌmʌ̀ lɛ̀n]		You are looking at an axe'
		?ìkì dáam -á	lén	
		2SG look.at:ANTIP-2SG	axe:SG	

It has not been examined whether there is any limit to the range of Utterance-final Lowering, but a sequence of lowered high tones may make up at least two words, as in (25b) and (26), and it may comprise at least five high tones, as in (26).

(26)	a.	[?ìkè pìitì mùkkìnì] ?ìkè píit -í múkkí-ní 3SG sow:ANTIP-3 maize -PL	'He is sowing maize'
	b.	[?ìkὲ mλʌjɔ̀ kùpkὲnὲ] ?ìkὲ mʎʌjɔ́ kúp -k -ɛ́n -ɛ́ 3SG Määjo open-BEN-PST-3SG>3	'He opened for Määjo'

Utterance-final Lowering may even take effect at the very beginning of an utterance consisting of more than one word, as in (27a). The pitch sequence of this utterance is indistinguishable from the pitch sequence of utterance (27b), which underlyingly begins with a low tone followed by two high tones.

(27)	a.	[lền ?ùlàŋ]	'The axe is black'
		lến ?úl -áŋ	
		axe:SG be.black-3	
	b.	[tòŋ ?ùlàŋ]	'The spear is black'
		tòŋ ?úl -áŋ	
		spear be.black-3	

5.3 Downstepped Raising. In non-prepausal position, a low tone is changed to a downstepped high tone ('H) after a high tone, as in (28)-(30). Like the other tonal processes, this raising is independent of word boundaries and phrase boundaries. In (28), for instance, the final high tone of $k\lambda tt\lambda$ causes the low tone of the fol-

lowing word $d\hat{u}k$ to be raised to ['dínk], the process thus crossing a word boundary. In (30), on the other hand, where the raising applies three times, it each time applies within the boundary of a disyllabic word.

(28)	[?ìkê kλttá ⁺díık yàakð] ?ìkê kλt -t -á dìık yáa -k - 3SG steal:ANTIP-PST-3 goat:PL PRO-PL	'He stole our goats' -5 -1PL
(29)	[?ámí ⁴ ká gữưŋ] ?ám-í kà gữưŋ eat -PST:3PL>3 PREP dog:SG	'It has been eaten by the dog' ¹¹
(30)	[?í⁴kến wáŋ⁺gến yúun⁺gû] ?íkền wáŋ-g -ền yúun-gù 3PL eye -PL-3PL rub -3PL>3	'They are rubbing their eyes'

There are two reasons why the raised low tone must be taken to be a high tone after the raising. First, the pitch of the raised tone is identical to the pitch of a following high tone in non-prepausal position, as in (30), and to the pitch of the high component of a following falling tone, as in (29). Second, the pitch of the raised tone is identical to the pitch of an underlying high tone that is downstepped after an underlying falling tone, see section 5.4 below.

A low tone following a raised low tone has a low pitch, so it does not itself undergo Downstepped Raising. Thus, in (31) the low tone of the syllable [ta] is left unaffected by the high tone of the preceding syllable [tbat], and in (32), similarly, the low tone of the syllable [$w\epsilon$] is left unaffected by the high tone of the preceding syllable [$tk\epsilon$].

(31)	[?áŋ +báttà ?à ?òɔtòn]				'But he is not sleeping'	
	?áŋ	bàttà	?à	?óot	-ón	
	and	not	be:3	sleep:	?-PTCP	

(32) [?ìkè bùggé 'ké tír'ké wènòk] 'His arms and legs swell'
?ìkè bùg-g-é kè tír -k -è wèn -òk
3SG arm-PL-3SG COM leg-PL-3SG swell-M:3

¹¹A postverbal third person singular subject noun (optionally preceded by the preposition $k\lambda$) is cross-referenced as 3PL in the verb.

In utterance-final position after a high tone, a low tone is not realized as a downstepped high tone, but as a downstepped falling tone. This is illustrated in (33)-(35) and can also be observed in (30) above. Thus, while *tinàl* surfaces as [tí⁺ŋál] in non-prepausal position in (33b), it surfaces as [tí⁺ŋâl] in prepausal position in (33a), here ending in a downstepped falling tone. The same variation is shown by *túwù* in (34), and in (35) the monosyllable *ing* surfaces as [⁺fing] prepausally.

(33)	 a. [tí⁺ŋâl] b. [tí⁺ŋál yèk ?ôɔn] tíŋàl yè -k ?ôɔn sand(PL)¹² PRO-PL:AG man:SG 	'sand' 'the man's sand'
(34)	a. [?ìkà bí tú⁺wû] ?ìkà bí túw-ù 1SG FUT die -1SG	'I shall die'
	b. [?ìkà bí tú⁺wú tìnֲnֲλk] ?ìkà bí túw-ù tìnֲnֲλk 1SG FUT die -1SG tomorrow	'I shall die tomorrow
(35)	[?ìkὲ tákí ⁴ ?îɪŋ] ?ìkὲ ták -í ?ìɪŋ 3SG want:ANTIP-3 woman:SG	'He wants a woman'

5.4 Fall Simplification. The behaviour of an underlying falling tone (\widehat{HL}) shows that it is a complex tone consisting of two simple tones, a high tone and a following low tone. A falling tone is manifested as a falling pitch in utterance-final position only, as in (36a) and (37a). In other positions the falling tone is manifested as a high level pitch, and a following high tone is downstepped. Thus, while $\widehat{ik}\widehat{con}$ surfaces with a falling tone in (36a), it surfaces with a final high tone in (36b), and the initial high tone of the following word, $\widehat{ikn}\widehat{on}$, surfaces as a downstepped high tone. The same type of variation obtains in (37). The downstep resulting from Fall Simplification must reflect the low component which is set afloat when the falling tone is simplified.

,

¹²Plural nouns with the morphemic gloss "(PL)" have no plural suffix.

(36)	a. [?ìkôɔn]	'we'
	b. [?ìkɔ́ɔn ⁺?ʎ⁺môn]	'We are eating'
	?ìkôon ?ʌ́m -òn	
	1PLEX eat:ANTIP-1PLEX	
(37)	a. [?ìkîin]	'we'
	b. [?ìkíin ⁺bí ?ʎ⁺mîn]	'We will eat'
	?ìkîin bí ?ám -ìn	

1PLIN FUT eat: ANTIP-1PLIN

hunger:SG PRO-SG:AG man:SG

The downstep effect also applies to a following underlyingly falling tone, as in (38), where *jâan* becomes ['jáan] after *750n*. By contrast, a low tone is left unaffected by a preceding underlyingly falling tone, as seen in (39), where $k \lambda p$ does not affect the realization of $y \epsilon n$.

(38)	[?ɔ́ən +jáan +?ʊ́ʊlɛ́ +t̪ʎɲɲàŋ]	'The man is pushing the tree now'
	?ôon jâan ?ứul -έ <u>t</u> àŋnáŋ	
	man:SG tree:SG push:M-3SG>3 now	
(39)	a. [kʌ̂ŋ]	'hunger'
	b. [kán yèn ?ôon]	'the man's hunger'
	kân về -n ?ôon	

There is no downstepped raising after a high tone that manifests a falling tone, as illustrated twice in (40). Here the initial low tone of biccan is not raised after [?5on], which is underlyingly ?5on, and similarly, the initial low tone of $tord\epsilon$ is not raised after [jáan], which is underlyingly jâan. This can be explained by assuming that the low component of the falling tone is not deleted, but left floating, as already assumed on independent grounds.

(40)	[?ɔ́ən bìiccàn jáan tòrdɛ]				'The ma	an is	pulling	the	root	of	the
					tree of	contir	nuously'				
	?ôon	bìic -càn	jâan	tòr -d -é							
	man:S	G root-SG:AG	tree:SC	6 pull-M-3SG>	3						

Jumjum Phonology

Syllables that are not word-final never have a falling pitch. Nevertheless, an underlying falling tone must be posited for a non-final syllable of some words in order to explain their pitch patterns and their effect on a following word.

Consider first the noun $merk_{\Lambda}$ 'children' in (42) and compare it with the noun will_{Λ} 'guest' in (41).

(41)	a.	[wíl¹lÂ]	'guest'
	b.	[wíl ¹ lá yèn ?ôɔn]	'the man's guest'
		wíl -là yè -n ?うon guest-SG PRO-SG:AG man:SG	
(42)	a. h	[mérkà] [mértká tvék ?ôon]	'children' 'the man's children'

?ôon

 $m\hat{\epsilon}r-k\hat{\lambda}y\hat{\epsilon}$ -k

child-PL PRO-PL:AG man:SG

Both $will_{\Lambda}$ and $merk_{\Lambda}$ surface with a high tone on their first syllable, but in utterance-final position the pitches of their last syllable are different from each other. While $[w\hat{1}^{\dagger}\hat{1}_{\Lambda}]$ in (41a) ends in downstepped falling tone, the second syllable of $[m \epsilon r k \lambda]$ in (42a) has a low tone. Before another word, both will λ and merk have the surface tone pattern [H⁴H], as seen in (41b) and (42b). However, while will_A does not trigger Downstepped Raising in the following word, $m \epsilon r k_A$ does. Hence, while will must be taken to end in a low tone underlyingly, merk must be taken to end in a high tone underlyingly, see section 5.3 above. So while the downstep in $[wíl^{+}l\Lambda]$ in (41b) must be caused by Downstepped Raising, the downstep in $[m \epsilon r^{+} k \Lambda]$ in (42b) cannot be the result of this rule. Instead, it must be taken to be the result of the other rule that produces a downstep, viz. Fall Simplification. The implication is that $merk_A$ has an underlyingly falling tone on its first syllable, so that its underlying form is $m\hat{\epsilon}rk\hat{\lambda}$. This analysis also explains the tonal difference between $will_{\lambda}$ and $m\hat{\epsilon}rk_{\lambda}$ in prepausal position: While the final downstepped falling tone in $[wil^{+}l\hat{\lambda}]$ is a result of Downstepped Raising triggered by the preceding high tone, the final low tone in [mérkà] is a result of Utterance-final Lowering after the floating low component of the preceding falling tone.

Consider next the noun kuura 'ball' in (43).

(43) a. [kúurà] 'ball'
b. [kúurà yèn ?ôon] 'the man's ball'
kûurà yè -n ?ôon ball PRO-SG:AG man:SG

This noun has the same pitch pattern as $m\hat{e}rk\hat{\lambda}$ 'children' in utterance-final position, see (43a) and (42a), respectively. In non-prepausal position, however, their pitch patterns differ from each other. Thus, while $m\hat{e}rk\hat{\lambda}$ in (42b) has the surface tones [H⁴H], *kuura* in (43b) has the surface tones [HL], and its underlying tone pattern must, therefore, be different from that of $m\hat{e}rk\hat{\lambda}$, which was analysed as \hat{HLH} above. The surface tone pattern [HL] of *kuura* cannot be its underlying tone pattern, since if it were, then its utterance-final surface tone pattern would be [H⁴HL], and its non-prepausal surface tone pattern would be [H⁴HL]. Instead, its behaviour can be explained by taking its underlying tone pattern to be \hat{HLL} , i.e. with a falling tone on the first syllable and a low tone on the final syllable. Thus, the falling tone is changed to a high tone by Fall Simplification, and the low tone set afloat by Fall Simplification prevents the low tone of the second syllable from undergoing Downstepped Raising.

5.5 Functions of tone. The function of tone is both lexical and grammatical, as exemplified below.

As seen in section 5.1 above, (singular or plural) nouns fall into tonal classes. There are also minimal pairs of nouns distinguished solely by tone, such as the monosyllabic (singular or plural) nouns in (44)-(45) and the disyllabic plural nouns in (46)-(48).

(44)	a. H	kúl	'wart-hog'
	b. L	kùl	'deep water'
(45)	a. HÎL	jînk	'words'
	b. L	jìnk	'hairs'
(46)	a. HH	kíygá	'streams'
	b. LH	kìygá	'feet'
(47)	a. HH	kл́лјgл́	'leopards'
	b. HL	kл́лјgλ̀	'hungers'

(48)	a. HL	?áŋkà	'suns'	
	b. LL	?àŋkà	'fishing nets'	

Unlike nouns, transitive verbal roots seem not to fall into tonal classes. Thus, they all behave in the same way tonally with respect to derivation and inflection. However, tone has a high functional load in person inflection of verbs with such roots. This is illustrated in Table 9, which shows the present tense forms with the simple transitive stem *daam*- 'look at'. Transitive verb stems take portmanteau suffixes which cross-reference both the subject and the object with respect to person and number, except that no number distinction is made for a third person object. Note also that a distinction is made between 1st person dual inclusive (1DUIN), 1st person plural inclusive (1PLIN) and 1st person plural exclusive (1PLEX). The vertical dimension of the table indicates the subject parameter (S), the horizontal dimension the object parameter (O).

$0 \rightarrow$	1SG	2SG	3	1DUIN	1PLIN	1PLEX	2PL
s↓							
1SG	-	dáam-èní	dáam-à		-	_	dâam-é
2SG	dáam-já	_	dáam-λ		_	dáam-jón	
3SG	dàam-à	dàam-èy	dáam-έ	dàam-í	dàam-ín	dàam-ón	dàam-é
1DUIN	-	_	dáam-ì	_		_	_
1PLIN	_	_	dáam-ìn	-	_	_	_
1PLEX	-	dáam-gì	dáam-òn	_	_		dâam-gé
2PL	dâam-à	_	dáam-è	_	_	dâam-òn	_
3PL	dáam-gà	dáam-gì	dáam-gì	dâam-gí	dâam-gín	dâam-gốn	dâam-gé

Table 9. Present tense forms with the simple transitive stemdaam- 'look at'

The paradigm in Table 9 includes six subsets of segmentally identical, but tonally distinct forms. Four of these subsets each contains two forms. For instance, the two forms with the segment sequence *daama* are distinguished tonally as spelt out in (49).

(49)	a.	LL	dàam-à look.at-3SG>1SG	'he is looking at me'
	b.	ĤLL	dâam-à look.at-2PL>1SG	'you are looking at me'
One s forms	subs s, as	set con shown	tains three forms, as shown n in (51).	in (50), and one subset contains four
(50)	a.	HL	dáam-òn look.at-1PLEX>3	'we are looking at him'
	b.	LH	dàam-ón look.at-3SG>1PLEX	'he is looking at us'
	c.	HLL	dâam-òn look.at-2PL>1PLEX	'you are looking at us'
(51)	a.	HH	dáam-έ look.at-3SG>3	'he is looking at him'
	b.	HL	dáam-è look.at-2PL>3	'you are looking at him'
	c.	HLH	dâam-é look.at-1SG≥2PL	'I am looking at you'
	d.	LH	dàam-é look.at-3SG>2PL	'he is looking at you'

Another example of the use of tone in the morphology is its role in case inflection of nouns. As seen in section 5.1 above, there are few constraints on the tones of nouns in their lexical form, which may be called the "absolutive" case. But when a noun is modified by a following possessor noun, as in (52b), all of its syllables have low tones, whatever its lexical tone pattern. Thus, the head of the possessive construction, the possessee, has a morphologically marked form, which I call the "antigenitive" case, while the possessor has the absolutive case. This possessive construction is an alternative to the construction in (52a), which was used as a tonal frame in section 5.1 above. Here the possessee has the absolutive case and is followed by a coreferential pronoun which is followed by the possessor in the absolutive case. The intervening pronoun agrees in number with the possessee (sg. y in, pl. y ik) and has a form which seems to be the antigenitive case.

(52)	a.	[bừrúŋgú ⁺ yék ?ôɔn] bừrúŋ-gú yὲ -k ?ôɔn cloth -PL PRO-PL:AG man:SG	'the man's clothes'
	b.	[bừrùŋ-gừ ?ôɔn] bừrừŋ-gừ ?ôɔn cloth -PL:AG man:SG	'the man's clothes'

The formal difference between the absolutive case and the antigenitive case is exemplified in (53) with some monosyllabic, disyllabic and trisyllabic nouns with various tone patterns in the absolutive. As can be observed, all tones are low in the antigenitive case, whereby the lexical tonal contrasts are neutralized.

(53)	Absolu	tive	Antiger	nitive		
	a. H	déeŋ	L	dèeŋ	'cow'	
	b. HL	yîın	L	yìm	'giraffe'	(giraffe:SG)
	c. L	kùun	L	kùun	'thorn'	(thorn:SG)
	d. HH	díir-gń	LL	dìir-gà	'birds'	(bird-PL)
	e. HL	cícàm	LL	cìcàm	'knife'	
	f. LH	càw-ná	LL	càw-nà	'arrow'	(arrow-SG)
	g. LL	mλʌn-gλ	LL	mλ̀лn-gλ̀	'women'	(women-PL)
	h. HHH	líyáŋ-gá	LLL	lìyàŋ-gà	'feathers'	(feather-PL)
	i. HLH	mín-nèní	LLL	mìn-nènì	'child'	(child-SG)
	j. LHH	bừrúŋ-gú	LLL	bùrùŋ-gù	'clothes'	(cloth-PL)

REFERENCES

- Andersen, Torben. 1999. "Vowel harmony and vowel alternation in Mayak (Western Nilotic)". *Studies in African Linguistics* 28: 1-29.
- Andersen, Torben. (to appear). "Layers of number inflection in Mabaan (Western Nilotic)". Journal of African Languages and Linguistics.
- Bender, M. Lionel. 1971. "The languages of Ethiopia". Anthropological Linguistics 13, No. 5.
- Evans-Pritchard, E. E. 1932. "Ethnological observations in Dar Fung". Sudan Notes and Records 15: 1-61.
- Köhler, Oswin. 1955. "Geschichte der Erforschung der nilotischen Sprachen". *Afrika und Übersee*. Beiheft 28.

Department of Communication Aalborg University Fibigerstraede 10, rum 3-10 9220 Aalborg Oest torben@hum.aau.dk

[Received September 16, 2004 accepted November 27, 2004] Studies in African Linguistics Volume 33, Number 2, 2004

THE ABSENCE OF THE LINKER IN DOUBLE OBJECT CONSTRUCTIONS IN N|UU¹

Chris Collins New York University

The linker in N|uu appears before various types of nominal expressions, but not before the second object in a double object construction. Linkers in Khoisan languages such as \pm Hoan and Ju|'hoansi do appear in this position. I will show that this property of the linker in N|uu is related to the fact that N|uu has a dative Case marker *-a* which appears after the first object of a DOC, whereas \pm Hoan and Ju|'hoansi do not.

1. Introduction²

The linker in N|uu, a southern Khoisan language of the !Ui branch, appears before various different types of nominal expressions, such as the locative in (1a) and the direct object of a causativized verb in (1b), but not before the second object in a double object construction, as show in (2a,b) (Decl stands for declarative, and Lk stands for linker).

¹ This field work for this paper was supported by the National Science Foundation under grant BCS-0236735 during the summers of 2004 and 2005. The data in this paper were provided by Katrina Esau, Hanna Koper, Griet Seekoei, Anna Kassie, and Andries Oleyn. I would like to thank Levi Namaseb who helped me transcribe the data. For helpful comments, I would like to thank Johanna Brugman, Daniele Goddard, Ken Hiraiwa, Richard Kayne, David Odden, two reviewers, and audiences at Paris 7 and Paris 8.

² I have recorded every sentence in this paper spoken by at least two speakers and in many cases by all five. In total, there are around 10 remaining speakers of N|uu in SA, and perhaps a few others in southern Botswana.

(1)	a.	ku -a	si	hoo	ku	aŋki	ŋ	g!ari
		3sg Decl	Fut	find	3sg	father	Lk	Upington
		'He will	find hi	is fathe	er in U	pingto	n'	
	b.	n -a 1sgDecl 'I made]	kx'u ^{make} him hu	q'õa- hunt-A Int a du	a ^{Asp³ liker'}	ku 3sg	ŋ Lk	‡qhee duiker
(2)	a.	Griet Griet 'Griet	ke Decl will g	si Fut jive hir	?ãa give n the c	ku-a 3sg-Da lonkey	doŋ at donl	ki-si key-Sg
	b.	ku -a 3sgDecl 'He will	si Fut show]	kajan show Nlangu	na ŋ a N a Isi the	ngusi- ngusi-D meat'	a Oo Dat mea	e at

The fact that the linker does not appear before the second object of a DOC is surprising because linkers in other Khoisan languages such as \pm Hoan and Jul'hoansi do appear in this position. I will show that this difference in the syntax of the linker is related to the fact that N|uu has a dative Case marker -*a* which appears after the first object of a DOC, whereas \pm Hoan and Jul'hoansi do not.

In section 2, I will discuss structures of the form [V Lk X], where the linker appears between the verb and one post-verbal constituent. In section 3, I will discuss how-constructions, which also involve a linker. In section 4, I will propose that the linker is a Case checking head. In section 5, I will discuss structures of the form [V DP Lk X] where the linker appears between two post-verbal constituents. In section 6, I will discuss causatives. In section 7, I will explain the absence of the linker in DOCs. In section 8, I will give a table summarizing the properties of the linker and how they vary amongst the Khoisan languages.

All the data in this paper are from N|uu unless otherwise indicated. All the data from \pm Hoan are from Collins (2003). All the data from Ju|'hoansi are from Collins (2003) and Dickens (1992). The N|uu sentences are written in the IPA alphabet.

 $^{^{3}}$ I gloss the final -*a* (low tone) that appears on verbs as Asp. It is used in the present tense with a subset of stative verbs (e.g., want, know). On non-stative verbs it indicates past tense. We have not yet systematically investigated the past imperfective (progressive, habitual) nor the past tense of stative verbs.

2. V Lk X

In this section I will discuss some cases where the linker appears between the verb and a post-verbal constituent. Consider first the case where a locative expression follows an intransitive verb. An intransitive verb is a verb without a direct object. In this case, a linker must precede the locative.⁴

- (3) a. n !ũu ke l?aa xa η g!ari Decl Past die 1sg grandfather Lk Upington 'My grandfather died in Upington' b. |ã?n-si ke ||?ŋ-a 100 η snake-Sg Decl go.out-Asp Lk hole 'The snake went out of the hole' lahõ[°] n c. ku -a llã?ẽ ku $\eta \|\eta\|$ dance Lk house in 3sg Decl 3sg 'He is dancing in his house'
 - d. ku -a +qheke ŋ ku ŋ||ŋ ||ã?ẽ
 3sgDecl sing Lk 3sg house in
 'He is singing in his house'

Similar sentences involving a locative following an intransitive verb can be found in \pm Hoan and Ju|'hoansi. Consider the following data from Collins (2003) (Trans stands for transitivity suffix).

(4) tsi a-kyxai ki !oa na (‡Hoan)
3pl Prog-dance Lk house in
'They are dancing in the house'

ku-a si |qhõ[°] !?ui 3sg-Decl Fut dance !?ui 'He will dance !?ui'

⁴ Note that I am not claiming that the verbs in (3) are always intransitive. For example, the verb 'dance' can be used transitively, and therefore the verb 'dance' has a transitive and an intransitive use.
(5) Lena koh djxani-a tju n!ang (Ju|'hoansi) Lena Past dance-Trans house in 'Lena danced in the house'

In (4), ki is the linker in \pm Hoan, and it appears preceding the locative. In (5), the transitivity suffix in Jul'hoansi must be used on the verb preceding the locative. In Collins (2003), I analyze the transitivity suffix in Jul'hoansi as a form of the linker (the other form being ko). Therefore, in both \pm Hoan and Jul'hoansi a linker must appear before a locative that follows an intransitive verb.

The following examples show that the linker also precedes various kinds of non-locative expressions. These examples show that the linker does not have a specific locative meaning (such as 'in' or 'at'). Furthermore, example (6a) shows that *!haeka* 'tomorrow' must be a DP, since the linker only appears before DPs (see section 3).

- (6) a. ku -a si |qhõ[°] ŋ !haeka
 3sgDecl Fut dance Lk tomorrow
 'He will dance tomorrow'
 - b. g|a ‡ao-a a ‡hoa ŋ ku mari (!?ama)
 2sg want-Asp 2sg talk Lk 3sg money about
 'Do you want to talk about his money?'

Temporal expressions, as opposed to locatives, can also appear in a preverbal position, in which case they are not preceded by a linker.⁵

(7)	a.	ku -a	si	!haeka	qhõ [°]
		3sg Decl	Fut	tomorrow	dance
		'He will	dance	tomorrow'	

⁵ Westphal p. 46 has other examples of preverbal temporal adverbs with no accompanying linker (see Güldemann's 2004a publication of Westphal's manuscript). In terms of the analysis presented in section 3, preverbal temporal adverbs have no Case feature, and post-verbal temporal adverbs have a Case feature (checked by the linker). I do not have independent evidence for this analysis of temporal adverbs.

b.	*ku	-a	si	g!ari	qhõ [°]
	3sg	Decl	Fut	Upington	dance
	'He w	vill danc	e in Up	oington'	

A linker does not appear between a transitive verb and its direct object.

- (8) a. n -a ‡hãu-a !qhaa lsgDecl pour-Asp water 'I poured water'
 - b. n -a si ||?au -|?e ||aβa 1sgDecl Fut bury -put.in bone 'I will bury the bone'
 - c. ku -a si ||x'oo Ooo-ke 3sgDecl Fut chop wood-pl 'He will chop wood'
 - d. ku -a si fioo ku aŋki
 3sg Decl Fut find 3sg father
 'He will find his father'
 - e. kunisi ke si ‡am !ao wagon Decl Fut hit rock 'The wagon will hit the rock'

Inserting a linker between a transitive verb and its direct object results in ungrammaticality.

(9) (*****ŋ) aŋki ku -a si fioo ku 3sg Decl Fut find 3sg father Lk 'He will find his father'

A similar constraint holds for \pm Hoan and Ju|'hoansi (recall, I am assuming that the transitivity suffix -*a* in Ju|'hoansi is a form of the linker, as discussed in Collins 2003). The following sentences are from Collins (2003).

(10)	*koloi car	i g∥on-a hit-Perf	ki Lk	Jefo Jeff	(‡Hoan)
(11)	*uto car	dchuun-a hit-Trans	Kae	ece ce	(Ju 'hoansi)

There are some verbs that take a locative complement without a linker.

- (12) a. ku -a si ||?ae g!ari
 3sgDecl Fut go Upington
 'He will go to Upington'
 - b. ŋ+hau ke |?ee !oo dungbeetle Decl enter hole
 'The dungbeetle is entering the hole'
 - c. ku -a ŋ||a a ŋ||ŋ ||ã?ẽ
 3sgDecl be.loc 2sg house in
 'He is in your house'
 - d. ca[°]bakusi ke ŋ||a g!ari
 ca[°]bakusi Decl be.loc g!ari
 [°]Cabakusi is in Upington'

In the above example (12c), y || a is the form of the verb 'be' that is used to express location. The simplest interpretation of these verbs is that they are transitive taking a locative direct object, just as 'enter' in English can be transitive ('John entered the room'). In terms of Case theory (see section 3), I assume that the verb checks the Case feature of its locative complement in these examples.

When a locative is extracted, the linker cannot be followed by a gap, instead the linker must be replaced by $y||\tilde{a}$ 'be.loc'⁶ (Q stands for question, Rel stands for relative complementizer).

cui !?ama a \neq ao-a a \neq hoa ŋ||ã what about 2sg want-Asp 2sg talk be.loc 'What do you want to talk about?'

168

⁶ \pm Hoan and Ju|'hoansi do not have this strategy for avoiding a linker followed by a gap. I have also found $\eta || \tilde{a}$ 'be.loc' used in when-questions, and in the following question.

- (13) a. kija xe ku !ũu xŋ |?aa ŋ||ã where Q 3sg grandfather Past die be.loc 'Where did his grandfather die?'
 - b. !hoe he ku !ũu xŋ |?aa ŋ||ã place Rel 3sg grandfather Past die be.loc 'the place where his grandfather died'
- (14) a. kija xe ku |qhõ[°] ŋ||ã where Q 3sg dance be.loc
 'Where is he dancing?'
 - b. !hoe he ku |qhõ[§] ŋ||ã place Rel 3sg dance be.loc 'the place where he is dancing'

In these cases, it is impossible to strand the linker at the end of the sentence.

(15)	a.	kija xe ku where Q 3sg		!ũu grar	!uu grandfather		?aa die	ŋ ã be.loc	(*ŋ) Lk	
	b.	kija where		xe Q	ku 3sg	qhõ [°] dance	ŋ ã be.loc	(*ŋ) Lk		

With verbs like 'go' (see (12)), which do not take a linker, $\eta || \tilde{a}$ 'be.loc' is not used when the complement is extracted.

- (16) a. ku -a si ||?ae g!ari
 3sgDecl Fut go Upington
 'He will go to Upington'
 - b. kija xe ku si ||?ae where Q 3sg Fut go 'Where will he go?'
 - c. !hoe he ku si ||?ae place Rel 3sg Fut go 'the place where he will go?'

 $y||\tilde{a}$ 'be.loc' can only replace the linker if there has been extraction.

- (17) a. ku !ũu ke xŋ |?aa ŋ (*ŋ||ã) g!ari
 3sg grandfather Decl Past die Lk be.loc Upington
 'His grandfather died in Upington'
 - b. ku -a |qhõ^s ŋ (*ŋ||ã) ku ŋ||ŋ ||ã?ē
 3sg Decl dance Lk be.loc 3sg house in
 'He is dancing in his house'

These extraction facts show that the system incorporates the following constraint.

(18) *Lk <XP>

This constraint says that the linker cannot be immediately followed by a gap (the notation $\langle XP \rangle$ means that the constituent XP has been displaced). In order to avoid (18), the linker y is replaced by $y || \tilde{a}$ 'be.loc' as a last resort.

3. How-Constructions

The only exceptions to the generalization that the linker cannot be placed between a transitive verb and its direct object come from causatives (see section 6) and how-questions. How-questions are illustrated below⁷ (*Mann* in the gloss stands for 'manner').

(19)	a.	ku si	Jee	ŋ	∥x'oo	o ŋ	0 00	-ke
		3sg Fut	how	Mann	chop	Lk	wood	-Pl
		'How w	rill he c	hop the	e woo	d?'		
	b.	ku si	Jee	ŋ	hoo	ŋ	ku	aŋki
		3sgFut	how	Mann	find	Lk	3sg	father
		'How w	ill he f	ind his	father	?'		

⁷ A tentative syntactic analysis of how-questions is that the *jee* 'how' originates in the immediate post-verbal position, and moves to Spec MannP. Since the direct object is no longer in the immediately post-verbal position, a linker is required. I do not have sufficient data as of yet to support this analysis.

C.	ku si	Jee	ŋ	?ae	ŋ	g!ari
	3sg Fut	how	Mann	go	Lk	Upington
	'How w	ill he g	o to Uj	pingto	n?'	

d. Simon si iee hoo ŋ ku g!ari ŋ ŋ Upington Simon Fut how Mann find Lk 3sg Lk 'How will Simon find him in Upington?'

A strong argument that the η following the verb in (19a-d) is the linker is based on the distribution of strong pronouns. The full paradigm for strong and weak pronouns is given below.

In these examples, the first and second person pronouns have a different form (which incorporates a dental click) after the linker. This form is the strong form (see Güldemann 2004b for a brief description of the distribution of the strong form). Weak pronouns can only appear as subject (Spec IP), as possesors (Spec DP), or objects adjacent to the verb. Since the position following the linker is none of these, the strong form is used. Note that weak forms can be used after the instrumental preposition 'with': y|a 2y 'with me'. Examples illustrating the syntactic distribution of weak pronouns are given below.

(21) a. Simon ke si fioo ?ŋ Simon Decl Fut find 1sg 'Simon will find me'

- b. ŋ ke si ||?ae 1sg Decl Fut go 'I will go'
- c. ŋ xaŋki lsgmother 'my mother'

If a how-question is formed with a verb taking a pronominal object, then the strong form is used, just like with other uses of the linker (see (34) below).

(22) Simon si jee n fioo n n|n Simon Fut how Mann find Lk lsg.strong 'How will Simon find me?'

The fact that the linker can appear before a non-locative direct object in how-questions reinforces the claim made above (see the discussion above (6)) that the linker is not a locative preposition. This conclusion will be strengthened when we consider causatives in section 6. In fact, I claim that the linker is semantically vacuous. It is a purely grammatical morpheme. In terms of Minimalist syntax, the linker is non-interpretable at the LF-interface. The property of being semantically vacuous is common to the linker in \pm Hoan, Jul'hoansi and N|uu, and so can be taken as one of its defining properties.

Interestingly, in Jul'hoansi, how-questions involve the presence of the transitivity suffix, which is one form of the linker (the other form is ko). This cross-linguistic evidence supports my claim that the post-verbal η in how-questions in N|uu is the linker.

(23) Ju re naun kuru-a tchi (Ju|'hoansi) person Q how make-Trans arrow 'How does a person make an arrow?' (Dickens 1992)

When the verb is intransitive, no linker appears following the verb in a how-construction.

- (24) a. ku -a si Oun
 3sgDecl Fut sleep
 'He will sleep'
 b. ku si jee ŋ Oun
 - 3sgFut how Mann sleep 'How will he sleep?'

This fact is consistent with the generalization that the linker always precedes a DP, which I account for in section 4.

I have glossed the preverbal y in the above examples as 'Mann', which stands for manner. The preverbal y is not a linker. It has none of the properties of a linker. First, linkers always precede DPs, which I will account for in the next section. The preverbal y precedes a verb, not a DP. Second, the preverbal y is restricted to how-questions and thus-constructions, illustrated below.⁸

(25) ku xa ŋ ||x'oo ŋ Ooo-ke 3sg Past Mann chop Lk wood-Pl 'He chopped the wood thus'

This suggests that the preverbal η is semantically restricted in a way unlike the linker, which appears in a wide variety of constructions (before locative, temporal and reason adjuncts, in causatives and in how-questions).

4. Case Analysis

I propose that the linker is a Case checking head. This analysis explains why the linker never appears before the direct object of a transitive verb (see (8,9)). Sentence (9) is ungrammatical because there are two Case checkers (the linker and the transitive verb), but only one DP. Second, the Case analysis predicts that the linker will only precede DPs. The prediction seems to hold true. First, in all the examples where the linker has appeared so far, it appears before a DP (g!ari 'Upington' in (1a), \neq qhee 'duiker' in (1b)).

The linker also appears before phrases headed by locative postpositions. Some examples of locative postpositions are given below.

- (26) a. n-a ||hoo-a mari !ao ghaa η 1sg-Decl put-Asp money Lk rock next.to 'I put the money next to the rock' b. a ke ts?i si hoo mari ŋ !ao
 - 2sgDecl Fut find money Lk rock behind 'You will find the money behind the rock'

⁸ In the following example, I assume that there is a null constituent meaning 'way/manner' occupying Spec MannP.

c. |x'esi ||?a sui |oba xu necklace go sit.down child front 'The necklace fell in front of the child' (From Collins & Namaseb 2005a)

d. Ku-a |?e ŋ||ŋ ||ã?ẽ 3sg-Decl go.in house in 'He goes into the house'

As argued in Collins (2001, 2003), locative postpositions in \pm Hoan and Ju|'hoansi are inalienable nouns. For example, locative postpositions in \pm Hoan trigger genitive Case on the 1sg pronoun. There is some evidence that locative postpositions are nominal in N|uu as well. For example, the locative phrases can appear as subject.

- (27) a. bekersi ||ã?ẽ ke |x'urixa cup inside Decl dirty
 'The inside of the cup is dirty'
 - b. ŋ xu ke |x'urixa lsg face Decl dirty 'My face is dirty'
 - c. ŋ |qha ke |x'urixa lsgside Decl dirty 'My side is dirty'

d. ŋ||ŋ tsĩ ke |x'urixa house back Decl dirty
'The back (backyard) of the house is dirty'

The fact that locative phrases can appear as subjects implies that they are DPs, since normally only DPs can be subjects. Therefore, I conclude that locative postpositions are nominal in N|uu as well. Therefore, in the examples in (26a,b), the linkers appear before DPs (see Baker & Collins 2005 for a discussion of nominal locatives in Kinande). Therefore the generalization that the linker always appears before DPs holds up.

Third, in how-constructions, the linker precedes the direct object of a transitive verb. In terms of the Case analysis, the verb loses the ability to check Case in the how-construction. The linker never follows an intransitive verb (see (24b)). Since there is no DP following the intransitive verb, no linker is necessary.

Fourth, the linker does not appear before with-phrases (as we will see in section 5), which unlike DPs do not have a Case feature that needs to be checked. Other than the locative postpositions, and the instrumental prepositions, there are no other prepositions or postpositions in N|uu. In particular, there is no preposition such as "across". Most locative concepts are expressed with verbal compounds (see Collins & Namaseb 2005b for examples).

Similarly, the linker never appears before complement clauses. This is illustrated in the following examples. I indicate the embedded clause with bracketing.

- (28) a. n -a \neq ?ãi-a [Siso xŋ ||?ai-a] 1sg Decl think-Asp Siso Past go-Asp 'I think Siso left'
 - b. n -a ‡ao-a [kin ||?ae] Isg Decl want-Asp they leave 'I want that they leave'
 - c. g|a ‡7ii-a [!Q'oma si ||?ae |q'õa]
 2sgthink-Asp !Q'oma Fut go hunting
 'Do you think !Q'oma will go hunting?'
 - d. n -a ||hae-a [ku ki mari] lsg Decl know-Asp 3sg have money 'I know he has money'
 - e. kinke ku-a ŋ ŋ|ŋ [ŋ ||?ae] 3pl Decl say-Asp Lk 1sg.Strong 1sg go 'They told me to leave'
 - f. n -a xŋ ku-a [Siso xŋ ||?ae-a] lsg Decl Past say-Asp Siso Past go-Asp 'I said Siso left'

The fact that the linker does not appear before complement clauses can be explained in terms of the assumption (common in the Principles and Parameters/Minimalist theory) that clauses do not have a Case feature. This assumption is usually justified in Principles and Parameters on the basis of the syntactic distribution of clauses. For example, the clausal complement of an adjective does not require 'of', as shown by the phrase 'proud that John won'. On the other hand, a DP complement of an adjective does require 'of', as in the phrase 'proud of John'. This contrast can be accounted for by the assumption that 'of' checks Case, and that DPs but not clauses have a Case feature.

In conclusion, the linker has the following properties: (a) it is post-verbal, (b) it is semantically vacuous, and (c) it requires the presence of a following nominal expression.⁹ I have analyzed the last property in terms of Case theory.

5. V DP Lk X

In this section, I will discuss some cases where the linker appears between two post-verbal constituents. When a locative appears following a verb with a direct object, a linker must precede the locative.

(29)	a.	ku -a	si	hoo	ku	aŋki	ŋ	g!ari
		3sg Decl	Fut	find	3sg	father	Lk	Upington
		'He will	find h	is fathe	er in U	pingto	n'	
	b.	ku -a	si	x'oo	000	o-ke	ŋ	!uu
		2sg Decl	Fut	chop	woo	d-pl	Lk	veld
		'He will	chop	wood ii	n the v	eld'		
	c.	kunisi	ke	si	+am	!ao	ŋ	!an
		wagon	Decl	Fut	hit	rock	Lk	road
		'The wa	gon w	ill hit tl	ne rocl	c in the	e road'	

Similar facts hold for *‡*Hoan and Ju|'hoansi (from Collins 2003:1).

⁹ The passive preposition by in English has the same three properties as the linker: (a) it is postverbal, (b) it is semantically vacuous and (c) it requires a following DP. However, I would not classify the passive preposition by as a linker. First, by is largely restricted to the passive. Second, it only occurs with the external argument of a verb. See Collins (2005) for a full analysis of the passive in English, where by is analyzed as a realization of the Voice head.

(30)	koloi	g on-a	‡'amkoe	ki	gyeo	na	(‡Hoan)
	car	hit-perf	person	Lk	road	in	
	'A ca	r hit a person	in the road'				

(31)	uto	dchuun-a	Kaece	ko	n!ama	n!ang	(Ju 'hoansi)
	car	hit-trans	Kaece	Lk	road	in	
	'A ca	r hit Kaece ii	n the road'				

This use of the linker is particularly frequent in locative compounds, which add a locative argument to a verb (see Collins 2002 on Khoisan compounds).

(32) a. n -a +hãu !qhaa si 1sgDecl Fut pour water 'I will pour water' b. n -a ≠hãu |?ee $!qhaa \eta xa\beta asi$ llã?ẽ si Fut pour put.in water Lk cup 1sg Decl in 'I will pour water into the cup'

The linker is also used between a direct object and various non-locative expressions (see (6)).

(33)	a.	n -a	si	ŋ!hoea	mari	ŋ	ku
		1sg Decl	Fut	ask	money	Lk	3sg
		'I will as	sk him	for mon	ey'		
	b.	n -a	si	gaake	mari	ŋ	ku
		1sg Decl	Fut	steal	money	Lk	3sg
		'I will st	eal mo	oney from	n him'		
	c.	ku -a	si	!xama	?ãki	ŋ	!haeka
		3sg Decl	Fut	cook	food	Lk	tomorrow
		'He will	cook	food tom	orrow'		

When a pronoun is used after the linker here, it is in the strong form (see also (22)) above.

(34)	ku	xa	ŋ!hoea	?ãki	ŋ	յ∣ŋ
	3sg	Past	ask	food	Lk	1sg.Strong
	'She	asked	him for food	'		

All the other strong pronouns can be used after the linker as well: g|a '2sg', ku '3sg', g|i '1pl', g|u '2pl', and *kike* '3pl'.

The order of the theme and the locative is necessarily [V Theme Lk Loc]. The inverted order [V Loc Lk Theme] is not allowed (compare to (29)).

(35)	a.	*ku	-a	si	hoo	g!ari		ŋ	ku	aŋki
		3sg	Decl	Fut	find	Uping	gton	Lk	3sg	father
		'He wil	l find	his fa	ather i	n Upii	ngtor	1'		
	b.	*ku	-a	si	x'o	00	!uu	ŋ	000-	ke
		3sg	Decl	Fut	chop)	veld	Lk	wood-	pl
		'He wil	l chop	woo	od in th	ne velo	d'			
	c.	*kunisi	ke	si	+ a	ım !a	an	ŋ	!ao	
		wagon	Decl	Fu	t hit	r ro	oad	Lk	rock	
		'The wa	agon v	vill h	it the	rock in	n the	road'		

Therefore, I conclude that N|uu disallows inversion in linker constructions. The lack of inversion in N|uu is interesting in that Afrikaans, which all of my informants (five all together) spoke fluently, has considerable freedom in the ordering of the theme and the locative.

As for the other Khoisan languages, Jul'hoansi allows inversion, and \pm Hoan disallows it. See Collins (2003) for a comparison between \pm Hoan and Jul'hoansi, and see Dickens (1992) who describes inversion in Jul'hoansi. These facts are illustrated below.¹⁰

(36) a. ha ku ||ohm-a !aihn ko g|ui (Ju|'hoansi) 3sgAsp chop-Trans tree Lk forest 'He was chopping the tree in the forest' (Dickens 1992)

178

¹⁰Since only Jul'hoansi has a transitivity suffix -a among the three languages in question, I conjecture that the presence of inversion in Jul'hoansi is related to the transitivity suffix.

	b.	ha ku 3sg Asp	ohm-a chon-Trans	g ui forest	ko Lk	!aihn		(Ju 'hoansi)
		Jsgrsh	chop-mans	loiest	LK	ucc		
(37)	a.	koloi	g on-a	+'am	koe	ki	gyeo na	(+ Hoan)
		car	hit-Perf	persor	1	Lk	road in	
		'The car	thit the perso	on in th	e road	d' (Coll	ins 2003)	
	b.	*koloi	g on-a	gyeo	na	ki	+'amkoe	(‡Hoan)
		car	hit-Perf	road	in	Lk	person	

Other possible word orders of the theme, Lk, and locative in N|uu are ungrammatical (similar sentences are ungrammatical in \pm Hoan and Ju|'hoansi as well, see Collins 2003).

(38)	a.	*ku -a 3sg Decl	si Fut	hoo find	ŋ Lk	ku 3sg	aŋki g father l	g!ari Upington
	b.	*ku -a 3sg Decl	si Fut	hoo find	ku 3sg	aŋki father	g!ari Upingto	ŋ on Lk

In (38a), the linker precedes both of the post-verbal constituents, and in (38b) the linker follows both of the post-verbal constituents. As I argued in Collins (2003), these ordering restrictions suggest that the linker is a functional head whose specifier is the theme and whose complement contains the locative (assuming the order Spec-Head-Complement).



In this structure, both the theme and the locative start out internal to the VP.¹¹ In \pm Hoan, Ju|'hoansi and N|uu, the theme can be moved into the specifier of the LkP, giving rise to the order: [Theme Lk Loc]. Only in Ju|'hoansi can the Loc move into the specifier of Lk giving rise to the order: [Loc Lk Theme].

An important source of evidence for the structure in (39) is the linker in the Bantu language Kinande, discussed by Baker & Collins (2005). In Kinande (as opposed to the Khoisan languages), the linker agrees with the preceding DP indicating a Spec-Head relation. The discussion of the linker in Kinande brings up a general question of the relationship between the linker and the Bantu associative marker. Both the linker in Kinande and the associative marker agree with a preceding nominal constituent, and they have a similar pattern of agreement. The big difference between the linker and the associative marker is that the linker appears internal to the vP, whereas the associative marker appears internal to the DP. See Baker & Collins for much more detail on the linker in Kinande, and some brief remarks on the associative marker.

When the theme is extracted, the linker continues to precede the locative as in the relative clause in (40a) and the constituent question in (40b).¹²

(40)	a.	000-ke	he	Griet	si	x'oo	ŋ	!uu
		wood-Pl	Rel	Griet	Fut	chop	Lk	veld
		'The wo	od that	Griet	will c	chop in t	he ve	ld'

b. cui xe Griet si ||?a ||x'oo ŋ !uu what Q Griet Fut go chop Lk veld 'What will Griet go chop in the veld?'

When the locative is extracted, the verb $y||\tilde{a}$ 'be.loc' must be used, just as with intransitives.¹³

¹¹See Chomsky (1995) and Collins (2003) for further discussion of the internal structure of the verb phrase. In these sources, it is assumed that the verb undergoes further movement to the head of vP (light verb phrase).

¹² ‡Hoan has the same pattern. In Ju|'hoansi, the Lk drops, yielding V-a DP (See Collins 2003).

¹³When the locative is extracted in ‡Hoan, the order becomes V Lk DP, as if there had been inversion (which is normally not allowed in ‡Hoan). In Jul'hoansi, the linker drops, yielding: V-a DP. These patterns indicate that constraints similar to (18) need to be postulated for ‡Hoan and Jul'hoansi as well (although there are a number of differences, see Collins (2003) for some discussion).

- (41) a. !uu-a he Griet si ||x'oo Ooo-ke ŋ||ã veld-Dem Rel Griet Fut chop wood-Pl be.loc 'The veld where Griet will chop wood'
 - b. kija xe Griet si ||x'oo Ooo-ke ŋ||ã where Q Griet Fut chop wood-Pl be.loc 'Where will Griet chop wood?'

Although the linker is used before locatives and non-locative expressions, it is not used before instruments. Rather, the instrumental preposition $y|\tilde{a}$ 'with' is used.

(42)	a.	n	-a	si	aa	Ooe	ŋ∣ã	ŋ‡ona	ı
		1sg	Decl	Fut	cut	meat	with	knife	
		ʻI w	vill cu	t the n	neat wi	ith a kı	nife'		
	b.	n	-a	si	x'oo	‡qhii	ŋ∣ã	!?oo	
		1sg	Decl	Fut	chop	tree	with	axe	
		ʻIw	ill ch	op the	tree w	ith an	axe'		
	c.	Gri	et	ke	kuru-a	a	ŋ∥ŋ	ŋ∣ã	qhe
		Grie	et	Decl	make-/	Asp	house	with	grass
		'Gr	iet bu	ilt a ho	ouse w	ith gra	ss'		-

In these examples, the linker is never found before $y|\tilde{a}$ 'with'. As noted in section 3, this fact can be accounted for on the assumption that with-phrases do not have a Case feature that needs to be checked.

6. Causatives

Another example where a linker appears between two post-verbal constituents (V X Lk Y) is the causative. Consider first the following examples of causatives of intransitive verbs.

(43) a. n -a si kx'u ||?ae ku lsgDecl Fut make go 3sg 'I will make him go'

- b. n -a kx'u Oun-a |oeke lsgDecl make sleep-Asp children 'I made the children sleep'
- c. n -a kx'u |q'õa-a ku lsgDecl make hunt-Asp 3sg 'I made him hunt'
- d. n -a xŋ kx'u ?ã ku 1sgDecl Past make eat 3sg 'I made him eat'

These examples illustrate that the form of a causative of an intransitive verb is [kx'u-V DP], where the only argument of V follows V. Kx'u 'make' used as an independent verb is shown below.

(44) n -a si kx'u ŋ||ŋ 1sg Decl Fut build house 'I will build a house'

When a transitive verb is causativized, a linker must appear before the theme.

(45)	a.	n	-a	kx'u	q'õa-	·a	ku	ŋ	‡qhe	e
		1 sg	g Decl	make	hunt-A	sp	3sg	Lk	duike	r
		ίI	made	him hư	ınt a dı					
	b.	n	-a	xa	kx'u	?ã	ku	ŋ	ku	?ãki
		ls	g Decl	Past	make	eat	3sg	Lk	3sg	food
		ίI	made	him ea	t his fo					

I explain this use of the linker by assuming that the causativized verb can not check accusative Case (see Baker 1988: 173 for a theoretical account¹⁴). In the causative, kx'u 'make' itself checks the Case of the embedded external argu-

¹⁴In Baker's terminology (p. 164), N|uu has a Causative Rule 2, which is also found in Kiswahili and Chamorro (which appears to have a linker).

ment (e.g., ku 3sg in 43a,b), and the linker checks the Case of the embedded internal argument ($\pm qhee$ 'duiker' in (45a), and ? $\tilde{a}ki$ 'food' in (45b)).

7. Double Object Constructions

One of the most striking aspects of the linker in N|uu is that it does not appear in double object constructions (DOCs). This is illustrated for 'give' in (46, 47), for 'show' in (48), 'send' in (49), and for benefactive arguments in (50, 51).

(46)	a.	ku -a	si	?ãa	ku	aŋki-a	ı	kea	mari
		3sg Decl	Fut	give	3sg	father-	Dat	that	money
		'He will	give h	is fath	er that	money	,,		
	b.	Griet	ke	si	?ãa	ku-a	doŋ	ki-si	
		Griet	Decl	Fut	give	3sg-Da	it donl	key-Sg	
		'Griet	will g	give hin	m the o	lonkey	,		
(47)	a.	?ãa	ku-a	ku	xaŋki	i ka-	hãi		
		Give	3sg-D	at 3sg	mothe	r Pl	breasts	5	
		'Give he	r her n	nother	's brea	sts'			
		(from Co	ollins &	& Nam	aseb 2	.005a)			
	b.	?ãa	ku-a	mai	r ku	xaŋki	se	ŋ ã	
		give	3sg-D	at but	3sg	mother	Poss	head	
		'But, giv	e her l	ner mo	ther's	head'			
		(from Co	ollins &	& Nam	aseb 2	.005a)			
(48)	a.	Griet	ke	si	kajar	na	ku-a	si	doŋki-si
		Griet	Decl	Fut	show		3sg-D	at 1pl	donkey-Sg
		'Griet w	ill sho [,]	w him	our do	onkey'			
	b.	ku -a	si	kajan	na ŋ	angusi	-a	Ooe	
		3sg Decl	Fut	show	N	angusi-I	Dat	meat	
		'He will	show	N angı	usi the	meat'			
(49)	n-	a	si	ee-a		ku-a	ki		

(49) n-a si |ee-a ku-a ki
1sg-Decl Fut send-Appl 3sg-Dat 3sg-NH
'I will send it to him' (NH stands for non-human)

(50)	ku	-a	si	kura-a	n-a	x'ãike			
	3sg	Decl	Fut	make-Appl	lsg-Dat	beads			
	'He will make beads for me'								

(51) kin ke ŋ!hau-a ku-a Ooe 3pl Decl roast-Appl 3sg-Dat meat 'They are roasting him meat'

In these examples, Appl in the glosses stands for 'applicative', which is a verbal suffix that allows a verb to take a benefactive or goal argument. I will argue that the -a (glossed Dat) that follows the first object in a double object construction is a dative Case marker.

In all of the above examples, there is no linker between the first DP (immediately following the verb) and the second (theme) DP. A double object construction with a linker present is judged ungrammatical.

(52)	n	-a	si	kajama	ku-a	(*ŋ)	ku	mari
	lsg	Decl	Fut	show	3sg-Dat	Lk	3sg	money
	'I wi	ll show	him h	his money'				

(53) n -a si ?ãa ku-a (*ŋ) ku mari
1sg Decl Fut give 3sg-Dat Lk 3sg money
'I will give him his money'

One informant said that (53) with the linker could mean that I will give him something out of his money, which is not the interpretation of a DOC. The lack of the linker in DOCs in N|uu is consistent with the little that is known about other southern Khoisan languages (see Güldemann forthcoming a,b, and Hastings 2001: 12 on |Xam).

The absence of the linker in DOCs is surprising for a number of reasons. First, we have already seen that the linker is used in a wide variety of environments, including preceding themes (how-questions and causatives), showing that there is no thematic restriction of the linker to locative expressions.

Second, in causatives (see (45)), two DPs follow the verb and the second DP is preceded by a linker. On analogy with causatives, we might expect to find such a linker before the second DP in a double object construction.

Third, other Khoisan languages (Ju|'hoansi and \neq Hoan) have a linker that does appear before the second object in a double object construction. This is shown below (from Collins 2003).

(54)	Gya'	'm-∣a'a	a-tsaxo-o	cu '	am	gye	ki	a''e	(+ Hoan)
	child-	Dim.pl	Prog-cool	k-give	lsg	mother	Lk	meat	
	the c	children	are cooki	ng mea	at for 1	my mo	ther'		
(55)	Besa	komm	'ama	- 'an	Oba	ko	tcis	i	(Ju 'hoansi)
	Besa	Emph	buy	-give	Oba	Lk	thin	gs	
	'Besa	ı bought	Oba som	e thing	gs'				

I propose that the absence of a linker in DOCs in N|uu is related to another striking feature of DOCs. Note that the first object (the one immediately following the verb) is followed by Dat -a in N|uu in the examples in (46-51). In the rest of the section, I will justify an analysis of Dat -a as a Case checker, and show how the presence of the Dat -a gives rise to the absence of the linker in a DOC.

Dat -a can also appear when there is no following theme argument. The only condition is that the Dat -a can only mark a goal or a benefactive argument.

(56) a. n -a si g uu-a ku-a lie-Appl lsgDecl Fut 3sg-Dat 'I will lie to him' b. n -a si sĩsən-a ku-a 1sg-Decl Fut work-Appl 3sg-Dat 'I will work for him' |ahõ[°]-a si Griet-a c. n -a dance-Appl Griet-Dat 1sgDecl Fut 'I will dance for Griet'

In these examples, I propose that the Case of the DP is not checked by the verb, but rather by the Dat *-a*. Evidence for this analysis is that the linker does not appear before a dative marked DP in a how-construction.

- (57) a. Griet si jee ŋ |qhõ[§]-a ku-a Griet Fut how mann dance-Appl 3sg-Dat 'How will Griet dance for him?'
 - b. Griet si jee ŋ g|uu-a ku-a Griet Fut how Mann lie-Appl 3sg-Dat 'How will Griet lie to him?'

Given that the linker checks the Case of the following DP, and that DP following the verb already has a dative Case marker, there is no need for the linker to be used. Predictably, Dat -a is impossible if the DP following the verb is not a goal or benefactive.

(58)	n	-a	si	ŋ∣aa	ku	(*-a)
	1sg	Decl	Fut	see	3sg	Dat
	ʻI wi	ll see h	nim'			

Dat -a morphologically resembles the declarative -a found following the subject of declarative clauses (they both have the same high tone). I will compare the properties of declarative -a and dative -a, and I will show that although dative -a and declarative -a are similar, they can not be identified. I give the complete pronominal paradigm for dative -a in (59) and for declarative -a in (60).

(59)	a.	Andries	ke	xa	?ãa	n-a	ĥa	mari
		Andries	Decl	Past	give	1sg-Dat	3sg	money
		'Andries	gave 1	me his	money	у'		
	b.	Andries	ke	xa	?ãa	?a	ĥa	mari
			Decl	Past	give	2sg-Dat	3sg	money
	c.	Andries	ke	xa	?ãa	ku-a	ĥa	mari
			Decl	Past	give	3sg-Dat	3sg	money
	d.	Andries	ke	xa	?ãa	s-a	ĥa	mari
			Decl	Past	give	1pl-Dat	3sg	money

186

 f. Andries ke xa ?ãa kik-a fia mari Decl Past give 3pl-Dat 3sg money¹⁵ (60) a. n-a si ?ae Isg Decl Fut go 'I will go' b. a si ?ae 2sg Decl Fut go c. ku a si ?ae 3sg Decl Fut go d. s -a si ?ae 1pl Decl Fut go e. b -a si ?ae 2pl Decl Fut go f. kin ke si ?ae 3pl Decl Fut go 		e.	Andries	ke Decl	xa Past	?ãa give	b-a 2pl-Dat	ha 3sg	mari monev
 (60) a. n-a si ?ae lsg Decl Fut go 'I will go' b. a si ?ae 2sg Decl Fut go c. ku a si ?ae 3sg Decl Fut go d. s -a si ?ae 1pl Decl Fut go e. b -a si ?ae 2pl Decl Fut go f. kin ke si ?ae 3pl Decl Fut go 		f.	Andries	ke Decl	xa Past	?ãa give	kik-a 3pl-Dat	fia 3sg	mari money ¹⁵
b. a si ?ae 2 sg Decl Fut go c. ku a si ?ae 3 sg Decl Fut go d. s -a si ?ae 1 pl Decl Fut go e. b -a si ?ae 2 pl Decl Fut go f. kin ke si ?ae 3 pl Decl Fut go	(60)	a.	n-a 1sg Decl 'I will gc	si Fut)	?ae go				
c. ku a si ?ae 3sg Decl Fut go d. s -a si ?ae 1pl Decl Fut go e. b -a si ?ae 2pl Decl Fut go f. kin ke si ?ae 3pl Decl Fut go		b.	a 2sg Decl	si Fut	?ae go				
d. s -a si ?ae lpl Decl Fut go e. b -a si ?ae 2pl Decl Fut go f. kin ke si ?ae 3pl Decl Fut go		c.	ku a 3sg Decl	si Fut	?ae go				
e. b -a si ?ae 2pl Decl Fut go f. kin ke si ?ae 3pl Decl Fut go		d.	s -a	si Fut					
f. kin ke si ?ae 3pl Decl Fut go		e.	b -a	si Si	?ae				
		f.	kin ke 3pl Decl	rut si Fut	go ?ae go				

With the exception of third plural,¹⁶ the dative pronouns and the declarative pronouns are identical. This series of pronouns is called the A-Form of the pronoun in Collins & Namaseb (2005b). I know of no other place in N|uu grammar where this particular combination of weak pronoun plus -*a* occurs. The parallelism between the subject -*a* pronouns and the dative pronouns suggests that Decl -*a* shares some properties with dative -*a*. While this may be true, there are some reasons for not identifying the two morphemes.

¹⁵The 3pl dative -*a* pronoun can be kin-a as in the following example:

cui xe Sini xŋ ?ãa kɨn-a what Q Sini Past give 3pl-Dat 'What did Sini give him?'

¹⁶The pronoun kin-a is not possible as the subject in (60f). While ke can appear following all subject types (pronouns, proper names, full DPs), the declarative -a seems restricted to pronouns for which the combination pronoun-a is monosyllabic (crucially, kua 3sg-decl is pronounced [kwa] and not [ku?a]). In all my field notes, and all the sentences in Westphal (see his examples (34-36), (121), (231), (260)), I have found no clear counter-examples to this generalization. No similar restriction exists for dative -a. Much more work is needed on this prosodic restriction.

In a declarative sentence, declarative -a can be used following the subject. Alternatively, declarative ke can be used following the subject. Furthermore, the two can not be used simultaneously (see Güldemann 2004b). Declarative ke is illustrated below.

(61)	ŋ	ke	xa	SOO	kia
	1sg	Decl	Past	sit	here
	ʻI wa	as sittin	ng here	e'	

The series of subject pronouns that can precede declarative ke is the following: η '1sg', a '2sg', ku '3sg', i, si '1pl, u '2pl', kin '3pl'. The following example shows that in some cases either *-a* or *ke* can be used as the declarative head.

(62)	a.	n -a	xŋ	?ae-a	!gari
		1sg Decl	Past	go-Asp	Upington
		'I went t	o Upir	ngton'	
	b.	ŋ ke	xŋ	?ae-a	g!ari
		1sg Decl	Past	go-Asp	Upington
		'I went t	o Upii	ngton'	

The following examples show that declarative -a and declarative ke cannot appear simultaneously.

(63)	a.	n	-a	si	Oun	
		$1s_{2}$	g Decl	Fut	sleep	
		ίI	will sl	eep'		
	b.	ŋ	ke	si	Oun	
		$1s_{2}$	g Decl	Fut	sleep	
		ʻI	will sl	eep'		
	c.	*n	1 -a	ke	si	Oun
		1s	g Decl	Decl	Fut	sleep
(64)	a.	n	-a	xŋ	?ae	g!ari
		1s	g Decl	Past	go	Upington
		ʻI	went t	o Upir	ngton'	

b. n ke xŋ ||?ae g!ari 1sgDecl Past go Upington 'I went to Upington'

c. *n -a ke ||?ae g!ari 1sgDecl Decl go Upington

Furthermore, neither declarative ke nor declarative -a appears in questions or relative clauses (or embedded clauses more generally).

(65) a. kita ||?ae si xe ku where 0 3sg Fut go 'Where will he go?' b. gao he ku si ||?ama thing Rel 3sg Fut buy 'the thing that he will buy'

So it appears that -a and ke following the subject are two realizations of single morpheme, which I will identify as the functional category Decl. Dative -a, on the other hand, does not alternate with a ke. In fact, ke cannot appear following the first object of a DOC.

(66)	a.	Griet	ke	si	?ãa	ku-a		doŋki-si
		Griet 'Griet	Decl will g	Fut ive hir	give n the d	3sg-Da lonkey	at	donkey-Sg
	b.	*Griet Griet	ke Decl	si Fut	?ãa give	ku 3sg	ke Decl	doŋki-si donkey-Sg

Furthermore, dative -a, unlike declarative -a, can appear in questions.

(67) a. ku -a si ?ãa ku-a mari 3sg Decl Fut give 3sg-Dat money 'He will give her the money' b. ku si ?ãa ku-a mari 3sg Fut give 3sg-Dat money 'Will he give her the money?'

In (67a), both the declarative -a and the dative -a appear. In the yes-no question in (67b), the declarative -a disappears, but the dative -a remains.

For these reasons, I will not identify dative -a and declarative -a. However, I propose that they are partially similar, in that both the declarative -a and the dative -a head clausal functional projections. In the case of declarative -a, it is the head of IP.¹⁷ In the case of dative -a, it is a functional projection which I will simply call DatP. In this way, dative -a resembles the linker as well, which heads a functional projection taking a VP complement. The two structures are given below (for more on the general framework for this kind of syntactic structure, see Chomsky 1995 and Collins 2003).

(68) IP DP Infl' Infl VP -a/ke (69) DatP DP Dat' Dat VP -a

I will assume that (unlike the linker) the dative -a does not check the Case of a following DP, thus accounting for the fact that the dative -a does not have to be followed by a constituent in contrast to the linker, see section 4. Rather, it seems natural to analyze the dative -a as checking the Case of the DP in its specifier.

Consider now the lack of inversion in DOCs, given the analysis in (69).

¹⁷Or conceivably the head of a lower projection in the left periphery.

(70)	a.	n	-a	si	?ãa	Griet-a	mari
		lsg	Decl	Fut	give	Griet-Dat	money
		ʻI wil	l give G	riet t	he mon	ney'	
	b.	*n	-a	si	?ãa	mari	Griet-a
		1sg	Decl	Fut	give	money	Griet-Dat
	c.	*n	-a	si	?ãa	mari-a	Griet
		1sg	Decl	Fut	give	money-Dat	Griet
(71)	a.	n	-a	si	kajam	a Simon-	a Ooe
(71)	a.	n 1sg	-a Decl	si Fut	kajam show	a Simon- Simon-I	•a Ooe Dat meat
(71)	a.	n 1sg 'I wil	-a Decl l show S	si Fut Simo	kajam show n the m	a Simon- Simon-I leat'	a Ooe Dat meat
(71)	a. b.	n 1sg 'I wil *n	-a Decl l show S -a	si Fut Simo si	kajam show n the m kajam	a Simon- Simon-I leat' a Ooe	a Ooe Dat meat Simon-a
(71)	a. b.	n 1sg 'I wil *n 1sg	-a Decl I show S -a Decl	si Fut Simo si Fut	kajam show n the m kajam show	a Simon-I Simon-I eat' a Ooe meat	a Ooe Dat meat Simon-a Simon-Dat
(71)	a. b. c.	n 1sg 'I wil *n 1sg *n	-a Decl I show S -a Decl -a	si Fut Simo Si Si Fut Si	kajam show n the m kajam show kajam	a Simon-I Simon-I leat' a Ooe meat 3 a Ooe-a	a Ooe Dat meat Simon-a Simon-Dat Simon

Sentences (70b) and (71b) are unacceptable, since given the analysis in (69), the only way to derive them would be for DatP to have two DP specifiers (to the left of the head Dat), presumably not allowed. Sentences (70c) and (71c) are disallowed, since they would involve inversion. The only way to derive these word orders would be for the theme to move to Spec DatP. But as we saw in (35) above, inversion is not allowed in N|uu.

Consider the following extraction facts in light of the above analysis in (69).

- (72) a. |eeki he Griet si ?ãa fia si doŋki-si woman Rel Griet Fut give 3sg-Dat 1pl donkey-Sg 'the woman Griet will give our donkey to'
 - b. cu xe Griet si ?ãa fia si doŋki-si who Q Griet Fut give 3sg-Dat 1pl donkey-Sg 'Who will Griet give our donkey to?'

- (73) a. |eeki he Griet si kajama fia si doŋki-si woman Rel Griet Fut show 3sg-Dat 1pl donkey-Sg 'the woman Griet will show our donkey to'
 - b. cu xe Griet si kajama fia si doŋki-si who Q Griet Fut show 3sg-Dat 1pl donkey-Sg 'Who will Griet show our donkey to?'

These examples show that when the first object of a DOC is extracted, there is a resumptive pronoun. Furthermore, a resumptive pronoun is also left by extraction from the position following an intransitive verb with an applicative.

(74)	a.	cu	xe	?a	si	g uu-a	ĥa
		who	Q	2sg	Fut	lie-Appl	3sg-Dat
		'Wh	io wi	ill you	lie to:	?'	
	b.	cu	xe	?a	si	qhõ [°] -a	ĥa
		who	Q	2sg	Fut	dance-Appl	3sg-Dat
		'Wh	io w	ill you	dance	e for?'	
	c.	cu	xe	?a	si	sĩsən-a	ĥa
		who	Q	2sg	Fut	work-Appl	3sg-Dat
		'Wh	io w	ill you	work	for?'	

Lastly, there is no resumptive pronoun when the second object of a DOC is extracted.

- (75) a. doŋki-si he Griet si ?ãa ku-a donkey-Sg Rel Griet Fut give 3sg-Dat
 'The donkey that Griet will give him'
 - b. cui xe Griet si ?ãa ku-a what Q Griet Fut give 3sg-Dat 'What will Griet give him?'
- (76) a. doŋki-si he Griet si kajama ku-a
 donkey-Sg Rel Griet Fut show 3sg-Dat
 'The donkey Griet will show to him'

b. cui xe Griet si kajama ku-a
 what Q Griet Fut show 3sg-Dat
 'What will Griet show him?'

The fact that a resumptive pronoun needs to be left behind suggests that the dative -a is a clitic. As a clitic, dative -a must combine with a preceding DP. If the DP has been extracted, a resumptive pronoun needs to be left behind to satisfy the morphological property of dative -a.

How can we explain the lack of a linker in double object constructions? First note that there is a no general prohibition against a linker following an DP marked by Dat -a.

(77) a. Griet ke ?ãa donki-si si ku-a g!ari n give 3sg-Dat donkey-Sg Lk Upington Griet Decl Fut 'Griet will give him a donkey in Upington'

- b. n -a si sĩsən-a ku-a ŋ g!ari
 1sg Decl Fut work-Appl 3sg-Dat Lk Upington
 'I will work for him in Upington'
- c. n -a si g|uu-a ku-a ŋ mari (!?am) 1sgDecl Fut lie-Appl 3sg-Dat Lk money about 'I will lie to him about the money'

Given the presence of the linker in these sentences it is impossible to attribute the lack of a linker in DOCs to a general constraint prohibiting a linker from following a verb with a dative object. Rather, the following generalization seems to be true for N|uu, Ju|'hoansi and \pm Hoan.¹⁸

(78) At most one unmarked DP can follow a verb

Unmarked here means a DP that is (a) not immediately preceded by the linker η and (b) not immediately followed by a dative -*a*. In Minimalism, we can recast this as follows.

¹⁸Khoekhoe would have the following constraint: at most one unmarked DP can appear in a clause. Usually the unmarked DP is the subject, the rest are marked by the clitic -a.

(79) The verb can check the structural Case of at most one DP

The generalizations in (78) and (79) are satisfied by the DOC in N|uu, but we still have not answered why there is no linker in DOCs in N|uu. For example, the sentence in (53), repeated below, would satisfy (78) and (79), even if the linker were present.

(80) n -a si ?ãa ku-a (*ŋ)ku mari
 1sg Decl Fut give 3sg-Dat Lk 3sg money
 'I will give him his money'

I propose that this example should be ruled out in the same way that we ruled out the presence of a linker with a transitive verb in N|uu (see (9) above).

(81) ku -a si fioo (*ŋ) ku aŋki
 3sg Decl Fut find Lk 3sg father
 'He will find his father'

I claimed that this example is bad because there are two structural Case checkers (the verb and the linker) but only one DP. Similarly, in (80) the verb is a structural Case checker that does not need to check the Case of the first object (because of the presence of the dative -a). Therefore, the verb is free to check the Case of the direct object, thus blocking the presence of the linker.

An objection to this analysis is that when a DOC is put into a howquestion, the second object is not preceded by a linker.

(82)	g a	Jee	ŋ	?aa	Griet-a	mari
	2sg	how	mann	give	Griet-Dat	money
	'How	v did y	ou give	e Griet	t the money?'	

We have already seen that no linker is needed before the dative (see (57)) in a how-question. But what about the second object? If the verb is checking the Case of the second object in a DOC, and if a how-question makes it impossible for a verb to check the Case of a direct object (see section 3), then we would expect to find a linker in (82) as well. I leave this question for further research.

8. Conclusion

In this paper, I have given an overview of the syntactic distribution of the linker in N|uu. I have shown how the properties of the linker follow from the theory that it is a Case checking functional head, with no interpretable features (no semantic content).

I then showed the linker in N|uu does not appear before the second object of a double object construction. The absence of the linker in DOCs is surprising for a number of reasons. First, I showed that the linker is used in a wide variety of environments showing that there is no thematic restriction on the distribution of the linker. Second, in causatives, two DPs follow the verb and the second DP is preceded by a linker. On analogy with causatives, we might expect to find a linker before the second DP in a double object construction. The third, and most compelling, reason for expecting a linker in DOCs is that in other Khoisan languages (Ju|'hoansi and \ddagger Hoan), a linker does appear before the second object in a double object construction.

I proposed that the absence of a linker in DOCs in N|uu is related to the presence of a dative marker following the first object of a DOC.

9. A Summary of Linkers in (Non-Central) Khoisan

In this section, I will present a brief summary of linker properties in the three Khoisan languages **+**Hoan, Ju|'hoansi, and N|uu.¹⁹

1.	Property Lk	‡Hoan yes (ki)	Ju 'hoansi yes (-a/ko)	N uu yes (ŋ)
2.	Lk thematically unrestricted	yes	yes	yes
3.	V Lk X	yes	yes (-a)	yes
4.	V X Lk (Lk must precede	no e X)	no	no
5.	V DP Lk X	yes	yes (ko)	yes
6.	V DP X Lk	no	no	no

¹⁹See Hastings (2001) for a preliminary survey of the linker in |Xam, and !Xóõ.

7.	V Lk DP X	no	no	no
8.	Subj Lk X V (see (7a))	no	no	no
9.	Lk in causative (V DP Lk DP)	yes	yes	yes
10.	Trans suffix	no	yes (-a)	no
11.	Inversion (V Loc Lk DP)	no	yes	no
12.	Lk in DOCs	yes	yes	no
13.	Dative	no	no	yes(-a)
14.	Declarative	no	no	yes (-a/ke)
15.	Lk-gap (see (18))	yes (sometimes)	no (for ko) yes (for -a)	no
16.	"be.in" as Last Resort	no	no	yes

Properties (1-9) define the core syntax of the linker, found in all three languages. Seen from the perspective of this chart, the linker in \pm Hoan is defined by the core properties. Properties (10-11) define what is distinctive about Ju|'hoansi, which is basically the presence of the transitivity suffix and all the consequences that follow (see Collins 2003).

Properties (12-16) define what is distinctive about N|uu. I speculate that some of the syntactic features distinguishing N|uu from \pm Hoan and Ju|'hoansi are the result of Khoekhoe influence. For example, Khoekhoe has a declarative *ke*, just like N|uu. Khoekhoe also has clitic *-a* as a general case marker (see Hagman 1977). It may be that this clitic influenced the development of dative *-a* and declarative *-a* in N|uu.

REFERENCES

Baker, Mark. 1988. Incorporation. Chicago: The University of Chicago Press.

Baker, Mark & Chris Collins. 2005. "Linkers and vP Structure". Ms., Rutgers University and New York University.

Chomsky, Noam. 1995. The Minimalist Program. Cambridge: MIT Press.

- Collins, Chris. 2001. "Aspects of Plurality in +Hoan". Language 77: 456-476.
- Collins, Chris. 2002. "Multiple Verb Movement in +Hoan". *Linguistic Inquiry* 33: 1-29.
- Collins, Chris. 2003. "The Internal Structure of vP in Ju|'hoansi and ±Hoan". *Studia Linguistica* 57: 1-25.
- Collins, Chris. 2005. "A Smuggling Approach to the Passive in English". Ms., New York University.
- Collins, Chris & Levi Namaseb. 2005. "N|uuki Stories". Ms., New York University and the University of Namibia.
- Collins, Chris & Levi Namaseb. 2005b. "N|uuki Grammar". Ms., New York University and University of Namibia.
- Dickens, Patrick. 1992. Jul'hoan Grammar. Ms., Nyae Nyae Development Foundation of Namibia, Namibia.
- Güldemann, Tom. 2004a. "The transliteration and interlinearization of Westphal's field notes on N|huki". Unpublished ms. Leipzig: MPI-EVAN/ Linguistics.
- Güldemann, Tom. 2004b. "A linguistic sketch of N|huki". Unpublished ms. Leipzig: MPI-EVAN/ Linguistics.

- Güldemann, Tom. Forthcoming a. "Morphology: |Xam of Strandberg". In Voßen, Rainer (ed.), *The Khoisan languages*. Routledge Language Family Series. London: Routledge.
- Güldemann, Tom. Forthcoming b. "Syntax: |Xam of Strandberg". In Voßen, Rainer (ed.), *The Khoisan languages*. Routledge Language Family Series. London: Routledge.
- Hastings, Rachel. 2001. "Transitivity Particles in |Xam and !Xóõ". Ms., Cornell University.
- Hagman, Roy S. 1977. Nama Hottentot Grammar. Bloomington: Indiana University.

Department of Linguistics New York University 719 Broadway, Fourth Floor New York, New York 10003 cc116@nyu.edul [Received February 3, 2005, accepted April 4, 2005]

Studies in African Linguistics Volume 33, Number 2, 2004

ON THE VERBAL SYSTEM IN LANGI A BANTU LANGUAGE OF TANZANIA (F.33)^{*}

Margaret Dunham Université de Paris 3 & Lacito-CNRS

This paper presents the Langi verbal system and the various ways in which tense, aspect and mood are encoded. Through a description of the structures and uses of the various forms, it attempts to demonstrate how the different conjugations fit together to form a coherent whole, morphologically and semantically, and how in some cases the system has been influenced by surrounding Cushitic languages.

1. Introduction.

Langi¹ (Rangi in Swahili) is a Bantu language spoken by approximately 300,000 people in the Kondoa Region of Tanzania. It is of interest linguistically, as its speakers have long been surrounded by non Bantu-speaking communities. In this paper I will show that this has not been without consequences for the language, as evidenced by certain areal features.

^{*} I thank the following for their helpful comments on earlier drafts of this paper: Christiane Paulian, Zlatka Guentchéva, Denis Creissels, Dave Odden and an anonymous reviewer for SAL. I am also indebted to Derek Nurse and Maarten Mous for pointing out (as well as providing) various articles of interest for this study.

¹ This language is relatively unknown to linguistics: when I began studying it in 1996, the only published work dated from 1916 (by Otto Dempwolff). The data presented here is all first hand, and was gathered during fieldwork I carried out in Tanzania during my doctoral studies, the funding for which was provided by the LACITO-CNRS. Oliver Stegen of SIL has started working on the language recently; so far he has presented a paper on the vowel system at CALL (Leiden) in 2000, and has published a paper on derivation (2002). A monograph on Langi is in press, see Dunham, in press.

This paper presents the verbal system of Langi as it is spoken in the towns of Kondoa and Pahi. Through the description of the structures and uses of the various verb forms, I will attempt to demonstrate how the different conjugations fit together to form a coherent whole, morphologically and semantically. I will begin with a brief presentation of the three types of elements which enter into verbal constructions in Langi: verb forms, copulative verbs and infinitival forms. I will then proceed to present the conjugations attested in the language.

1.1. Verb forms. The verb form in Langi, and in Bantu languages in general, is composed of several elements. These elements are not all necessarily present in a given verb form, but always appear in a fixed order. In Langi, a verb form may contain up to seven elements, which can be represented as follows.

(1)	1	2	3	4	5	6	7
	negative	subject	TAM	-object,	RAD	extensions	TAM
	pre-pfx ²	pfx	TAM	reflexive			

Minimal forms contain only a radical and a suffix. This corresponds to the imperative (see section 2.18 below).

(2) doma³ 'go!' dom -a go -imp2sg

Maximal forms contain seven elements.

² Abbreviations used in this paper are as follows. ADV advisory; APPL applicative extension; ASC associative extension; AUX auxiliary; CAUS causative extension; CONN connective; COP copula; DEC decisional; DEICT deictic; DEM demonstrative; DET determinative; DP dependant nominal prefix; HAB habitual; IMP imperative; INF infinitive; INJ injunctive; INV = inevitable; IP independant nominal prefix; lit literally; LOC locative suffix; n/a not applicable; NAR narrative; NEG negative; NEUT neutral extension; OBJ object marker; PASS passive extension; pers.com. personal communication; PFT perfect; PFV perfective; PERS personal pronoun; PFX prefix; PL plural; POSS possessive; P.PFX negation pre-prefix; PRES presentative; PROG progressive; RAD radical; REFL reflexive; REL relator; SFX suffix; SG singular; SP subject prefix; STAT stative extension; SUB subjunctive; TAM tense-aspect-mood marker; V verb. Numbers in glosses refer to noun classes.

³ Tones will not be marked on examples. Tonal distinctions in Langi, particularly at the grammatical level, are so restricted that not marking them in this study does not affect the analysis.

(3)	siviiyəvasəmera tuku	د،	They are not reading to them.'
	si - va - iyo - va - som -er	-a	tuku
	p.pfx - sp2 - prog - obj2 - read -appl	-prog	neg

Position 1. The first element in the verb form is the negative pre-prefix si. Its presence negates the verb form. Negative forms are generally accompanied by the adverb *tuku* at the end of the sentence. This is the only negative marker in the verb form, the suffix, for example, never being modified. We will see in sections 2.20 and 2.21 that imperatives and subjunctives are negated in a different manner.

Position 2. The second element in the verb form is the subject prefix. With the exception of imperatives, where the person is marked in the 7th position, a subject prefix in the second position is obligatory. The subject prefixes most frequently encountered in this paper will be the person prefixes. Below is a table showing the prefixes as they appear when followed by a consonant.

(4)		Singular	Plural
	1	nı-	tu-
	2	U-	mu-
	3	а-	va-

When followed by a vowel, the forms vary somewhat: the prefix vowel either drops (as in example (3) above) or elides, depending on the articulatory nature of the two vowels in contact and also depending on the nature of the following morpheme. In general, when the prefix vowel is altered, the following vowel is lengthened. The rules governing vowel contact may be found in Dunham (forth-coming).

The subject prefix for the 1^{st} person singular has several distinct variants: it is *ni*- in verb forms, but when prefixed to the copula *-ri*, it is *nde*- in the affirmative and *ndu*- in the negative.

 (5) nīkadʒɛŋga numba 'I built a house.'
 nī -ka -dʒɛŋg -a n -umba splsg -nar -build -nar ip9 -house
(6) ndεrī muhindā, sindurī mutavana tuku
 'I'm a girl, I'm not a boy.'
 ndε -rī mu -hindāa si -ndu -rī mu -tavana tuku splsg -cop ipl -girl p.pfx -splsg -cop ipl -boy neg

The subject prefix for the 3^{rd} person singular is *a*- (\emptyset before a vowel) in all cases except preceding the habitual marker -*p*- where it is *y*-.

(7)	akadʒɛŋga ɲumba	'He built a house.'		
	a -ka -dʒɛŋg-a ɲ -umba spl -nar -build -nar ip9 -house			
(8)	yəədəma ndzırii	'He goes on the path.'		
	y -ɔ -dɔm -a n -dʒɪra -i sp3sg -hab -go -hab ip9 -path -loc			

Position 3. The third position contains tense-aspect-mood markers. These will be largely discussed in the following sections. In brief, conjugations in the verb forms are determined by the elements present in the third and the seventh positions (in the case of the imperative and the subjunctive, by a null marker in the 3^{rd} position). In example (5) above, one can see that I have given the same gloss (nar) to the elements found in what corresponds to the 3^{rd} and 7^{th} positions of the verb form. In the representation of the verb form (at the beginning of this section), there are two sets of elements appearing in the 3^{rd} position as one TAM marker, since t_2 may combine with other TAM markers in this position. This phenomenon will be discussed below, in section 2.8.

Position 4. This position contains object and reflexive markers. Object marking is relatively limited in Langi, and is mostly, but not exclusively, reserved for beneficiaries. See for example (3), (38), (39), (81). The reflexive marker *-i*-, roughly speaking, indicates that the situation applies to the subjects themselves, and is widely encountered in Langi. See for example (20), (21), (74), (104).

Position 5. This position contains the verb radical. The most common structure in Langi is -CVC-, however many other structures are encountered, such as -VC-, -VCVC-, etc.

Position 6. This position contains extensions, which modify the valency of the verb. The most common are the applicative (3), (20), the passive (102), (106) and the causative (45). Several extensions may be present in a given verb form (20), (106).

Position 7. This position also contains tense-aspect-mood markers, which, in combination with those in the third position, determine the conjugation of the verb form. These will be described in detail below.

1.2 Other predicative elements. The other elements which enter into Langi conjugations, alone or as part of verbal constructions, are the copulas $-r_1$ and $-id_3a$ on one hand, and infinitival forms on the other hand. In this section I will present them as they are used independently, and in section 2 we will see how they enter into verbal constructions.

The copula -r1. The copula *-r1* always carries a subject prefix, and in certain cases is inflected for the perfect aspect. On its own, not in a verbal construction, it serves as predicative center with certain nominal determiners, certain types of spatial localizations, etc. In this case it is negated in the same manner as verb forms, i.e. with the pre-prefix *si*-.

(9)	ŋga	ırı ŋk	əndə	siirı t	oi tuku	'The dress is red, it isn't grey.				
	ŋ	-go	I	-rı	ŋkəndə	si	-I	-rı	bi	tuku
	ip9	-dress	sp9	-cop	red	p.pfx	-sp9	-cop	grey	neg

When the copula is followed by the connective na it denotes possession. In this case, when negated, the copula is replaced by the negative copula -si.

(10)	twaar	ı na	mpes	sa		'We had money.'			
	tu sp1pl	-a -pft	-п -cop	na conn	m ip10	-pes -mor	a ney		
(11)	twaas	si na	mpes	sa tuk	U		'We didn't have any money.'		
	tu sp1pl	-a -pft ·	-si -cop.r	ı neg o	na conn	m ip10	-pɛsa -money	tuku neg	

In verbal constructions, we shall see that *-r1* combines with infinitives and with inflected verbs.

The past copula -id3a. -id3a is a past tense copula. It always bears a subject prefix. On its own, it denotes a durative (as opposed to a punctual) situation in the past.

(12)	twiidʒa vumbi	'We used to be singers.'		
	tu -idʒa va -ımb -i sp1pl -cop ip2 -sing -sfx			
(13)	kwiidʒa kɪt∫aka aha ndʒi vi	'There used to be woods just right here.'		
	k∪ -idʒa kı -t∫aka aha sp17 -cop ip7 -woods dem16	ndʒi vi deict only		
(14)	sikwiidʒa kıt∫aka tuku	'There didn't used to be woods.'		
	si -ku -idʒa kı -t∫aka p.pfx -sp17 -cop ip7 -woods	tuku neg		

In verbal compounds, $-id_{3a}$ combines with inflected verbs. In some cases, it bears the pre-stem aspect marker -a- (see sections 2.16 and 2.18).

1.3 Infinitival forms. The infinitival forms found in verbal constructions show two structures, either *RAD-a* or kv-*RAD-a*, where kv- is the class 15 prefix which marks all verbo-nominal forms of verbs. The choice between the form with or without the class 15 prefix is based upon how closely bonded, semantically, the two verbal elements are. In Hadermann (1996: 159), it is mentioned that in the languages under study the infinitive sometimes lacks a prefix. She attributes this to morphological factors (whether or not the radical is vowel initial), but says the conditioning may also depend on the syntactic status of the verbo-nominal form within the utterance.

In Langi the difference is independent of the radical initial, and is fully distinctive. For example, there are two future tenses (which will be examined in detail in section 2.19), distinguished by the presence or absence of the class 15 prefix.

(15) sakaata turi 'We are about to hunt.'

sakaat -a tu -rī hunt -sfx sp1pl -cop

(16) kusakaata turi 'We will hunt.' (At some indeterminate future time.)

ku -sakaat -a tu -rī ip15 -hunt -sfx sp1pl -cop In (15) the form corresponds to an immediate future, something that is about to happen, and in (16) the form corresponds to an indeterminate future.

The same structural distinction is found for example when the infinitival form functions as direct object.

(17) voosaka vina 'They always want to dance / they adore dancing.' va -o -sak -a vin -a sp2pl -hab -want -hab dance -sfx
(18) voosaka kuvina 'They want to dance.' va -o -sak -a ku -vin -a sp2pl -hab -want -hab ip15 -dance -sfx

In example (17), wanting to dance is part of the subjects' personality, a permanent feature, whereas in (18) it is a passing fancy. We will see more examples of this phenomenon in section 2.10.

Now that we have had a look at the different elements involved, let us turn to the conjugations in which they are used.

2. Conjugations

Table 1 is a list of the conjugations which I have observed in Langi. Where applicable, negative forms are indicated by the pre-prefix in parentheses (si-) or, where the pre-prefix does not apply, by the independent form following the affirmative form(s) to which it applies.

It can be seen from this list that the Langi verbal system, like that of most Niger-Congo languages (Welmers 1973: 344), makes use of both 'simple' verb forms and verbal constructions. Several features however are atypical in comparison to Proto-Bantu or even to most East African Bantu languages. The most striking is the future formation /infinitive + auxiliary/ which runs counter to Greenberg's prediction (1966: 84) that languages where the usual word order is verb-object (VO) also tend to show the the word order /auxiliary + infinitive/.

Table 1: Langi Verb Conjugations

Form	Value	<i>Abbreviation</i> ⁴
(si-)sp-iyo-RAD-a	Progressive	prog
(si-)sp-o-RAD-a	Habitual	hab
(si-)sp-a-RAD-a	Perfect	pft
(si-)sp-a-RAD-ire	Perfective	pfv
(si-)sp-a sp-a-RAD-a	Anterior perfect	pft pft
(si-)sp-a sp-a-RAD-ire	Anterior perfective	pfv pfv
sp-ɛndɔ-RAD-a	Inevitable	inv
sp-tɔ-RAD-a	Decisional	dec
(si-)sp-ka-RAD-a ⁵	Narrative	nar
sp-rı (ku-)RAD-a ⁶	Narrative present	n/a
ku-RAD-a sp-a-п	Narrative past progressive	n/a
sp-a-rī sp-iyo-RAD-a	Dynamic past progressive	n/a
sp-a-гі sp-э-RAD-а	Dynamic past habitual	n/a
(si-)sp-idʒa sp-ɔ-RAD-a	Stative past habitual	n/a
sp-idʒa sp-a-RAD-a	Stative past perfect	n/a
sp-idʒa sp-a-RAD-a	Stative far past perfect	n/a
sp-a-idza (si-)sp-a-RAD-ire	Stative past perfective	n/a
(si-)sp-a-idʒa sp-a-RAD-irɛ	Stative far past perfective	n/a
(ku)-RAD-a sp-ri	Future	n/a
si-sp-п RAD-a	Negative future	n/a
RAD-a	Imperative 2sg	imp2sg
RAD-E	Imperative 1pl	imp1pl
RAD-i	Imperative 2pl	imp2pl
sp-RAD-ε	Subjunctive	sub
sp-ka-RAD-a	Advisory	adv
sp-ka-RAD-ε	Injunctive	inj
apa ku-RAD-a /	Negative imperative	n/a
ku-RAD-a tuku		

⁴ In compound forms, each component has its specific abbreviation (such as cop for 'copula').

⁵ This form is identical to the Advisory. They are only distinguished by the context in which they are used (narratives vs. direct speech), and not for example by tone: the tones are identical.

⁶ On the presence vs. absence of kv-, see the discussion of examples (15) through (18) above.

However, Langi is not alone in presenting such atypical word order. It is also found in several other Bantu languages, such as Mbugwe (F.34, Mous 2000, 2004), Gusii (E.42, Whiteley 1960), Kuria (E.43, Whiteley, 1955), and a number of languages from the zones B.40-B.50 and H.10-H.30 (Hadermann, 1996).

It is highly probable that the counter-universal structures in Langi were calqued on surrounding Southern Cushitic languages, the most likely candidates being Burunge and Alagwa (Mous, pers. com.). Further indication of influence from these languages is the fact that there are no clear distinctions between degrees of past in Langi.⁷ We will see in the following paragraphs that many conjugations can only incidentally be interpreted as past tenses, and that in most cases, the past is indicated by a distinct copula, which precedes the verb form. This situation is contrary to what is found in most Bantu languages (although comparable to what is found in Standard Swahili, Nurse 2000a: 249), as well as in Proto-Bantu, but closely resembles Southern Cushitic where aspectual distinctions are much more central than temporal ones (Nurse 2000b: 524). Furthermore, the only clear temporal distinctions are expressed in verbal constructions, whereas most aspectual and modal distinctions are expressed directly in the verb form.

2.1 Progressive: (si-)sp-iyo-RAD-a. The progressive indicates a process which is on-going at the time of speech. It is often used to signify that one is in the middle of doing something, and therefore unable to respond to a demand.

(19)	niiyəluusıka	'I'm talking'
	nı -iyə -luus -ık -a sp1sg -prog -talk -neut -prog	
(20)	<u>siiyəseya</u> kıntu tuku, <u>iiyəkirırıkana</u> 'He isn't saying anything, he's thinking.'	
	si -a -iyo -sɛy -a kı -ntu tuku p.pfx -sp1 -prog -say -prog ip7 -thing neg	
	a -iyo -k ⁸ -i -r -ır -ık -ar	1 -a
	sp1 -prog -k -refl -think -appl -neut -as	c -prog

⁷ Or at least in the dialect spoken in the towns of Kondoa and Pahi. I have been told by Nurse that Stegen has found several distinct pasts, however, as Stegen's aim is to establish a 'pan-Langi' orthography, he has not distinguished between the various dialects.

⁸ This consonant is inserted to avoid the merging of the two vowels. It is not part of the reflexive or the radical, the form in the infinitive is *kwiirmkana* 'to think'.

(21) vasinga <u>siviiyokivaa</u> tuku, <u>viiyobwnta</u>'The children aren't fighting, they're playing.'

va -siŋga si -va -iyo -k -i -va -a tuku va -iyo -bwiit -a ip2 -child p.pfx-sp2 -prog -k -refl -hit -prog neg sp2 -prog -play -prog

The progressive is never used with stative verbs. For a state current at the time of speech, either the habitual, the perfect or the perfective is used. For example the verb $k_{22}fa$ 'to be afraid' is usually conjugated in the habitual $(n_{22}k_{22}fa nd_{32}ka$ 'I am afraid of snakes / I fear snakes'), whereas the verb k_{UVIha} 'to be bad' is usually conjugated in the perfect $(n_{aavIha}$ 'I am bad').

2.2 Habitual: (si-)sp-o-RAD-a. The habitual is mostly used to describe situations that are characteristic of an extended period of time, to express recurrent events, statements of general truth. It is used for example to describe what one does in life, but, contrary to the progressive, does not stress that something is on-going at the time of speech.

The habitual shares many features with the perfect (see below), the difference being that the habitual does not refer to an earlier situation, and is mostly used with dynamic verbs, whereas the perfect is commonly used with both dynamic and stative verbs.

(22)	nəərima	'I farm / I'm a farmer.'
	nı -ə -rım -a sp1sg -hab -farm -hab	
(23)	nt∫uŋgula <u>yəərya</u> ndʒuu	'The hare eats beans.'
	n -t∫uŋgula 1 -ɔ -ry -a n - ip9 -hare sp9 -hab -eat -hab ip10 -	-dʒuu beans
(24)	sinootereka tuku 'I don't cook/ I'm not the one who cook	cs in our house.'
	si -nı -ɔ -tɛr -ɛk -a tuku p.pfx -sp1sg -hab -cook -neut -hab neg	

(25) <u>sivootema lukwi tuku</u>

'They are not cutting wood.'

si -va -ɔ -tɛm -a lu -kwi tuku p.pfx -sp2 -hab -cut -hab ip11 -wood neg

In the following example, a stative verb usually inflected in the perfective is inflected in the habitual, to show sarcasm.

(26) tootanga

'We know/we are knowing (we're not stupid).'

tu -ə -taŋg -a sp1pl -hab -know -hab

2.3 Perfect: (si-)sp-a-RAD-a. This form is certainly the reflex of the (tentative) Proto-Bantu 'preterite ipf.' $(-\dot{a}-...-a)$ (Meeussen 1967: 109), however, as its values are closer to what Comrie calls the perfect (Comrie 1976: 52-65), I have chosen the label which will have meaning for the largest number of readers.

The interpretation of the perfect is linked to the type of verb: stative vs. dynamic. When used with stative verbs, the time reference is the present, as in:

(27)	vit∫iŋgɔ <u>vyaabɔ</u>	<u>ha</u>	'The beads are beautiful.'		
	vi -t∫iŋgo vi ip8 -bead sp8	-a -bɔh -pft -be.bea	-a autiful -pft		
(28)	aya madzi <u>siyaa</u>	<u>ıhəla</u> tuku			'This water is not cold.'
	aya ma -dʒi dem6 ip6 -water	si -ya p.pfx -sp6	-a -hɔl -pft -be.cold	-a -pft	tuku neg

When used with dynamic verbs, the perfect refers to a present situation which results from a preceding process, the latter having produced a state which is either still current or the effects of which are still felt.

(29) mwaana <u>aakula</u>
'The child has grown.' (One deduces that he used to be short.)
mu -ana a -a -kul -a
ip1 -child sp1 -pft -grow -pft

(30) <u>naadoma</u> kaayii yaavo

'I have been to their house (already, therefore I don't want to go back).'

nı -a -dəm -a \varnothing -kaaya -i ı -a -və splsg -pft -go -pft ip9 -house -loc dp9 -det -poss3pl

(31) <u>sinaadoma</u> kaayii yaavo tuku'I haven't been to their house (but would like to go).'

si -nı -a -dɔm -a \varnothing -kaaya -i ı -a -vɔ tuku p.pfx -sp1sg -pft -go -pft ip9 -house -loc dp9 -det -poss3pl neg

2.4 Perfective: (si-)sp-a-RAD-ire. This conjugation can also be traced back to Proto-Bantu, however it is not clear whether it is the reflex of Meeussen's 'recent pf.' (-*a*-, $-id\hat{e}$) or his 'pret. pf.' (-*á*-, $-id\hat{e}$) (Meeussen 1967: 113), in neither case do the tones correspond (the pattern in Langi is usually $-\hat{a}$ -, $-id\hat{e}$, but may vary according to context).

The perfective denotes a completed situation. Contrary to the perfect, it does not refer to a past situation.

(32) mwaana <u>aakurire</u> 'The child grew.' mu -ana a -a -kul⁹ -ire ip1 -child sp1 -pfv -grow -pfv
(33) kuri <u>sidʒaalumire</u> musungaati tuku 'The dogs didn't bite the rich man.' Ø -kuri si -dʒi -a -lum -ire mu -sungaati tuku ip10 -dog p.pfx -sp10 -pfv -bite -pfv ip1 -rich.man neg
The perfective can be used with all types of past adverbs, 'yesterday' as well as 'last year'.

(34) id30, <u>naadomire</u> na ludzii'Yesterday I went to the watering hole.'

idzo nı -a -dom-ire na lu -dzi -i yesterday sp1sg -pfv -go -pfv conn ip11 -watering.hole -loc

⁹ /l/ is pronounced [r] before /i/.

(35) mwaaka <u>waalokire</u>, <u>twaadomire</u> na aru∫a 'Last year, we went to Arusha.'

mu -aka u -a -lok -ire tu -a -dom -ire na arusha ip
3 -year sp3 -pfv -pass -pfv sp1pl -pfv -go -pfv conn Arusha

One could be tempted to consider the perfective a 'past tense'. In my opinion, however, the emphasis is on the completed aspect rather than on the fact that it is past. This is supported by the use of the perfective in conditional phrases.

(36) koni <u>naadomire</u> kaayii, rya nderi
'If I go home, I will eat.' (lit. 'If I went home, I will eat'.)
koni ni -a -dom -ire Ø -kaaya -i ry -a nde -ri
if splsg -pfv -go -pfv ip9 -house -loc eat -sfx splsg -cop

2.5 Anterior perfect: (si-)sp-a sp-a-RAD-a. The anterior perfect has the same aspectual values as the perfect but refers to a situation which takes place prior to another situation, which is necessarily in the past. The structure of this form is unusual, as in Bantu verb forms a TAM marker must be prefixed to a radical, not to another TAM marker. According to my informants, the verb is simply "doubled".

(37) fatuma <u>aa aatumama</u> hantu ali uudzire
 'Fatuma was working when Ally came.'

Fatuma a -a а -a -tumam -a Fatuma sp3sg -pft sp3sg -pft -work -pft Ally a -ntu -udz ha -a -ire ip16 -place Ally sp3sg -pfv -come -pfv

(38) hantu viinukire, <u>sinaa naanda</u> ira kazi waampeere¹⁰ tuku
 'When they left, I hadn't started the work you gave me.'

¹⁰ This form is imbricated. Imbrication affects verbs inflected in the perfective, generally those bearing certain extensions, where, roughly speaking, the extension and the suffix merge. The outcome varies both according to the extension and to the radical final consonant. Other examples can be found in (41), (70), (71). For more information on imbrication, see Bastin (1983) and Dunham (forthcoming).

ha -ntu -a -i -nuk -ire si -a -and -a va -nı -a nī ip16-place sp2pl -pfv -refl -leave -pfv p.pfx -sp1sg -pft splsg -pft -begin -pft Ø-kazi u -a -N -heer -ire tuku -ra T dp9 -dem ip9 -work sp2sg -pfv -obj1sg -give -pfv neg

(39) hayi vaari ava vasinga? sinaa naavoona tuku'Where were the children? I hadn't seen them.'

va -a -rī va -singa havi ava where sp2 -pft -cop dem2 ip2 -child tuku si -a -a -va -on -a -nı nī p.pfx -splsg -pft splsg -pft -obj2-see -pft neg

A possible explanation is that these structures, as well as those presented in the following section, at one point contained an auxiliary which has since been deleted.

2.6 Anterior perfective: (si-)sp-a sp-a-RAD-ire. The anterior perfective has the same aspectual values as the perfective but, as for the anterior perfect in the preceding section, refers to a situation which takes place prior to another, past situation.

(40) hantu <u>twaa twaaturire</u> ma ali akaloka'When we appeared, then Ally left (lit. passed).'

ha -ntu -a -tul -ire tυ -a tυ ip16 -place sp1pl -pfv sp1pl -pfv -appear -pfv ali -lok -a ma а -ka then Ally sp3sg -nar -pass -nar

(41) twaa twiiſεεnε, ma nıkaanda tudʒa na idukii
 'We had met, then (after which) I started running off to the store.'

-i tπ -a tπ -a -ſi -an -ire splpl -pfv sp1pl -pfv -refl -meet -rec -pfv -ka -and -a -tııdz -a i -duki -i ma nı na then splsg -nar -start -nar -run -sfx conn ip5 -store -loc **2.7 Inevitable:** sp-endo-RAD-a. Verbs in this conjugation refer to a situation which the speaker considers will happen in the near future and which is inevitable. The marker $-\varepsilon ndo$ - is probably a grammaticalized form of the verb kweenda 'want, love'. This is quite frequent in Bantu languages, to the point that Heine (1997: 2) includes it in his catalogue of probabilistic predictions that can be made about African languages: "(v) If a language develops a future tense marker then most likely it will use either of the motion verbs 'go (to)' or 'come (to)' or a verb of volition 'want'."

As will be seen from the following examples, Langi has retained the notion of volition, but in the sense that when something 'wants' to do something, it means that that something is bound to happen. This type of value is found for example in Moore (Raphael Kaboré, pers. com.), a Gur language spoken in Burkina Faso, where when one says 'it wants to rain', it means that the clouds are so full that it is bound to rain, or 'the glass wants to fall' meaning that if the glass is not moved it will fall from the table.

(42) mpaka mnenya tundopata habari 'All the way to Mnenya we are bound to get the news.' mpaka mnɛnya tu -ɛndɔ -pat -a Ø -habari -get -sfx ip10 -news Mnenya sp1pl -inv until (43) ma ha mpit∫i yɛɛndɔkwatwa 'Then, here, the hyena is going to get caught.' ma ha m -pit∫i 1 -endo -kwat -w -a then here ip9 -hyena sp9 -inv -touch -pass -sfx kıntu <u>mu</u>ndərya nı waarı (44)'What you are going to eat is porridge.' (Whether you like it or not.) mu - ϵ ndo¹¹ -rv -a kı -ntu nτ Ħ -arı ip7 -thing sp2pl -inv -eat -sfx pres ip14 -porridge (45) akafumya ngo dʒaat∫wɛ na <u>ɛndokoowa</u> 'She took off her clothes (in order) to wash.' -ka -fum -y -a dʒi -a -t∫wε ŋ -go na а

 $^{^{11}}$ Preceeded by the vowel /u/, the marker's initial / $\epsilon/$ drops.

spl -nar -exit -caus -nar ip10 -clothes dp10 -det -poss3sg conn a -endo -k -ow -a spl -inv -k -wash -sfx

2.8 Decisional: sp-to-RAD-a. It is highly probable that $-t_{2}$ - is a grammaticalized form of a full verb (Güldemann 2003: 185), as is the case for $-end_{2}$ -. Contrary to $-end_{2}$ - however, it is difficult to say which full verb it stems from. Language-internally, the only likely source is kutoola 'take'. I have been unable to find any examples of other Bantu languages where 'take' has been grammaticalized, how-ever the phenomenon is widely attested in the languages of the world (Hagège 1975, Sebba 1987, Li and Thompson 1974, Lord 1993, Ozanne-Rivierre 2004). Contrary to what is found in Langi, in most of the cases described, the verb 'take' is serialized and tends to undergo a gradual reanalysis as a preposition or a case marker. The one example I have found that is somewhat similar to Langi is in the Polynesian Outlier language, Pileni. According to Åshild Næss (2004: 242), two constructions use the verb *toa* 'take', with different semantic and syntactic properties: one where *toa* introduces an object argument, and another where it contributes a volitional or inceptive meaning to the clause, similar to the English 'to take to V-ing'.

One other possibility is that the form derives from a verb 'to leave' or 'to go' (in Langi *kv-tamanya* and *kv-doma* respectively) followed by an infinitive. Botne (1999: 484) mentions two Bantu languages where a marker *-to-* is found. In the first case it is derived from 'to leave, to go', in *to-kos-e* 'va prendre' (Ntomba C.66: Gilliard 1928), and in the second case (Lozi S.34: Gorman 1950) it is derived from 'come' + *ku*-inf: *mu-to-ng-a* 'come (Pl) and get (it)'. Hadermann (1999: 454-455) mentions one case where a marker *-too-* is found, and also posits its origins in a verb 'to go': "En nkengo (C.61), une des formes du futur comporte la marque *-too-* qui pourrait refléter une séquence ancienne *-ta-ko-* où *-ta-* remonterait à un verbe 'aller'."

In the examples I have found in Langi, the use of *-to-* signifies that the speaker is about to/has decided to/intends to undertake an action voluntarily.

(46) aakwiire, <u>tutosongola</u> mayiti yaat∫we

'He is dead, we are going to remove his corpse.'

a -a -kwi -irɛ tu -tɔ -sɔŋgɔl -a Ø -mayiti 1 -a -t∫wɛ sp1 -pfv -die -pfv sp1pl -dec -remove -sfx ip9 -corpse dp9 -det -poss3sg (47) nkukulume <u>itoveka</u> vi, keyi kakatoola

'Just when the rooster was about to crow, he (small bird) took (clothes).'

n -kuku -lume 1 -to -vek -a vi keyi ka -ka -tool -a ip9-chicken -male sp9 -dec -crow -sfx just then sp12 -nar -take -nar

A particularity of the decisional marker is that it can follow other TAM markers in the 3^{rd} position of the verb form, for example, the following:

Habitual + Decisional. When the decisional is combined with the habitual marker, the speaker indicates that a decision has been taken, and that it is in the process of being carried out.

(48) <u>nootokoowa</u> ludzii

'I'm going to wash in the watering hole.' (Despite contrary orders.)

nı -ɔ -tɔ -k -ɔw -a lu -dʒi -i sp1sg -hab -dec -k -wash -hab ip11 -watering.hole -loc

Perfect + *Decisional*. Here the speaker indicates that a decision was taken, and has already been carried out:

 (49) naatokoowa 'I still went and took a bath.'
 nI -a -to -k -ow -a splsg -pft -dec -k -wash -pft

Progressive + *Decisional.* Here, too, the use of the marker $-t_{2}$ - seems to add a level of intention or decision to the basic verb, but, as the form is in the negative, to indicate that the decision is not in the process of being carried out.

(50) mbula yoovaa, ava vadala <u>siviiyotorima</u> tuku 'It's raining, the women are not going to farm.'

> m -bula I -ɔ -va -a ip9 -rain sp9 -hab -beat -hab ava va -dala si -va -iyɔ -tɔ -rım -a tuku dem2 ip2 -woman p.pfx -sp2 -prog -dec -farm -prog neg

Inevitable + *Decisional.* When $-\varepsilon nd_2$ - and $-t_2$ - are combined, the resulting values are inevitability and intention/decision.

(51) no ut jiku <u>yεεndotorya</u> ndʒuu 'It's at night (that) he goes and eats beans.'
 (Even though he knows he's bound to be caught).

no u -t∫iku 1 -endo -to -ry -a n -dzuu rel ip14 -night sp9 -inv -dec -eat -sfx ip10 -bean

(52) hapana <u>kundətəkəwa</u>

'Do not go and wash anyway.' (Despite our orders).

hapana ku -endo -to -k -ow -a neg ip15 -inv -dec -k -wash -sfx

From these examples it can be seen that the inevitable $-\epsilon nd_{2}$ - and the decisional $-t_{2}$ -, show rather atypical behavior. This leads me to hypothesize that they entered the Langi tense/aspect/mood system more recently than the other markers we have seen thus far, such as the anterior or the progressive, for example.

The first distinctive feature of $-t_2$ - and $-\varepsilon nd_2$ - is that the corresponding full lexical verbs are still in common use (with reservations however concerning the origins of $-t_2$ -, see above).

(53)	<u>təəla</u>	idihwa			'Take some white sorghum.'
	təəl take	-a -imp2sg	i ip5	-dihwa -sorghum	

(54) simba kıntu yeenda nı mahonge 'The lion, what he loves is steak.'

 \varnothing -simba ki -ntu i -a - ε nd -a ni ma -h ε ng ε ip9 -lion ip7 -thing sp9 -pft -love -pft pres ip6 -steak

I would further hypothesize that $-t_{2}$ - was grammaticalized before $-\varepsilon nd_{2}$ - as the latter, in its full form, is still attested as the first verb in a verb combination. For example.

 (55) mpit∫i dʒɛɛnda kusakaata 'Hyenas love to hunt.'
 m -pit∫i dʒi -a -ɛnd -a ku -sakaat -a ip10 -hyena sp10 -pft -love -pft ip15 -hunt -sfx Furthermore, to can combine with other markers in the 3rd column. In my opinion, the value common to these two verbs, *kwɛɛnda* 'want, love' and *kutɔola* 'take' which led to their grammaticalization, and which was lacking in the Langi verb system, is that of showing the speaker's involvement, or attitude towards the state of affairs expressed in the clause. This type of 'attitude' marker is described by Maarten Mous for Ma'a (2003: 116, 129, 131). A further similarity between the markers mentioned by Mous and the Langi -to- is their possibility of combining with other pre-radical TAM markers.

2.9 Narrative: (si-)sp-ka-RAD-a. This conjugation is identical in structure to what I have dubbed the 'advisory' (see below, section 2.22). A similar state of affairs was attested by Meeussen (1967), who mentions three distinct conjugations with -ka- (motional, inceptive and subsecutive) and notes (p. 109): "The relationship between $-k\dot{a}$ -, -ka- and an often attested near homophone formative with meaning 'and (he did...)' (subsecutive) is not clear."

In Langi, the narrative and the advisory are never used in the same speech context: the narrative, as its name indicates, is only used in narration, whereas the advisory, and the injunctive (with the suffix $-\varepsilon$) is always addressed to someone, even if the advice concerns a third person.

The narrative is never used as the first verb form in a text, the time frame always having been previously established, either by the traditional introduction *ahs kale* 'Once upon a time...' or by a preceding verb form. Once the time frame is established, successive events are presented in the narrative, most often separated by *ma* 'then'.

(56)	ma <u>akadoma</u> na ludzii						'The	'Then she went to the watering hole.'				,
	ma then	a sp1	-ka -nar	-dəm -go	-a -nar	na con	lu n ip11	-d3i -water	-i -loc			
(57)) ma <u>ıkawuluka</u> na dʒira ŋgɔ						'The	n it flev	w aw	/ay v	with those clothe	s.'
	ma then	ı sp9	-ka -nar	-wul -fly	-uk -sep	-a -nar	na conn	dʒi -1 dp10-c	ra lem	ŋ ip10	-go 0-clothes	

Verbs inflected in the narrative are usually translated by a preterit in English, but in Langi it cannot be considered a 'past tense', as by itself it carries no reference to time, and is dependent on a separate form. In this respect, the narrative is comparable to the 'aorist' described by Guentchéva (1990: 107) for Bulgarian, and Robert (1996) for Wolof. Robert states:

L'Aoriste est une forme verbale qui n'indique pas de repérage temporel. (...) Les événements à l'Aoriste n'ont pas par eux-mêmes ni lieu ni temps ; ils s'inscrivent dans un cadre situationnel prédéfini à l'aide d'une autre conjugaison. [p. 377]

2.10 Narrative present: sp-r1 (ku-)RAD-a. In this construction, the verb radical following the copula may or may not carry the infinitive prefix kv-, depending on how closely, semantically, the subject is bound to the lexical verb. In example (58), the lion is coming of his own free will — he could choose not to — thus the verb carries the infinitive prefix. In example (59), the hare is struggling to get by and can do nothing about it, thus the prefix is absent. See also examples (15)-(18).

This conjugation is only attested in narratives. It is used when there is a break in the narration, either in time or in space.

(58) haha simba <u>III kuud3a</u> no jiana mpitji'Now the lion is coming to meet the hyena.'

haha \emptyset -simba 1 -r1 ku -udz -a nɔ $\int ian -a m -pit \int i$ now ip9-lion sp9-cop ip15 -come -sfx rel meet -sfx ip9 -hyena

This example is taken from a story.¹² The preceding sentences present the lion and the hyena, and the narrative present is then used to signify that the speaker has finished the introduction and is entering the narrative present.

 (59) ntʃuŋgula <u>Irī kwɛta</u> 'The hare is struggling (to get by).'
 n -tʃuŋgula I -rī kwɛt -a ip9 -hare sp9-cop struggle -sfx

Here too, the hare has been introduced as part of the story in the preceding sentences, example (59) serving to describe the state the hare is in, in the narrative present.

¹²This story and others may be read and listened to on the LACITO Archive Project website: http://lacito.vjf.cnrs.fr/archivage.

2.11 Narrative past progressive: ku-RAD-a sp-a-ri. Unfortunately I have only one example of this construction in my data, and it is elicited,¹³ but it is also mentioned in Nurse (2003: 97). There is further mention of this same construction, in this order (infinitive + copula) as well as in the inverse order (copula + infinitive) in Mous (2000: 475, however his examples were given by Nurse).¹⁴ According to Mous, the change in word order is probably due to emphasis (pers. com).

(60) kuseka twaari ma uhu maka akaanda rira 'We were laughing then this guy started to cry!' ku -sek uhu Ø -maka -a tu -a -m ma ip15 -laugh -sfx sp1pl -pft -cop then dem1 ip1a-guy -ka -and -a а rπ -a sp1 -nar -start -nar cry -sfx

Despite the scarcity of data, I have given this example as it shows a counteruniversal word order (infinitive + copula) — see the general introduction to section 2).

2.12 Dynamic past progressive: sp-a-r1 sp-iyo-RAD-a. This structure is also only attested in elicitation, in one example. However, it confirms that the copula *-r1* is largely productive in verbal constructions, both with and without the perfect marker, and both preceding and following the lexical verb.

(61) twaari twiiyoseka 'We were (in the process of) laughing.'

tu -a -rı tu -iyə -sɛk -a sp1pl -pft -cop sp1pl -prog -laugh -prog

2.13 Dynamic past habitual: sp-a-r1 sp-o-RAD-a.

(62) twaarı təəsɛka 'We were laughing.'
tu -a -rı tu -ə -sɛk -a splpl -pft -cop splpl -hab -laugh -hab

¹³It must be noted that this is a common problem when one works on a language that has not been previously described.

¹⁴There seems to be some confusion in the translation in Mous ('We are buying'), which should read as it appears in Nurse (2003) 'We were laughing'.

From examples (60)-(62), it seems that the copula inflected in the perfect situates a process in the past. We will see in the following sections that when the lexical verb is stative, as opposed to dynamic, the copula $-id_3a$ is used.

2.14 Stative past habitual: (si-)sp-id3a sp-3-RAD-a. This construction is made up of a verb inflected in the habitual preceded by the copula $-id_3a$, which serves to place the situation in the past. The aspectual values of the main verb are kept.

(63) twiidza tookimba hantu tooyenda, haha tootiidza 'We used to sing while walking, now we run.' -э -k -ımb -a tυ -idza tu ha -ntu tu -vend -a -0 splpl -cop splpl -hab -k -sing -hab ip16 -place splpl -hab -walk -hab haha tu -0 -tııdz -a splpl -hab -run now -hab (64) twiidza tookikala Kondoa ma tukasaama na London 'We used to live in Kondoa, then we migrated to London.' -ikal -a Kondoa -idza -k tu tu -0 -reside -hab Kondoa sp1pl -cop splpl -hab -k ma tu -ka -saam na London -a splpl -nar -migrate -nar conn London then hantu nookikaala amerikani¹⁵ siniidza nootumama benkii tuku (65) 'When I lived in America I didn't use to work in a bank.' -k -ikaal amerika -ni ha -a -ntu nτ -0 -reside -hab America -loc

ip16 -place sp1sg -hab -k -reside -hab America -loc
si -ni -id3a ni -> -tumam -a Ø -benki -i tuku
p.pfx -sp1sg -cop sp1sg -hab -work -hab ip9 -bank -loc neg

2.15 Stative past perfect: sp-id3a sp-a-RAD-a. In this construction the copula is combined with verbs inflected in the perfect aspect. The latter keeps its aspectual values, but is placed in the past.

¹⁵ This locative suffix is borrowed from Swahili.

(66)	<u>niidza neenda</u> tudza na t \int uuri 'I used to like to run in the morning.'										
	nı -idʒa nı -a -ɛnd -a tɪɪdʒ -a na kı -uri sp1sg -cop sp1sg -pft -love -pft run -sfx conn ip7 -morning										
(67)	<u>mwaasu wiid3a</u> waavarika 'When the sun was shining.' (in response to: 'When did you usually eat?')										
	mu -asu u -idza u -a -var -1k -a ip3 -sun sp3 -cop sp3 -pft -shine -neut -pft										
(68)	ıra siku <u>iidʒa yaabəha</u> 'That day was beautiful.'										
	ıra Ø -siku ı -idʒa ı -a -bəh -a dem9 ip9 -day sp9 -cop sp9 -pft -be.beautiful -pft										

2.16 Stative far past perfect: sp-a-id3a sp-a-RAD-a. This is one of the rare cases where one finds distinctions between degrees of pasts in Langi. The only other case is shown below (section 2.17 vs. 2.18). Unfortunately I have only one example of this conjugation, and no examples where this conjugation is in the negative.

(69) wookati naadza mududi, <u>naadza naadudya</u> sana'When I was small, I was very small.'

u -oka	ti	nı	-a	-idʒa	mu -	dudi	
ip14 -mon	nent	splsg	-pft	-cop	dpl -:	small	
nı	-a	-dʒa	nı	-a	-dudi	-a	sana
splsg	-pft	-cop	spls	sg -pft	-be.smal	l -pft	very

2.17 Stative past perfective: sp-id3a (si-)sp-a-RAD-ire. Verbs in the stative past perfective refer to situations which were current in the past.

70)	(70)	<u>wiidza</u>	waatee	<u>te</u> ¹⁶ wa	ami	'You used to have a corral				
		U an2ag	-idza	U sp2sg	-a	-tɛt	-ire	U in14	-ami	
		sp2sg	-cop	sp2sg	-a -pfv	-have	-ne -pfv	ip14	-corral	

¹⁶ This verb is defective, appearing only in the perfective, where it shows an irregular form of imbrication. A comparable verb for 'have' is found in Chaga (M.-L. Montlahuc, pers. com.), where it is also defective (although to a lesser extent than in Langi).

(71) <u>niidza siniifyɛɛnɛ</u> na mama wiitu tuku 'I used to not look like our¹⁷ mother.'

nı	-id3a	a si	-nı	-a	-i	-fy	-an	-ire
splsg	-cop	p.pfx	-sp1	sg -pi	fv -refl	-ressemble	-asc	-pfv
na	Ø	-mama	U	-a	-itu	tuku		
conn	ip1a	-mother	dp1	-det	-poss1p	l neg		

We shall see in example (74) below that in the stative far past perfective, it is the copular verb which carries the negative pre-prefix.

2.18 Stative far past perfective: (si-)sp-a-idza sp-a-RAD-ire.

(72) <u>vaadza vaateete</u> udzusi wooruta¹⁸

'They used to be blacksmiths.' (lit. 'They used to have the profession of pulling.')

va -a -idza va -a -tɛt -irɛ u -dzusi u -ɔ -rut -a sp2 -pft -cop sp2 -pfv -have -pfv ip14 -profession pp14 -rel -pull -rel

(73) uhu musinga hantu aadʒa mududi, <u>aadʒa iifyɛɛnɛ</u> na iyo waavo 'When this child was small he looked like their mother.'

uhu mu -singa ha -ntu a -a -idʒa mu -dudi, a -a -idʒa dem1 ip1 -child ip16-place sp1-pft -cop dp1 -small sp1 -pft -cop a -a -i -fy -an -irɛ na Ø -iyɔ u -a -vɔ sp1 -pfv -ref1 -resemble -asc -pfv conn IP1a -mother dp3 -det -poss3pl

(74) ava vasinga hantu vaadza vadudi sivaadza viifyeene na iyo waavo tuku
 'When these children were small they didn't used to look like their mother.'

ava va -siŋga ha -ntu va -a -idʒa va -dudi si -va -a -idʒa dem2 ip2 -child ip16 -place sp2 -pft -cop dp2 -small p.pfx -sp1 -pft -cop va -a -i -fy -an -irɛ na Ø -iyɔ u -a -vɔ tuku sp2-pfv -refl-ressemble -asc -pfv conn IP1a -mother dp1 -det -poss3pl neg

¹⁷ Family members (and homes) are never possessed in the singular in Langi.

¹⁸ Relative forms apparently share the same TAM markers as the habitual, only the prefix differs (the pronominal prefix is used as opposed to the subject prefix).

As can be seen from the examples in the preceding five sections, in accordance with its use independently, verb constructions with the past copula always refer to durative as opposed to punctual situations.

2.19 Future: (ku-)RAD-a sp-ri. This form denotes either an immediate or an indeterminate future, depending on the presence of the class 15 prefix ku-. I have grouped the two forms together for practical reasons: in several cases, such as in the negative or the interrogative, only one form is possible, so presenting the two forms together eliminates the need to go back and forth between sections.

The prefixless form denotes an immediate future, something that is about to happen.

(75)	neya arı	'He is about to sleep.'
	леу -a a -гі sleep -sfx sp1 -cop	
(76)	mutı <u>wiya urı</u>	'The tree is about to fall.'
	mu-tı wiy-a u -ı ip3 -tree fall -sfx sp3 -c	л cop

The prefixed form is used to denote an indeterminate future.

(77)	muti <u>kuw</u>	<u>iya uri</u>	'The tree will fall (some day				
	mu -tı	ku -wiy-a	U	-ri			
	ip3 -tree	ip15-fall -sfx	sp3	-cop			

The prefixed form is also used whenever a time or place is mentioned.

(78)	<u>kuke</u>	ra turi	mut	ı lamu	təndə	'We will cut the tree tomorrow.				
	ku ip15	-kɛr -cut	-a -sfx	tu sp1pl	-п -cop	mu ip3	-tı -tree	lamutəndə tomorrow		
(79)	<u>kudə</u>	ma ar	<u>1</u> na c	lədəm	a	•	He w	vill go to Dodoma.'		
	ku ip15	-dəm -go	n -a -sfx	a spl	-rı -cop	na conn	dod: Dode	oma		

(80) mwaarimu <u>kuvasomera ari</u> vasinga kitabu Julii
'The teacher will read a book to the children at school.'

mu-arımu ku -va -səm -ɛr -a a -rı va -siŋga ipl -teacher ip15-obj2 -read -appl -sfx spl -cop ip2 -child kı -tabu Ø -∫ulɛ -i ip7 -book ip9 -school -loc

In conditional phrases, the prefixless form is used in the apodosis, probably to indicate that the action/event/state will immediately follow the fulfillment of the condition.

(81)	kənı	naado	mire	kaayi	i, <u>rya ı</u>	"	'If I go home, I will eat.'					
	kənı ifsp1	ni -a sg -p	0 fv -g	lom · 30 ·	-irε ∙pfv	Ø −kaa ip9 −hou	iya se	-i -loc	ry -a eat -sf	nda x spl	e - sg -	rı cop
(82)	kəni	wiimt	oirε, n	u <u>vina</u>	nderi		"	If yo	u sing	, me,	I wil	l dance.'
	konı if	U sp2sg	-a -pfv	-1mb -sing	o -ire -pfv	nı pers1sg	vir dar	1 nce	-a -sfx	nde sp1sg	-rı -cop	0

However the prefixed form is used when there is an object marker in the verb form, certainly in part because of possible confusion between the object prefix and the infinitive prefix (in the following example, both kv-).

(83)	kənı	wiimb	oire <u>l</u>	<u>cukuv</u>	inıra ı	nderı	'If you sing, I will dance for you.					
	kənı	U	-a	-ımb	-irɛ	ku	-ku	-vin	-ır	-a	ndɛ	-m
	if	sp2sg	-pfv	-sing	-pfv	ip15	-obj2sg	-dance	-appl	-sfx	splsg	-cop

To the best of my knowledge, this constituent order (verb + copula) is limited to a very small number of Bantu languages. Among these languages, I believe a distinction must be made between those which accept SOV order and those with strict SVO order. In the first category are found languages from Guthrie's zones B.40-50 and H.10-H.30. These have been studied by Hadermann (1996) who states:

La deuxième construction que nous avons analysée est celle où l'infinitif est antéposé à un verbe auxiliant. Dans cette structure l'infinitif est généralement introduit par le préfixe locatif mu- et l'ensemble traduit l'aspect progressif du procès verbal en question. Donc, contrairement à ce qui se passe dans les constructions à verbe redoublé, le morphème locatif (ou sa trace) propre au temps du progressif (cf. la reconstruction *PV-di-mu-NV) n'apparaît pas dans la forme verbale conjuguée mais dans la forme nomino-verbale antéposée. Cette caractéristique est un signe du fait que la structure 'infinitif + auxiliant' est issue de la séquence 'auxiliant + infinitif' mais à un moment où celle-ci pouvait encore s'interpréter en 'verbe + complément_[nomino-verbal]'. Puis, il y a eu antéposition du complément, ceci probablement pour des raisons d'emphase. De nouveau, l'existence de l'ordre SOV dans les langues en question a sans doute facilité l'antéposition du locatif infinitival. Au moment où la structure emphatique '(S) Loc-Inf Aux' devient une structure non marquée, un processus de grammaticalisation peut se déclencher et le locatif infinitival sera réinterprété en noyau verbal, suivi d'un verbe-auxiliant 'être'. [p. 167]

The second category are languages with strict SVO order which nevertheless show constructions where the infinitival form precedes the auxiliary. In this category are found for example the closely related Gusii (E.42) and Kuria (E.43). In Gusii, the forms with an auxiliary (the copula *-re*) following a main verb correspond to what I call a narrative present (Whiteley 1960: 57) (tones not marked).

(84) nkogenda¹⁹ nde boono korigia embori
 'Now I'm going looking for the goat.'

When the auxiliary carries the verbal prefix -a, the form corresponds to what I call a narrative past progressive.

(85) nkorema naare boono ngotimoka nde'I've been hoeing and now I'm resting'

In Kuria, the forms with an auxiliary following the main verb correspond to something between what I call present progressive and habitual (Whiteley 1955: 92).

¹⁹ The class 15 prefix is ko-, these forms almost certainly bear the cliticized focus marker n(i)-(Nurse and Muzale 1999).

(86)	ŋgokəra are emeremo kira urusiku	'He's working every day.'
(87)	ŋkorema nde bono ŋkumunya nde	'I'm hoeing and am resting

In Mbugwe (F.34) the situation is slightly different. All the tenses that make use of an auxiliary in combination with an infinitive show the order verb-auxiliary, in other words, the order never varies, contrary to what is found in Gusii, Kuria and Langi. In Mbugwe, three different auxiliaries may follow the main verb: the present progressive *-kende*, the future *-je* (which is the verb *ja* 'come' conjugated in the optative, marked by the suffix *-e*) and the habitual *-anda* (Mous 2000: 471).

ngulu

9:force

'The rain falls with force.' (88)mbula o -tova e -kende na 15 -rain 9 -pres.prog with 9:rain 'We will eat cassava.' (89) ora ko -ie mohogo 15:eat 1pl -come:sbj 3:cassava (90) 'Do you eat fish?' nsive? -anda ora w 15:eat 2sg -hab 9:fish

One feature shared by Gusii, Kuria, Mbugwe and Langi is that they have long been surrounded by communities speaking non-Bantu languages, where the object precedes the verb (Nurse 2000b: 525-6), which seems to indicate that this phenomenon is areal (Dahl 2001: 1456 who states 'Areal linguistics is traditionally concerned with similarities between geographically contiguous languages, in particular when they cannot be ascribed to a common proto-language.'). Furthermore, the adoption of these structures may well have been facilitated by internal factors. In Langi, the order *infinitive* + *copula* is only attested in main clause affirmatives, and in all other cases the order *copula* + *infinitive* is found.

(91) nadi <u>uri doma</u> 'When will you go?'

nadi u -rı dəm -a when sp2sg -cop go -sfx

(92)	ŋɔmbɛ siirı nwa madʒi yɔɔsi tuku 'The cow will not drink all the water.'						
	Ø- ŋɔmbɛ si -ı -rı nw -a ma -dʒi ya- ɔsi tuku ip9- cow p.pfx -sp9 -cop drink -sfx ip6 -water dp6- all neg						
(93)	ni mwaarimu no <u>ari soma</u> kitabu 'It is the teacher who will read a book.'						
	nı mu -arımu nə a -rı səm -a kı -tabu pres ipl -teacher rel spl -cop read -sfx ip7 -book						
(94)	<u>kuudʒa arı</u> konı <u>arı rɛɛta</u> t∫aakurya 'He can come if he brings food.' (lit. 'He will come if he is bringing food.')						
	ku -udʒ -a a -rī konī a -rī rɛɛt -a kī -a -ku -ry -a ip15-come -sfx sp1 -cop if sp1-cop bring -sfx ip7 -det -ip15 -eat -sfx						
(95)	hantu <u>uri turira, kuni∫ana uri</u> naalokirε 'When you appear, you will find me gone.' (lit. 'When you will appear')						
	ha -ntu u -rı tul -ır -a ip16 -place sp2sg -cop appear -appl -sfx						
	ku -nı -∫an -a u -rı nı -a -lək -irɛ ip15 -obj1sg -find -sfx sp2sg -cop sp1sg -pfv -leave -pfv						

We also saw that constituent order in Langi can change in other circumstances, namely for reasons of emphasis, such as in example (60).

2.20 Imperatives: RAD-sfx. Imperatives are characterized by the absence of both subject prefixes and pre-radical TAM markers. There are three affirmative imperative forms in Langi, distinguished by their suffixes.

(96)	2sg	dəma dəm go	-a -imp2sg	'go!'	ımba ımb sing	-a -imp2sg	'sing!'
(97)	lpl	palɛ		'let's count!'	saire		'let's taste!'
		pal count	-E -imp1pl		sa taste	-Ir -appl	-E -imp1pl

(98)	2pl	rɛki mburi		'leave the goat!'	iti	'pour!'	
		rek -i	m	-buri	it	-i	
		leave -imp2pl	ip9	-goat	pour	-imp2pl	

There are three negative imperative forms, apparently interchangeable: *apa ku-RAD-a*, *hapana ku-RAD-a* or *ku-RAD-a tuku*. These forms are impersonal, equivalent to the English 'no V-ing' or 'do not V'. They are composed of a verb in the infinitive preceded by either *apa* or *hapana* (the latter form is borrowed from Swahili and appears to be replacing the former). The third possible negative form is composed of a verb in the infinitive accompanied by the adverb *tuku* placed at the end of the sentence.

(99) apa kutudza 'No running.' apa ku -tudz -a neg ip15 -run -sfx
(100) hapana kuyenda ndzirii 'No walking on the path.'

hapana ku -yend -a n -d31ra -i neg ip15 -walk -sfx ip9 -path -loc

(101) taata anteheere kurima kidundii tuku 'Papa told me: do not farm in the hills.'

 \varnothing -taata a -a -N -t ε h - ε r -ir ε ku -rim -a ki -dunda-i tuku ipla -father spl -pfv -objlsg -listen -appl -pfv ip15-farm -sfx ip7 -hill -loc neg

2.21 Subjunctive: sp-RAD- ε . The subjunctive mood is marked by a \emptyset TAM marker in the 3rd position, combined with the suffix - ε . Contrary to the imperative, the verb bears a subject marker in the 2nd position. This mood is used to express wishes, orders, obligations, etc. Whereas the imperative is necessarily addressed to someone, this is not the case for the subjunctive.

(102)	uloolv	NЕ			'that you marry/ you should marry, I want you to get married'				
	U sp2sg	-ləəl -marry	-w -pass	-E -sub					

```
(103) adzenge numba
                              'that he build a house'
         -dzeng -e
                      n -umba
      a
      sp1 -build -sub ip9 -house
(104) tiilaire kirume
                              'that we show each other our magic'
                 -la
            -i
                                  kı
                                      -rume
      fπ
                       -tr
                             -8
      sp1pl -refl -show -appl -sub ip7 -magic
(105) mumpeere madzi
                              'that you give me water'
           -N
                     -heer
      mu
                            -8
                                  ma -dzi
      sp2pl -obilsg
                    -give
                            -sub ip6 -water
```

The subjunctive mood does not have its own negative form, and instead shares that of the imperative. Furthermore, many negative subjunctive forms are rendered by verbs with negative meaning, such as *kudira* 'to not go'.

(106) adiriwe 'that she be made to not go'

a -dir -i -w -ɛ spl -not.go -caus -pass -subj

2.22 Advisory: sp-ka-RAD-a. This form bears what is commonly called a distal marker, -ka-, in the 3rd position. This marker is thought to derive from a verb meaning 'go' (see Botne 1999 for a comprehensive study of the marker -ka-). In Langi, the distal imparts the notion of movement, even though the direction is not always itive. These forms are used to denote advice or weak orders.

(107) <u>tukarɛɛta</u> mpɛmbɛ ma <u>dʒɪkavıkırwa</u> ıra myooda
'We must bring horns, and the horns must be filled with potion.'
tu -ka -rɛɛt -a m -hɛmbɛ ma dʒɪ -ka -vɪk -ır -w -a sp1pl -adv -bring -adv ip10 -horn then sp10 -adv -put -appl -pass -adv I -ra mı -oda dp4 -dem ip4 -potion

(108) <u>ukatahira</u> madʒi vi, <u>ukuudʒa</u> na kaayii 'Just go and fetch water, and come back home.'

u -ka -tah -ır -a ma -dʒi vi u -ka -udʒ -a
 sp2sg -adv-fetch -appl -adv ip6 -water only sp2sg -adv -come -adv
 na Ø -kaaya -i
 conn ip9 -house -loc

(109) <u>vakasakwa</u> vaosi 'Elders must be found.'

va -ka -sak -w -a va -osi sp2 -adv -find -pass -adv ip2 -elder

2.23 Injunctive: sp-ka-RAD- ε . Combining the distal marker with the suffix $-\varepsilon$, these forms denote forceful orders. -ka- is the only pre-radical TAM marker attested in combination with the subjunctive suffix $-\varepsilon$.

(110) uka uka <u>ukaloolwe</u> ni mambeya vaa waari
 'Come, come, you must be married to Mambeya vaa Waari.'

uk -a uk -a П -ka -lool -w 3come -imp2sg come -imp2sg sp2sg -inj -inj -marry -pass mambeya nı va -a U -arı pres Mambeya dp2 -det ip14 -porridge

(111) tukaneye 'We must go sleep.'

tu -ka -ŋεy -ε sp1pl -inj -sleep -inj

3. Conclusion.

The Langi verbal system is similar to most Bantu, and indeed Niger-Congo languages in that it expresses tense, aspect and mood through both 'simple' verb forms and syntactic verbal constructions. However, it seems to have adopted certain areal features, which probably originated through contact with the neighboring Cushitic languages, Alagwa and Burunge. Langi also seems to have adopted a new system of distinctions within the verbal paradigm, favoring aspectual oppositions over temporal ones, as evidenced by the fact that temporal distinctions are expressed through verbal constructions, whereas aspectual and modal distinctions are expressed directly by morphological verb form. Langi thus contributes to the study of how language contact, or perhaps more importantly, language enclavement, can lead to the adoption of grammatical elements, and even to a change in word order.

REFERENCES

- Bastin, Yvonne. 1983. La finale verbale -ide et l'imbrication en bantou. Tervuren: Musée Royal de l'Afrique Centrale.
- Botne, Robert. 1999. "Future and distal -ka-'s: Proto-Bantu or nascent form(s)?". *Bantu Historical Linguistics*, ed. by Jean-Marie Hombert & Larry M. Hyman. Stanford, CA: CSLI. Pp. 473-515.
- Comrie, Bernard. 1976. Aspect: an Introduction to the Study of Verbal Aspect and Related Problems. Cambridge: Cambridge University Press.
- Dahl, Östen. 2001. "Principles of areal typology". Language Typology and Language Universals, ed. by Martin Haspelmath, Ekkehard König, Wulf Oesterreicher & Wolfgang Raible. vol 2. Berlin & New York: Walter de Gruyter. Pp. 1456-1470.
- Dempwolff, Otto. 1916. "Beiträge zur Kenntnis der Sprachen in Deutsch-Ostafrika - 8. Irangi", Zeitschrift für Kolonialsprachen 6: 102-23.
- Dunham, Margaret. Forthcoming. *Eléments de description du langi, langue bantu F33 de Tanzanie.* Louvain-Paris: Peeters.
- Gilliard, L. 1928. *Grammaire synthétique de lontomba*. Bruxelles: Editions de l'Essorial.
- Gorman, W. A. R. 1950. Simple Silozi: A Guide for Beginners. London: Longmans Green and Co.

- Greenberg, Joseph H. 1966. "Some universals of grammar with particular reference to the order of meaningful elements". *Universals of Language*, ed. by Joseph Greenberg. Cambridge, MA: M.I.T. Press. Pp. 73-113.
- Guentchéva, Zlatka. 1990. *Temps et aspect : l'exemple du bulgare contemporain*. Sciences du langage. Paris: CNRS.
- Güldemann, Tom. 2003. "Grammaticalization". *The Bantu Languages*, ed. by Derek Nurse and Gérard Philippson. London: Routledge. Pp. 182-194.
- Hadermann, Pascale. 1996. "Grammaticalisation de la structure Infinitif + Verbe_{conjugué} dans quelques langues bantoues". *Studies in African Linguistics*, 25: 155-169.
- Hadermann, Pascale. 1999. "Les formes nomino-verbales de classes 5 et 15 dans les langues bantoues du Nord-Ouest". *Bantu Historical Linguistics*, ed. by Jean-Marie Hombert & Larry M. Hyman. Stanford, CA: CSLI. Pp. 431-472.
- Hagège, Claude. 1975. Le problème linguistique des prépositions et la solution chinoise. Louvain: Editions Peeters.
- Heine, Bernd. 1997. "Grammaticalization theory and its relevance for African linguistics". African Linguistics at the Crossroads: Papers from Kwaluseni, 1st World Congress of African Linguistics, Swaziland, ed. by Robert K. Herbert. Köln: Rüdiger Köppe Verlag. Pp. 1-16.
- Li, Charles N. & Thompson, Sandra A. 1974. "Coverbs in Mandarin Chinese: verbs or prepositions?". *Journal of Chinese Linguistics* 2: 257-278.
- Lord, Carol. 1993. *Historical Change in Serial Verb Constructions*. Amsterdam: John Benjamins.
- Meeussen, Achiel Emiel. 1967. "Bantu grammatical reconstructions". Africana Linguistica III. MRAC Annales 61. Tervuren. Pp. 79-121.
- Mous, Maarten. 2000. "Counter-universal rise of infinitive-auxiliary order in Mbugwe (Tanzania, Bantu F 34)". 'Mehr als nur Worte...', ed. by Rainer

Vossen, Angelika Mietzner & Antje Meissner. Köln: Rüdiger Köppe Verlag. Pp. 469-481.

- Mous, Maarten. 2003. *The Making of a Mixed Language: The case of Ma'a / Mbugu*. Amsterdam: John Benjamins.
- Mous, Maarten. 2004. *A Grammatical Sketch of Mbugwe, Bantu F34, Tanzania.* Köln: Rüdiger Köppe Verlag.
- Næss, Åshild. 2004. "Serial verbs and complex constructions in Pileni". *Complex Predicates in Oceanic Languages: Studies in the Dynamics of Binding and Boundness*, ed. by Isabelle Bril & Françoise Ozanne-Rivierre. Berlin: Mouton de Gruyter. Pp. 225-249.
- Nurse, Derek & Muzale, Henry R. T. 1999. 'Tense and aspect in Great Lakes Bantu languages'. *Bantu Historical Linguistics*, ed. by Jean-Marie Hombert & Larry M. Hyman. Stanford, CA: CSLI. Pp. 517-544.
- Nurse, Derek. 2000a. Inheritance, Contact, and Change in Two East African Languages. Köln: Rüdiger Köppe Verlag.
- Nurse, Derek. 2000b. "Diachronic Morphosyntactic Change in Western Tanzania". '*Mehr als nur Worte...*', ed. by Rainer Vossen, Angelika Mietzner & Antje Meissner. Köln: Rüdiger Köppe Verlag. Pp. 517-534.
- Nurse, Derek. 2003. 'Aspect and tense in Bantu languages'. *The Bantu Languages*, ed. by Derek Nurse and Gérard Philippson. London: Routledge. Pp. 90-102.
- Ozanne-Rivierre, Françoise. 2004. 'The evolution of the verb 'take' in New Caledonian languages'. *Complex Predicates in Oceanic languages: Studies in the Dynamics of Binding and Boundness*, ed. by Isabelle Bril & Françoise Ozanne-Rivierre. Berlin: Mouton de Gruyter. Pp. 331-346.
- Robert, Stéphane. 1996. 'Aspect zéro et dépendance situationnelle: L'exemple du Wolof'. Dépendance et intégration syntaxique: Subordination, coordination, connexion, ed. by Claude Müller. Tübingen: Niemeyer. Pp. 153-161.

- Sebba, Mark. 1987. The Syntax of Serial Verbs: An Investigation into Serialization in Sranan and Other Languages. Amsterdam: John Benjamins.
- Stegen, Oliver. 2000. "Rangi vowel system: five or seven?" Paper presented at the 30th Colloquium of African Languages and Linguistics, Leiden, August 28-30, 2000.
- Stegen, Oliver. 2002, "Derivational processes in Rangi". *Studies in African Linguistics* 31: 129-153.
- Welmers, William E. 1973. *African Language Structures*. Berkeley: University of California Press.
- Whiteley, Wilfred Howell. 1955. *The Structure of the Kuria Verbal and its Position in the Sentence*. London: Ph.D. dissertation.
- Whiteley, Wilfred Howell. 1960. *The Tense System of Gusii*. Kampala: East African Institute of Social Research.

34 Bd. de l'Hôpital 75005 Paris, France madunham@club-internet.fr [Received July 7, 2004, accepted January 6, 2005]

THE IMPORTANCE OF WORD ORDER IN EXPLAINING TONE PATTERNS IN AVOKAYA VERBS

Eileen Kilpatrick & Leoma Gilley SIL International

This paper presents a set of ordered rules accounting for tone changes that occur in Avokaya verbs. The most prevalent shape for Avokaya verbs is monosyllabic. Disyllabic verbs roots behave the same way as disyllabic verbs which are composed of a derivational prefix plus a monosyllabic verb root. The derivational prefixation process is the same for verbs in all grammatical constructions. Verbs in SVO constructions and SOV constructions are treated separately, since different rules apply for these two sets of constructions. Avokaya syllables may carry a high, mid, low or rising tone, but not a falling tone. In this analysis of Avokaya, we show that the expected HL tone pattern has merged to form a M tone in the verbal system.

1. Introduction.

Avokaya belongs to the Moru-Ma'di sub-group of the Central Sudanic language family, a branch of the Nilo-Saharan family. The dialect of Avokaya used in this paper is spoken in southern Sudan. The data for this paper were gathered over approximately 23 years of work in the language. This paper provides a principled account of tonal alternations in the verbs in Avokaya.

Avokaya has a seven vowel system, three Advanced Tongue Root ([+ATR]) vowels [i], [u] and [ə], and four vowels that are [-ATR], [I], [ɛ], [ɔ], and [a]. Vowel harmony causes a [-ATR] vowel in a verb prefix to assimilate to the corresponding [+ATR] vowel (Callinan 1981). The nominal system of Avokaya demonstrates three level tone patterns, High, Mid and Low. In addition, there is a Rising contour on short vowels, but no Falling contour on short vowels. Phonetically a Falling or Rising contour may occur on long vowels, but these long vow-

els are considered sequences of two Tone Bearing Units (TBU).¹ There are only a few examples of vowel sequences in Avokaya with either contour tones or level tones.

In the nominal system, there are no restrictions on the occurrence of the three level tones, as shown by the representative examples below.

(1)	Monosyllabic nouns						
	Η		М		L		
	pá	'leg'	fē	'tree'	fà	'bone'	
(2)	Disyl	labic nouns					
	HH		HM		HL		
	ídrí	'vein'	írī	'eagle'	ínà	'crocodile'	
	MH		MM		ML		
	īmbá	'net'	ōzē	'rain'	īndrì	'child minder'	
	LH		LM		LL		
	ìfź	'millet'	ìlī	'wind'	ìzà	'sadness'	

A rising tone can occur on monosyllabic nouns.

(3) fɔ̈́ 'flower' s̈́i 'hail'

In multisyllabic nouns a rising tone occurs only on the first syllable of disyllabic nouns with a monosyllabic root.

¹ The following abbreviations are used in this paper: 2S second person singular, 3S third person singular, 3rd IMP third person imperative, conj conjunction, DIR direction toward speaker (ventive), dpfx derivational prefix, EOC end of embedded construction, H high tone, IMP imperative, INFIN infinitive, L low tone, M mid tone, neg negation, O-FOC focus on object, PERF perfective aspect, PURP purpose marker, R rising tone, SM subject marker for sequential action and relative clauses, SEQ verb marker for sequential action, S-FOC focus on subject, sfx suffix, vbrt verb root.

(4) yı̃-wá

'fly type'²

Rising tone also occurs on the first syllable of multisyllabic multi-morphemic nouns with syllabic reduplication.

(5) dǐ dī -yá 'bird type' dǔ dū-wá 'headache type'

Since contour tones on short vowels are restricted to initial syllables, this could be analyzed as right-to-left association of tone, but in-depth discussion of this is beyond the scope of this paper.

There are contour tones on phonetically long vowels, analyzed as vowel sequences, in Avokaya nouns as follows.

(6)	dīī	'person'	(High-Mid)
	ràtáà	'chant'	(High-Low)
	gĭipírí	'mosquito'	(Low-High)

By way of contrast, the tone patterns of Avokaya verbs have shown some interesting restrictions. There are only four tone patterns in evidence.³

(7)	Class 1	Class 1A	Class 2	Class 2A
	Mid	Mid-Mid	Low	Mid-Low
	$c\bar{a}^4$ 'arrive'	ājī 'accompany'	nì 'go'	ətì 'lean (against)'

² A group of multisyllabic verbs derive from nouns. These verbs, which are not within the scope of this paper, have regular surface tone patterns of Mid High (High) (High) Low, e.g. $ib\dot{a}dik\dot{a}$ 'tie around (something)', $ib\dot{a}b\dot{a}tiz\dot{a}$ 'baptize' The first syllable is a derivational prefix with preassigned HL tone. The verb root has a HL melody. The tone changes in the various verb forms of this class of verbs are similar to the changes in the Mid-Low verb class.

³ There are a group of multisyllabic verbs derived from nouns. These verbs, which are not within the scope of this paper, have regular surface tone patterns of Mid High (High High) Low. The first syllable is a derivational prefix with preassigned HL tone. The verb root has a HL melody. The tone changes in the various verb forms of this class of verbs are similar to the changes in the Mid-Low verb class.

⁴ The symbol c represents a phoneme that alternates between [ts] and [tʃ] depending on context, and the symbol j represents a phoneme that alternates between [dz] and [dʒ] depending on context.
Class 1 and Class 2 are always monosyllabic verbs. Class 1A and Class 2A are always disyllabic verbs. Traditionally the verbal system has been viewed as having four verb classes (Callinan 1986). Analyses of the verbal system of other Moru-Ma'di languages are based on tone without determining whether some verb classes could be derived from others (Wright 1995, Andersen 1984).

To date, there has been no attempt to provide an analysis of the verb system of Avokaya that accounts for all the tonal changes evident in the data. A full listing of those changes is found in example (1). Each structure listed will be handled separately.

(8)	(Class 1	Class 1A	Class 2	Class 2A	Morphological Structure
	SVC) construc	tions			
	a. c	cā	ājī	nì	ətì	citation form ⁵ (imp.: 3.1)
	b. c	cá-lé b $\bar{\epsilon}^6$	ājí-lí bē	nì-lí bē	ātì-lí bē	when (tail- head linkage: 3.2)
	c. c	cá kō	ājí kō	nĭí kō	ətii kə	neg. statement: 3.3
	(S)	OV constri	uctions			
	d. c	cà -zố	ājí-zó	nì-zố	ətì-zó	aspect/mode inflection: 4.1
	e. c	cǎ	ājí	nĭ	əti	S conj. O V: 4.2
	f. c	cà kā	ājī kā	nì kā	ətì kā	genitive: 4.3
	g. (cà cà	ājī ājī	nì nì	ətì ətì	participle: 4.4
	('arrive'	'accompany'	ʻgo'	'lean'	

Looking down the columns, it is clear that there are significant tonal changes from construction to construction, especially in the Class 1 verbs.

⁵ There are no uninflected forms of the verb in Avokaya. Therefore the underlying form for verbs in classes 1 and 1A cannot stand alone. The imperative has been used as the citation form since it has no affixes containing segmental phonemes. The resulting mid tone in this form is explained in section 3.1.

⁶ Both this and the infinitive suffix, also $-l\varepsilon$, are manifested as -li after the [-ATR] vowel *i*, and as -li after the [+ATR] vowels *u* and *i*. See section 3.1.1 for the discussion of this tail-head linkage construction.

In this paper, we show that Avokaya has two basic tone classes of verbs, and that the forms found in 1A and 2A are subsets of those basic tone classes. When a derivational prefix is added to a verb root, the resulting verb has the same tone patterns as the verb forms found in Classes 1A and 2A. In this case, the derivational prefix does not change the verb to a different word class, but changes the syllable structure of the verb, resulting in tone patterns distinct from the monosyllabic verb roots. A series of rules are presented that account for all the tonal changes which occur in the verbal system.

This analysis assumes that the underlying tones of the verb roots are either High or Low. However, it should be noted that the underlying High tone form never occurs in isolation in the monosyllabic verbs. The Low tone verbs do not present as much of a problem, but the tone changes in the High tone verbs are of particular interest since the High tone monosyllabic verbs often surface as having a Mid tone. It is assumed that there is an additional Low tone morpheme that results in a lowering of the surface tone from High to Mid. This morpheme is discussed in section 3.1.

As is often the case, derivational affixes are added to a root before inflectional ones. Also, derivational affixes may have a more extensive effect on the word than inflectional affixes do. It is assumed in this analysis that any derivational affixation will be completed before any inflectional affixation.⁷ The order of the nuclear constituents of a clause determines the tonal pattern of verbs in some contructions. The underlying word order is SVO, but there are certain constructions that have an SOV order. First, we will discuss underlying tone, then the tones of verbs in the SVO order, and finally the tones of verbs in the SOV order.

2. Underlying Tones.

Two lexical tone classes are posited for (monosyllabic) verb roots in Avokaya. In the first class, a High tone is posited as the underlying tone while the second class has a Low pattern. There is a disyllabic subset of verbs within each of these two classes. (Historically the disyllabic verbs were composed of a derivational prefix plus monosyllabic verb root.) When a derivational prefix (which is always Mid toned) is added to the High tone monosyllabic verb roots of Class 1, the behavior

⁷ In fact, if a derivational prefix were to be added after the application of the rules affecting tone changes in the verb, an ungrammatical form would result. For instance, if the directional prefix \bar{a} - were to be added to the root $c\dot{a}$ 'arrive' after the rules of level 2 were applied, the following ungrammatical forms would result: $\bar{a}c\dot{a}l\dot{\epsilon}$ 'to arrive here', $\bar{a}c\ddot{a}$ (lest...) arrive'.

of the new verb stems is identical to the verbs of Class 1A. Also the derivational prefix added to the monosyllabic verb roots of Class 2 behave like the verbs of Class 2A. Therefore the disyllabic verbs of Class 1A are assumed to have the underlying tone pattern Mid High, while those of Class 2A have the underlying tone pattern of Mid Low.



1.1 Derivational prefixes. There are various derivational verb prefixes which, when added to verbs, form new verb stems. Examples of these prefixes are given below. These prefixes differ from inflectional prefixes in that their addition results in a disyllabic verb with all the tone changes that disyllabic verb stems have. Inflectional prefixes, on the other hand, do not affect the tone of the verb root.

The derivational prefixes that can be added to Avokaya verbs are as follows. The first is "direction toward speaker" (ventive) \bar{a} -.⁸

(10)	cā	'arrive (there)'	ā-cā	'arrive (here)'
	nì	'go'	ā-nì	'come'

This prefix is highly productive. The second is the multiplicative, denoting multiple action or participants, $\bar{2}$ - or $l\bar{2}$ -⁹ depending on the specific verb.

(11)	fū	'kill'	ū-fū	'kill (many)'
	6à	'put, make'	15-6à	'make (many things)'

This prefix is also highly productive. The third is the transitivizer, \bar{a} -, $l\bar{a}$ -, $l\bar{a$

⁸ This prefix is manifested as \bar{a} - before verb roots with [+ATR] vowels and as \bar{a} - elsewhere.

⁹ In both cases, the prefix $\bar{2}$ - is manifested as \bar{u} - before verbs with [+ATR] vowels and $\bar{2}$ - elsewhere.

¹⁰ The prefix $l\bar{c}$ - is manifested as $l\bar{u}$ - before verb roots with [+ATR] vowels.

(12)	a. rā	'run'	5 - rā	'strain (liquid) out'
	b fi	'enter'	l5 - fi	'bring (something) in'
	c. fò	'exit'	lō - fò	'take out (of a house)'
	d. drā	'bubble/parboil'	ā- drā	'parboil (something)'
	e. trē	'shout/make noise'	$\bar{\epsilon}$ - tr $\bar{\epsilon}$	'clear (throat)'
	f. cī	'be lit'	ī- cī	'light (fire)'

This type of prefix is fairly productive.

For each derivational prefix the prefix tone surfaces as a Mid tone. There are also restrictions as to which derivational prefixes can be added to which verb roots. For instance, intransitive motion verbs can take a directional prefix, but not the prefix denoting multiple action.

Verbs in Class 1 are monosyllabic verbs with a High tone root, and Class 2 are monosyllabic verbs with a Low tone root. These are the two basic verb classes. Verb Classes 1A, disyllabic verbs with MH tone pattern, and 2A, disyllabic verbs with ML tone pattern, are linked to Class 1 and Class 2 respectively, since disyllabic verbs stems derived from the monosyllabic verb roots of Class 1A, and disyllabic verb stems derived from the monosyllabic verb stems derived from the monosyllabic verb roots of Class 2 act exactly like the verb roots of Class 2A. The same tone changes occur in the corresponding grammatical constructions. Compare the tone patterns of the monosyllabic verb roots of Class 1 and 2 in (13) below having the added directional prefix \bar{a} -, with the verbs of Class 1A and 2A.

(13)		Prefixed Class 1	Class 1A	Prefixed Class 2	Class 2A	Morphological Structure
		MH	MH	ML	ML	
		'arrive (here)'	'accompany'	'come'	'lean'	
	a.	ā-cā	ājī	ā- nì	ətì	citation (imp.)
	b.	ā-cá-lé bē	ājí-lí bē	ā-nì-lí bē	ətì -lí bē	when (tail-head linkage) ¹¹
		'when arrived'	'when accompanied'	'when came'	'when leaned'	
	c.	ā-cá kō 'did not arrive'	ājí kō 'did not accompany'	ā- nǐí kɔ̄ 'did not come'	ətii kə ʻdid not lean'	negative statement
	d.	ā-cá-zó 'arrived'	ājí-zó 'accompanied'	ā-nì-zɔ́ 'came'	ətì-zó 'leaned'	aspect/mode inflection
	e.	ā-cá '(lest) arrive'	ājí '(lest) accompany'	ā-nǐ '(lest) come'	əti '(lest) lean'	S conj. O V
	f.	ā-cā kā 'of arriving'	ājí kā 'of accompanying'	ā-nì kā 'of coming'	ətì kā 'of leaning'	genitive
	g.	ā-cā ā-cā 'arriving'	ājī ājī 'accompanying'	ā-nì ā-nì 'coming'	ətì ətì 'leaning'	participle

¹¹Tail-head linkage is a means of slowing down the action of a discourse. Information of a previous clause is repeated in a when-clause introducing the next sentence. An example would be 'John went to town. When he went, he met an old man on the way. When he met the old man, the man asked him for a ride.' In Avokaya this is a grammatical construction that is distinct from other temporal clauses.

At level I, prefixed verbs of Class 1 become Class 1A verbs, and prefixed verbs of Class 2 become Class 2A verbs.

1.2 Patterns for the initial syllable of a disyllabic verb. The only four patterns for the initial syllable of a disyllabic verb are as follows.¹² The most common initial syllable of a disyllabic verb is \bar{a} - and its [+ATR] equivalent, \bar{a} - (see (8)).

(14) ājī 'accompany' ātì 'lean'

The prefixes \overline{i} -, and $\overline{2}$ - are equally possible and can occur on verbs regardless of the vowel of the verb root.

(15) īcā 'be able' ōpī 'shell (peanuts)'

There are also disyllabic vowel initial verbs where the same vowel occurs in both the first and second syllables.

(16)	ēdē	'prepare'	ījī	'ask'
	ūgù	'steal'	īpì	'press down'

The first syllable of disyllabic verbs may also be $l\bar{2}$ -, $l\bar{a}$ -, or $l\bar{i}$ -.

(17) lōpē 'extend' lāvī 'play' lībā 'creep'

The first syllables of all disyllabic verbs in Avokaya are restricted to these forms, which are similar to the forms of known derivational prefixes.

There are only a small number of disyllabic verbs which can take a recognized derivational prefix.¹³ To date, only two examples have been encountered in which a disyllabic verb takes the derivational prefix. In both cases, the initial vowel of the verb root is deleted when the derivational prefix is added.

(18)	ājī	'accompany'	lājī	'accompany many'
	ətì	'lean'	lūtì	'support a leaning object'

The deletion of this initial root vowel is shown in the following rule.

¹² There is no restriction to the inventory of phonemes that can occur in disyllabic nouns.

¹³ All verbs can take inflectional prefixes.

(19) Root Vowel Deletion Rule (RVD) $V \rightarrow \emptyset / V_{-dpfx} __CV_{vbrt}$

This vowel deletion does not occur when an inflectional prefix is added.

(20) á-ājī 'be accompanied' (agentless passive)

1.3 Derivational prefixation process. In light of the above examples, the process involved in adding derivational prefixes to the underlying form (UF) of a verb root is as follows.

(21)	Class 1	Class 1A	Class 2	Class 2A	
	H ca 'arrive'	M H a ji 'accompany'	L ni 'go'	M L ə ti 'lean'	UF of root
	M + H $ $ $a + ca$ (dir.)	$M + M H$ $\begin{vmatrix} & \\ & \\ _{0} + a \\ ji \\ (mult.)$	$M + L$ $ _{a + nI^{14}}$ (dir.)	$M + M L$ $\begin{vmatrix} & \\ \\ & \\ \\ l_{2} + & 3 t_{1} \end{vmatrix}$ (trans)	Prefixed form
	$ \begin{array}{c} M + H \\ & \\ a + ca \end{array} $	$ \begin{array}{c} M + H \\ & \\ l_{0} + j_{I} \end{array} $	M + L $\begin{vmatrix} & \\ & \\ a + nI \end{vmatrix}$	$ \begin{array}{c} M + L \\ & \\ l_{0} + t_{1} \end{array} $	RVD
	[ācá] 'arrive	[lɔ̄jí] 'accompany here'	[ānì] 'come many'	[lūtì] 'support a leaning obje	Uninflected form

Having given evidence for the underlying tones of the two basic classes of verbs and having demonstrated their similarities through derivational processes, the next section will demonstrate how the tone patterns shown in (8a-c) are derived.

¹⁴ Different derivational prefixes are used in the examples due to semantic restrictions on which prefixes each verb can take.

3. Tonal Patterns For Verbs In SVO Constructions.

In this section, SVO constructions will be considered. Initially there will be a discussion of the constructions using the citation form of the verb found in (8a). In each of these constructions there is a floating L tone attached to the verb which accounts for the Mid tone pattern of Class 1 verbs and the Mid-Mid pattern of Class 1A verbs. This is followed by the dependency construction found in the 'when' clauses for tail-head linkage (example 8b). Finally, the negative statement verb form of example (8c) will be discussed.

3.1 SVO constructions with floating low tone. There are a number of SVO constructions which have an inflectional suffix on the verb consisting of a floating Low tone. With the exception of the imperative clause, the constructions also have a particle that occurs later in the clause. These constructions include Perfective, with the particle $tr\dot{a}$; Focus constructions, with the particles $n\dot{i}$ or Tr, and, the Imperative construction. In these constructions, when the floating low tone is added, the H+L combination results in a phonetic Mid tone. When the underlying tone is Low (as in Class 2 verbs), the L+L combination does not change the surface form. Furthermore, this floating low tone does not result in downstep as in many African languages. The level of the High tone in $tr\dot{a}$ is the same level as the High tone of $m\dot{a}$.

(22)	Perfective		
	a. /má cá ` trá/	[cā]	'I arrived'
	I arrive PERF		
	b. /má ājí `gólā trá/	[ājī]	'I accompanied him'
	I accompany 3S PERF		
	c. /má nì ` trá/	[nì]	'I went'
	d. /má ātì `gólā trá (fɛ̃ rú)/	[ətì]	'I leaned it (against a tree)'
(23)	Focus		
	a. /má cá ` nì/	[cā]	'I am the one who arrived'
	I arrive S-FOC		
	b. /má ājí `gólā ʔī/	[ājī]	'he is the one I accompanied
	I accompany 3S O-FOC		
	c. /má nì ` nì/	[nì]	'I am the one who went'
	d. /má ātì `gɔ́lā nì/	[ətì]	'I am the one who leaned it'

(24)	Imperative		
	a. /ní cá `/	[cā]	'arrive (at a certain place)'
	b. /ní ājí `gólā/	[ājī]	'accompany him'
	c. /ní nì $$ vól $\hat{\epsilon}$ /	[nì]	'go away'
	d. /ní ātì `gʻɔlā fē rú/	[ətì]	'lean it against a tree'

In the constructions in (22-24), there is a floating grammatical Low tone that comes after the verb root (as well as a particle that comes later in the clause to distinguish the clause type). This floating Low tone attaches to the verb root according to the Floating Tone Association rule shown in (25). It also coalesces with the High tone on the root of the Class 1 verbs to form a Mid tone according to the Tone Coalescence rule (26).



The Low tone morpheme is added to the underlying tone on the root vowel. In these constructions with a floating Low tone, there is coalescence of tone where a HL tone coalesces to a Mid tone.

(26) Tone Coalescence (TC)

$$H \ L \qquad M$$

 $\bigvee \rightarrow \qquad \downarrow$
 $V \qquad V$

This rule accounts for certain Mid tones that exist in the surface level form in (22-24). The result of this rule is a neutralization of tones; a HL sequence could give rise to a M, as well as an underlying /M/ surfacing as a M. This coalescence of HL into M explains why there are no falling tones on short vowels while there are rising tones. The output of these constructions was used as the "citation" form (tone patterns Mid, Mid-Mid, Low, and Mid-Low) of example (8a).

The following derivation of the Perfective, Focus and Imperative verb forms show how these rules apply.



We assume that the grammatical Low tone merges with the Low tone of the melodies of the Class 2 verbs, in accordance with the Twin Sister Convention expounded by Clements and Keyser (1983).

3.2 'When' dependency construction. There is only one SVO construction where the verb stem takes a segmental inflectional suffix, the 'when' construction which repeats the action of the previous clause and takes the suffix $-l\epsilon$. The repeated verb provides tail-head linkage in order to slow down the progress of the narrative. This suffixed verb form¹⁵ undergoes no tone changes but keeps the underlying tone of the verb stem (see 8b). Consider the examples below.

(28)S V 0 a. (má ndrē ámā átá trá.) má ndré -lé ámā átá bε rī... 1S see my father PERF1S see -when when my father EOC '(I saw my father.) When I saw my father...' ámā átá rī... -lí bε b. má **āi**í 1S accompany -when when my father EOC

'When I accompanied my father,..'

¹⁵There is also a form of this construction which has no segmental suffix. In that construction, the verb form has a floating low tone attached, like the constructions in section 3.1.

- c. mánì -lí bē rī...
 1S go -when when EOC
 'When I went...'
- d. gʻɔlā ātì -lí bē rī...
 3S lean -when when EOC
 'When he/it leaned...'

In each case, the verb stem retains the underlying tones.

(29)	Class 1	Class 1A	Class 2	Class 2A
	cá-lέ	ājí-lí	nì-lí	əti-li
	'when arrived'	'when accompanied'	'when went'	'when leaned'

The only change that occurs is in the vowel quality of $-l\hat{\varepsilon}$, in that it harmonizes with the height of the previous vowel. Since the tone of the suffix $-l\hat{\varepsilon}$ is invariantly High, it is assumed that it bears a pre-assigned High tone.

The tone assignments for the verb of the dependency construction are:

(30)	Н Н	МНН	LH	ML H
	$ca + l\epsilon$	$a' j' + l\epsilon'$	$n\mathbf{l} + l\mathbf{\dot{\epsilon}}$	$\frac{1}{2}t_{1}^{\dagger}+l_{\epsilon}^{\dagger}$
	[cá-lɛ́]	[ājí-lí]	[nì-lí]	[ətì-lí]

2.3 Negative statement. The final SVO verbal construction with distinct tone patterns is the Negative Statement (8c) with a negation morpheme on the verb as well as a negation word.

(31)	Negative statement					
	a.	gólā cá kō		'he did not arrive'		
		3S arrive not				
	b.	gʻolā ājí	mā kō	'he did not accompany me'		
		3S accompany	y me not			
	c.	gólā nìí kō		'he did not go'		
	d.	gólā ētii kē		'he did not lean'		

The negation morpheme on the verb is realized as a single empty V slot with H tone suffixed to the verb root. Since this V is empty of all features except tone, it takes all other features from the final V of the verb stem. This is explained by the following Vowel Identity Rule, which is applied in this construction.

(32) Vowel Identity Rule (VI)
H
[...C V] vbrt + [V] neg
[
$$\alpha$$
F]

In this rule, the vowel features from the root, except for the tone, are spread to the empty set of features in the vowel slot of the suffix. The High tone is preassigned to the empty vowel slot. For verbs of Class 1 and 1A, a vowel reduction rule in also in effect.

The rules involved in producing the Negative Statement form of the verb are shown below.



At the post-lexical level, there is a vowel reduction rule. If the vowel features including tone are identical and create a long vowel, then the vowel is shortened. If the tone is different, then the two vowels must be retained.

(34) Vowel Reduction Rule
$$(VR)^{16}$$

 $\acute{v}\acute{v} \rightarrow \acute{v}$

When the Vowel Reduction Rule is applied, the outcome (shown in (36)) has an identical surface form to the examples in (8c).

(35)[cá][ājí][nǐi][ētìi]'didn't arrive''didn't accompany''didn't go''didn't lean'

All of the tonal patterns for the verbs in SVO constructions have now been accounted for.

3.4 Inflectional prefixes on the verbs of SVO constructions. At this stage inflectional prefixes may be added to the verbs in SVO clauses. There are two inflectional prefixes in Avokaya: 5- indicates an agentless passive, and 5- indicates irrealis. The agentless passive prefix can be added to the verb in any SVO construction. The irrealis prefix is used in conditional clauses, for indicating reported speech, and for indicating a third person imperative. These constructions have SVO word order with a floating Low tone on the verb like the constructions in Section 3.1. Both of these inflectional prefixes assimilate fully to the initial vowel of a vowel initial verb stem, and they assimilate to the vowel u before consonant initial verb stems containing [+ATR] vowels. They remain σ before consonant initial verbs with [-ATR] vowels. Both of these prefixes may be added to a single verb stem. Neither one affects the tones of the verb stem.

In the example below, we represent each tone class by a transitive verb. The inflectional prefixes are added to the citation form of the verb (8). This citation form is used to indicate, among other things, perfective aspect or imperative mode.

¹⁶This rule applies only to vowels in verb roots/stems, as shown by such examples in other word classes as *itin* 'for a long time', *ymbāā* 'open', *fàà* 'aimlessly'

$(\mathbf{J}\mathbf{U})$	10100	with inflectional pr	CJINCS		
	Class 1	Class 1A	Class 2	Class 2A	
	ndré	ājí	6à	ətì	UF of verb
	'see'	'accompany'	'put'	'lean'	
	ndrē	ājī	бà	ətì	basic citation
	óndrē	áājī	56à	áātì	Passive
	'he was seen'	'he was accompanied'	ʻit was put'	'it was leaned'	
	gʻolā `ondrē	gólā àājī	gólā òbà	gólā àātì	3 rd IMP
	'let him see'	'let him accompany'	'let him put'	'let him lean'	
	óòndrē	áàājī	óòbà	á ðātì	both prefixes
	'let it be seen'	'let him be accompanied'	'let it be put'	'let it be leaned'	

(36) Verb forms with inflectional prefixes

Since the above prefixes do not affect the tone of the verb stem, no tone rules are necessary. In the last line of data of (36), note that a sequence of two vowels has a High Low sequence, not a Mid tone as would occur on a short vowel.

4. SOV Tonal Patterns.

There are certain grammatical constructions in which there is an SOV word order. The first of these includes constructions which have segmental inflectional suffixes for aspect or mode on the verb (see (8d)). An example of this type of construction is the sequential action construction used in procedural texts and to trace the event line in narrative texts.

(37)	S	0	\mathbf{V}			
	gólā	drí	ámā	sà	-zś	'(next) he slapped me'
	he	SM	me	slap	SEQ	

The other construction occurs when certain subordinate conjunction constructions have an S conj O V order (see also (8e)).

(38)	S	conj	0	V		
	gʻolā	òzś	ámā	fŭ ¹⁷	rī	'lest he kill me'
	he	lest	me	kill	EOC	

The two examples (37) and (38) have Class 1 verbs. The High tone Class 1 verb root has a Low tone in SOV constructions which have segmental inflections on the verb (37). The High tone Class 1 verb root has a rising tone in the S conj O V constructions (38). To account for all tone changes in Class 1 verbs, it is proposed that monosyllabic verb roots of Class 1 and Class 2 undergo a second derivational process, that of tone replacement in these SOV constructions. This rule only applies to monosyllabic verb stems without derivational prefixes. The tone of the monosyllabic verb stem is replaced by a Low tone.¹⁸ For Class 1 verbs without derivational prefixes, High tone is replaced by Low tone. For Class 2 verbs the tone replacement results in a Low tone, since Low is replaced by Low. The tones of the remaining verb classes remain the same. This process applies after any derivational prefix is added. The addition of a derivational prefix produces a disyllabic verb stem which does not undergo the tone replacement derivation. Therefore the two derivational processes are ordered.

(39) Tone Replacement Rule (TR) $T \rightarrow L$ for monosyllabic verb stems in (S)OV constructions

The resulting tone assignments for verbs in SOV constructions are shown in (40).

(40)	Class 1	Class 1A	Class 2	Class 2A	
	H ca	М Н а ј ^ї	L nı	M L ə ti	UF
	L ca [cà] 'arrive'	M H a jī [ājí] 'accompany'	L ni [nì] 'go'	M L ə ti [ətì] 'lean'	TR

¹⁷ The rising tone in this construction is explained in section 4.2.

¹⁸ We would like to express our appreciation to Keith Snider for bringing to our attention the phenomenon of tone replacement.

The SOV constructions are of two types, those with a segmental inflectional suffix on the verb, and those with a conjunction intervening between the Subject and Object (S conj O V). Each of these two types have their own tone pattern, so will be treated separately in this paper.

4.1 SOV constructions with segmental inflectional suffixes. The possible verb suffixes for SOV constructions include -?a 'imperfective', -z5 'sequential action, purpose', $-l\epsilon$ 'dependent/infinitive "to", and $-r\epsilon$ 'embedded dependent'. Each of these morphemes surfaces as a -CV.¹⁹ As with the "when" $-l\epsilon$ construction, the addition of these suffixes has no effect on the tone of the verb stem. However, as mentioned above, the SOV word order results in the Low tone replacement. When the verbal suffixes mentioned above are added to the verb stem, the tone of each of these verb suffixes is pre-assigned to the suffix, as shown below. This pattern can apply to all the inflected forms mentioned at the beginning of this section with the result shown in (41).

(41)	With mode inflection -z5								
	Class 1	Class 1A	Class 2	Class 2A					
	H ca	M H 	L nī	M L ə tı	UF				
	L ca	M H a j1	L nı	M L əti	TR				
	$\begin{array}{c c} L & H \\ & \\ ca + z 2 \end{array}$	$ \begin{array}{c c} M H & H \\ & & \\ a & jI + ZO \end{array} $	L H ni +zə	$ \begin{array}{c c} M L & H \\ & & \\ \neg & ti + z \neg \end{array} $	+ mode -zś				
	[cà-zɔ́] 'arrived'	[ājí-zɔ́] 'accompanied'	[nì-zɔ́] 'went'	[ətì-zɔ́] 'leaned'					

4.2 S conjunction O V. The second major set of SOV constructions have a conjunction between the S and the O. The possible conjunctions in this construction

¹⁹ Only one inflectional suffix at a time may be added to a verb stem.

type are $\partial z \beta$ 'lest', $\partial t r i$ 'when' for actions in the past that have not previously been mentioned, and $\partial r \beta$ 'if' for contrary-to-fact actions.

An example of such a construction is given below in (42).

(42) gólā òzó cǎ rī 'lest he arrive' 38 lest arrive:conj EOC

In this construction the verb $c\check{a}$ has a rising tone. Due to the fact that this is an SOV construction, the Tone Replacement rule accounts for the Low tone on the verb. In addition, we posit a conjunction morpheme after the verb stem which consists of a floating High tone. This floating High tone associates leftwards. There is a strong tendency in Avokaya for right to left association. (See the discussion of floating low tone in section 3.1.)

In example (43) the processes involved in the tone associations for the 'lest' construction are shown.

(43)	$\begin{array}{c} Class \ l \\ H + H \\ l \\ ca \end{array}$	$\begin{array}{c} Class \ IA \\ M H + \textcircled{H} \\ \\ a \\ \end{matrix}$	Class 2 $L + \bigoplus$ $\prod_{n_{I}}$	Class 2A M L + (H) $ _{2} ti$	UF
	L+H ca	$ \begin{array}{c} M H + H \\ \\ a \\ ji \end{array} $	L + (H) ni	M L+H ə ti	TR
	L H L',' ca [că] '(lest) arrive'	M H H $\begin{vmatrix} & & \\ a & \\ ji \\ [\bar{a}ji] \\ `(lest) accompare}$	L H L'' ni [nĭ] ny' '(lest) go	M L H / ´ ə ti [ətì] ` (lest) lean'	FTA

The two High tones on $\bar{a}ji$ merge since they are the same tone, so only a level High tone results. The Low-High combination on the other verb roots surfaces as a short rising tone on the root vowel. The same pattern applies to all constructions using the conjunctions listed above.

4.3 Genitive construction and participial form. In the genitive construction (8f), the word order of the phrase is noun head $+ O + V + k\bar{a}$. We posit a floating

Low tone occurring before $k\bar{a}$ 'of' which will associate to the word to its left, in this case the verb. Due to the OV word order, the Tone Replacement rule operates on the monosyllabic verb roots of Class 1 and Class 2. If we look at the verbs of Classes 1, 2 and 2A, there would be no need to posit the floating low tone. However the derivation of Class 1A would be incorrect without the floating low tone $(*\bar{a}j\hat{i})$. By positing the floating low tone, the High and Low tones combine to form a phonetic Mid tone on the short vowel, which is the correct form $(\bar{a}j\hat{i})$.



For a transitive verb the participial form of the verb occurs in an OV construction. As with the genitive construction, the participle is formed by the verb stem, followed by a floating Low tone, then the whole verb stem is reduplicated, as shown below. The presence of the floating Low tone is necessary to account for the tone on the verbs of Class 1A.



Thus all reduplicated verb forms surface with either L or M tones. No High tones occur in this construction.

5. Conclusion.

All tone changes in Avokaya verbs can be accounted for through a series of rules operating on the verb forms in different grammatical constructions. We have shown that derivational prefixation occurs before other processes which affect the tone of verbs. We have also shown that the processes involved in SVO constructions differ from the processes in SOV constructions, in that monosyllabic verb roots undergo the derivational process of tone replacement affecting SOV constructions before inflectional changes operate.

The ordering of the derivational processes account for why the tone patterns change when a derivational prefix is added to a Class 1 verb in an SOV construction.

(46)	[[cà] - zɔ́]	'arrive (there)'
	[[ā-[cá]]-zɔ́]	'arrive (here)'

The addition of the derivational prefix changes the Class 1 verb to Class 1A before the derivational process of tone replacement, which only affects Class 1 and Class 2 verbs.

Our analysis of tone replacement also accounts for why the Class 1 verb forms in the following constructions are different, even though the two suffixes are homophonous.

(47) má ndré -lé bē nī rī 'when I saw you'
 I see -when when 2S EOC

(48) má lè ánī ndrè -lé 'I want to see you'
 I want 2S see INFIN

Example (47) is an SVO construction, and example (48) is an SOV construction undergoing tone replacement.

We posited floating tones which account for tone changes resulting at the inflectional level. The differences in the perfective and genitive forms of the Class 1 verbs are also due to the fact that they have different word orders.

(49)	má I	gī buy	à'wá meat	trá PERF	'I bought meat.'
(50)	làfā money	à'wá meat	gÌ buy	kā of	'money of/for buying meat'

Both constructions have a floating Low tone. In example (49), the perfective construction, which is SVO, has underlying High tone in Class 1 verbs. The floating Low tone merges with this High tone to be manifested as a Mid tone. However, in example (50), the genitive form, which is SOV, has had the High tone replaced by a Low tone before the floating Low tone is applied.

Therefore previously difficult tone issues in Avokaya verbs are accounted for in our analysis. Significantly, these few rules account for all the tone changes in all the verbal constructions in Avokaya in an orderly way. (See the chart of tone changes on the following page).

Table 1: Summary chart of the changes							
Verb Class	1	1A	2	2A			
Root/Stem forms	Н	ΜН	L	ML			
Prefix Derivation	Н	ΜН	L	ML			
Floating Low, SVO, Perf, Focus, Imper, Surface	HL M	M HL M M	LL L	M LL M L			
High tone on empty V slot SVO, Negative, Surface	HH H	M HH M H	L H LH	M L H M LH			
Derivation (SOV) Tone Replacement Floating High, SOV	L L H	M H M HH	L L H	M L M L H			
S Conj O V, Surjace Tone Floating Low, SOV Genitive Surface Tone	к LL L	мн MHL MM	к LL LI	M K M LL M I			
Service, Surjuce 10/10	-						

REFERENCES

- Andersen, Torben. 1984. "Aspect and Word Order in Moru". Journal of African Languages and Linguistics 6: 19-34.
- Callinan, Lynne. 1981. "A Preliminary Study of Avokaya Phonemes". Occasional Papers in the Study of Sudanese Languages 1: 64-73. Summer Institute of Linguistics.
- Callinan, Lynne. 1986. "Sentence Constructions in Avokaya". Occasional Papers in the Study of Sudanese Languages 5:48-72. Summer Institute of Linguistics.

- Clements, George N. and Samuel Jay Keyser. 1983. CV Phonology: A Generative Theory of the Syllable. Linguistic Inquiry Monographs. Cambridge: MIT Press.
- Goldsmith, John. 1990. Autosegmental and Metrical Phonology. Oxford: Black-wells.
- Kilpatrick, Eileen. 1981. "Avokaya Predication". Occasional Papers in the Study of Sudanese Languages 1: 91-109. Summer Institute of Linguistics.
- Mohanan, K.P. 1986. *The Theory of Lexical Phonology*. Studies in Natural Languages and Linguistic Theory. Dordrecht: D. Reidel Publishing Co.
- Pulleyblank, D. 1986. *Tone in Lexical Phonology*. Studies in Natural Languages and Linguistic Theory. Dordrecht: D. Reidel Publishing Co.
- Wright, Douglas. 1995 *The Verbal System of Logo (Logoti)*. Dallas, Texas: Graduate Institute of Applied Linguistics, M.A. thesis.

SIL, Box 750 Entebbe, Uganda eileen_kilpatrick@sil.org

[Received January 18, 2004 accepted January 20, 2005]

7614 Breckenridge Lane Knoxville, TN 37938 leoma.gilley@suremail.org

Studies in African Linguistics Volume 33, Number 2, 2004

PUBLICATIONS RECEIVED

Akinlabi, Akinbiyi & Oluseye Adesola. Proceedings of the 4th World Congress of African Linguistics, New Brunswick 2003. (World Congress of African Linguistics, vol. 4). Cologne: Rüdiger Köppe Verlag. 2004. Pp. x, 575. 3 b/w photos, 3 maps, numerous tables and charts. ISBN 3-89645-338-6. Hardcover. €64.00

This volume presents a selection of papers from the 4th WOCAL, held June 17-22, 2003 at Rutgers University in New Brunswick, NJ. There are 3 papers from plenary sessions: "Cognation Percentages in Benue-Congo: Implications for Internal Classification" by Ben Elugbe & Tayo Bankale; "Why Describe African Languages?" by Larry Hyman; "Segments and Prosodies in Chadic: On Descriptive and Explanatory Adequacy, Historical Reconstructions, and the Status of Lamang-Hdi" by Ekkehard Wolff. The remaining 39 papers are from the general sessions. The section on phonetics and phonology [67-164] covers coda deletion in Yoruba (Adesola), Lokaa vowel harmony (Akinlabi & Iwara), Konni tone (Cahill), pitch realization in Mambila (Connell), phrasing in Sandawe (Dobashi), complex segments (Downing), and tone in Lokaa (Iwara). The 15 papers in the section on morphology, syntax and semantics [165-349] deal with negation in Urhobo (Aziza), object pronouns in Bantu (Beaudoin-Lietz, Nurse & Rose), split word-order / distributed predicate syntax in Doyayo (Elders), subjects in Hausa (Frajzyngier & Munkaila), predicate clefts in Nupe (Kandybowicz), serialization in West Ring (Bantoid) languages (Kiessling), cardinal numbers in Nilo-Saharan (Levaw) and prefixal reduplication in Lusaamia (Marlo). Also included in that section are papers on the conversation function of hava in Swahili (Rudd), locative arguments in Bantu (Rugemalira), logophoricity (Safir), constructions for expressing biting, punching and hitting in Emai (Schaefer & Egbokhare), the discourse relevance of Toposa verb morphology (Schröder), circumfixes in Hone (Storch), and locative predication in Tigrinya (Weldyesus).

The section on computational linguistics [363-390] includes three papers, covering a quantitative study of Yoruba speech surrogacy (Adegbola), a corpus study of zulu *-kazi* (Gauton, de Schyver & Mohlala) and automatic disambiguation of Swahili nouns (Nganga). The section on historical linguistics [391-443] includes a study of comparative pottery terminology in Great Lakes Bantu (Bostoen), reconstruction of velars and labials in certain stages of Benue Congo (Ohiri-Aniche), changes in young people's speech in the Kotoko of Goulfey (Tourneux) and implosives in Mande-Atlantic-Congo (Williamson). The final section [445-571] has 10 papers organized under the rubric sociolinguistics: gender in Basaa (Banoum), language shift in Nama-speaking Herero (Batibo), problems of linguistic under-description in Bi-

lin (Fallon), the Sheng language of Nairobi (Ferrari), the status of Arabic versus African languages in Nyala, Darfuyr (Idris), development of Tanzanian Sign Language (Muzale), the endangered languages Sosan and Ukue (Oyebade & Agoyi) and the Akpes languages (Oyetade), African language contact in Brasil (Petter), and language marginalization (Urua).

Anyawu, Rose-Juliet. Stress and Tone — The African Experience. (Frankfurter Afrikanistische Blätter 15, 2003). Cologne: Rüdiger Köppe Verlag. 2003. Pp. 200. ISBN 3-89645-017-4. Paperback. €34.80.

Ten papers on topics of tone, stress and intonation in African languages are presented in this issue of FAB. "Tone and Accent in the Igbo verb" [5-22] by Anyawu presents a typological argument for an accentual analysis of Igbo tone. "The use of Feet to Account for Binary Tone Spreading: Evidence from Chilungu" by Bickmore [23-47] examines the utility of metrical foot structure in accounting for tone doubling, and shows that both metrical and pure tonal accounts of the process are tenable. "Some Remarks on Tonal Processes at Utterance Boundaries" by Creissels [49-58] looks at intonation-like phonological effects on tone in Baule, Kita Maninka and Tswana, which are conditioned by initial and final pauses. The fourth paper, "Stress, Tone and Focus in Chichewa and Xhosa" by Downing [59-81] discusses accent-like properties of tone in these Bantu languages and argues that tone realization is sensitive to word position in a phrase, whose shape is determined partially by focus. "Tone And Vowel Deletion, Insertion And Syllable Structure" by Frajzyngier [83-98] investigates tone and its interaction with other aspects of segmental phonology in the Chadic languages Mina, Gidar and Lele.

"The Feature of Stress in Nilotic" by Gilley [99-120] argues that Shilluk and Dinka have both contrastive tone and contrastive stress, the latter being realised phonetically, surprisingly, with decreased duration on stressed syllables. In "Tone Mapping in Leggbó" by Hyman & Uhoh [121-139], the authors argue for the significance of independent tone melodies in nouns and verbs, the issue of directional tone mapping, and the relevance of morphology for mapping. The eighth paper, "Tonogenesis in Southern Cushitic (Common West Rift)" by Kießling [141-161] investigates the nature of tone / accent in Southern Cushitic, and it is argued that the system arose historically from the interaction between intonation and a quantity-sensitive stress system. "The Prosodological Structure of Herero" by Möhlig [165-179] presents a range of prosodic properties relevant to Herero including tone, respiratory accent, loudness and intonation. The final paper, "Verbal Tone and Stress in ||Ani (Central Khoisan)" by Voßen [181-191] treats the prosody of this language in terms of 5 tone classes interacting with stress. The volume ends with a review by Anyawu of a festchrift in honor of the late Gabriel Manessy.

Hamidu, Assibi A. Reflexives and Reflexivization in Kiswahili Grammar. (East African Languages and Dialects, vol. 14). Cologne: Rüdiger Köppe Verlag. 2004. Pp. xviii, 527. ISBN 3-89645-028-X. Paperback. €49.50.

Numerous topics relating to transitivity and reflexivization in Kiswahili are investigated in this book, considered from various theoretical perspectives. Chapter 1, "Introduction" [1-8] sets the stage by introducing some of the facts and theoretical approaches to be considered, especially the author's data-oriented perspective which downplays the theoretical "Eurocentric Idiom" and focuses on the author's "Nilotic Idiom" approach. The pronominal root -envewe (as in mnazi wenvewe 'the coconut tree itself') is investigated in Chapter 2, "The reflexive -envewe in Kiswahili grammar" [9-41], where a wide range of the uses of this root are documented. Chapter 3, "Other reflexive nominals of Kiswahili" [42-51] covers nafsi 'soul, self', roho 'soul, life', peke 'alone', which are seen to have semantic connections to reflexives and envewe. The "classic" reflexive prefix -ji- is discussed in Chapter 4, "The reflexive affix of Kiswahili" [52-184], revealing a number of possibly surprising uses (e.g. kimejipolea 'it healed of itself^{*}). There is much discussion of whether -ii- is an object marker, and the author uses data from reflexives to argue that passives are transitive contructions in Kiswahili. The fifth chapter, "Unaccusativity, ergativity, NP Crash and reflexivization in Kiswahili Bantu" [185-298] focuses on verbs such as -enda 'go' in connection with reflexivization, especially on the question of transitivity. Chapter 6, "Some implications of {ii} for transitiveness and class classification" [299-387] further pursues the matter of transitiveness and its relation to the verbal prefix -ji-, especially how -ji- functions as an incorporated NP. Chapter 7, "Reflexivization in Kiswahili and crosslinguistic studies" [388-416] raises the question whether Kiswahili poses a problem for Chomsky's binding principles, and the final chapter "Repeated MPs and other reflexive patterns in Kiswahili syntax" [417-502] reiterates the author's conclusion that -*ji*- is not an object prefix.

Jakobi, Angelika & Joachim Crass. Grammaire de beria (langue saharienne). (Nilo-Saharan Linguistic Analyses and Documentation, vol. 18). Cologne: Rüdiger Köppe Verlag. 2005. Pp. xii, 292; 1 map, numerous tables. ISBN 3-89645-136-7). Paperback. €35.80

The Saharan language Beria, spoken in Chad and Sudan, is described in this book. The introduction [1-19] lays out the objectives and organization of the grammar, and provides historical context for research on the language and for its position within Nilo-Saharan. "Phonologie" [11-33] lists the 17 consonant and 9 vowel phonemes, their allophones, and notes some general phonological processes (ATR harmony; /y, $J/ \rightarrow J$ initially; dd, gg but not bb devoice). This chapter also presents syllable structure (the maximum syllable CVVC has restricted distribution and syllables usually have the form CV(X)), phonotactic restrictions, and the three tone levels of Beria plus its rising and falling contour tones. "Morphophonologie" [35-45] documents the rich verbal morphophonemics, which includes processes of intervocalic lenition, fortition, harmony, sonorant vocalization (/kǎn-ṛ-ɛ́/ \rightarrow [kǎirɛ́], /kór-ṛ-ī/ \rightarrow [kǎírɛ́-ī]), and various segmental deletions. Verbs can be separated into three groups: finite, the copula, and converbs. The structure of the finite verb is treated in the fourth chapter [47-98]. As is common in Saharan languages, there are 3 classes of verbs, reflecting person marking. Ten morphemic slots are available in the verb, but actual forms are often reduced due to morphophonological processes. Inflectional morphology often fuses person and number, aspect, and negation markers. Perfective and imperfective aspects are distinguished in part by tonal distinctions. Forms of the copula are treated in "Copule" [99-106], and the sixth chapter "Nom" [107-120] covers noun morphology. Nominal number is indicated via tone change, and nouns can be organised into 10 groups, according to the tone patterns of tone selected in the singular and the plural. This chapter also presents nominal derivation, including deverbal nouns.

The chapter "Quasi-nominaux" [121-132] presents pronouns, demonstratives, adjectives and numbers. "Syntagme nominal" [133-145] treats NP syntax. Basic NP structure is head-initial, except in "synthetic genitives" where the modifier precedes the head noun. Chapter 9 "Relations grammaticales primaires" [147-156] discusses the syntax of subjects, verbs, and objects. Word order is SOV, and the language exhibits split ergativity. Intransitive verbs split lexically as to whether their subject is encoded the same way as a transitive agent versus a transitive patient. Additionally, the language marks focus with enclitics which reflect grammatical relation, and a focused intransitive subject is marked the same way as a focused patient. Various adverbial phrases (place, manner, time etc.) are presented in "Syntagme adverbial" [157-163]. The penultimate chapter presents the converb [165-176], which is a reduced verb form marked for subject and aspect, but lacking mood inflection, and which (unlike finite verbs) does not come clause-finally. "Propositions subordonnées" [177-184] outlines sentence embedding: relative clauses, temporals, conditionals and others. Two texts [185-192] are provided along with translation and interlinear glosses, and an extensive appendix [193-236] provides verb inflectional pardigms covering various roots. A French-Beria glossary [237-274] gives about 1,000 words.

Kaji, Shigeki. A Runyankore Vocabulary. (Asian and African Lexicon, vol 44). Tokyo: Research Institute for Languages and Cultures of Asia and Africa, Tokyo University of Foreign Studies. 2004. Pp. xxxiii, 603, 1 map. ISBN 4-87297-867-6. Hardcover.

Aspects of the lexicon of Runyankore, a Bantu language of Uganda, are presented here. The main part of the book is a vocabulary of 1136 numbered entries covering 500 pages. The entries are semantically arranged, beginning with 'head' and ending with 'thank you', so that body parts are presented, then animal terms, then fruits and so forth. Each entry may contain a number of words, for example 'flower' contains two apparently synonymous nouns *ekirábyo* and *ekimuri*, as well as the more specific term *obuga:ra* 'papyrus flower'. Thus, the entry 'species of bird' contains 37 specific bird species. Words are glossed in English and Japanese, and numerous examples are provided to illustrate phonological and grammatical properties. Since there are extensive phrasal phonological alternations especially in tone, nouns are given in numerous contexts to document these alternations. Noun class information is also provided; en-

tries for verbs also include nominalizations. Following the main vocabulary are three indices, in English, Japanese and Runyankore, with words followed by the entry number. Preceding the vocabulary is a 20 page grammatical synopsis explaining the phonemes and allophones, phonological processes, the noun class system, and verb derivation and inflection.

Kisseberth, Charles W. & Mohammad Imam Abasheikh. *The Chimwini Lexicon Exemplified*. (Asian and African Lexicon, vol 45). Tokyo: Research Institute for Languages and Cultures of Asia and Africa, Tokyo University of Foreign Studies. 2004. Pp. xxxiv, 533. ISBN 4-87297-868-8. Hardcover.

The Chimwini dialect of Swahili, historically spoken in the town of Brava (Mwiini) in southern Somalia, is the topic of this volume. The book begins by explaining the goals and organization of this lexicon followed by a 23 page morphological and phonological sketch of the language. This is followed by the 530 page Chimwini-English lexicon, organized alphabetically according to the stem of the head word; there are approximately 10,000 headword entries. Each headword entry encompasses some number of related words, each marked for grammatical category and includes relevant derivational and inflectional information. Thus noun entries will indicate noun class and also give the plural, and verb entries will give the perfective stem and numerous derivational forms such as the reciprocal, causative and applied. Words are very often exemplified sententially, and as such, the data of the book will be very useful for the study of syntactic conditions on Chimwini's famous vowel shortening rule. In addition, accent, which is lexically predictable but morphosyntactically complex, is widely marked in the data. Because Chimwini draws its vocabulary from both Bantu and Somali sources, information on source language is often given, including a page reference to one of the major lexical publications on Swahili or Somali.

Newman, Paul. *Klingenheben's Law in Hausa*. (Chadic Linguistics, vol. 2). Cologne: Rüdiger Köppe Verlag. 2004. Pp. 102. ISBN 3-89645-521-4. Paperback. €44.80

This volume treats Klingenheben's Law, a sound change historically affecting syllable-final consonants in Hausa. The book is composed of two parts. The second part [71-102] provides an annotated translation of the original article "Die Silbenuslautgesetze des Hausa" published almost 80 years ago by August Klingenheben, which is the fountainhead for later work on what has come to be known as Klingenheben's Law. Besides making the article more widely accessible to linguists who do not read German, this translation provides a transliteration of Klingenheben's Arabic examples from the original Arabic script. The first part [1-69] provides Professor Newman's critical analysis of the law. After a preliminary explanation of transcription conventions and the phonological basics of Hausa, the book gives an overview of KL, which roughly speaking weakens syllable-final consonants: it is argued that word-medial contexts form the primary context for KL and that application word-finally is a secondary development. This section then treats exceptions to KL: geminate consonants are exceptions (a fact which has

been exploited in various theoretical phonological accounts of the phenomenon), as are ideophones and loanwords. Section 3 [12-39] makes up the bulk of this part, and analyzes the 4 separate changes which make up Klingenheben's Law, carefully scrutinizing examples and contexts of the rules' application. This section covers such oft-neglected questions as the development of the two r's and the question of how sibilants fit into KL. Section 4 covers the changes to syllable-final nasals, and the fifth section discusses the relative chronology of these changes. The sixth and seventh sections lay out the synchronic consequences of these rules, which have resulted in some synchronic alternations especially under reduplication.

Vycichl, Werner. Berberstudien & A Sketch of Siwi Berber (Egypt). (Berber Studies, vol. 10). Cologne: Rüdiger Köppe Verlag. 2005. Pp. xxxvi, 258. numerous tables and charts. ISBN 3-89645-389-0. Paperback. €39.80

The first part of this book is an extract of a manuscript by Werner Vycichl (1909—1999), a leading specialist in Berber historical linguistics, which was intended to provide a comprehensive overview of Berber historical linguistics, phonetics and morphology. This publication presents part of that work in the form of article-like chapters in the first part of the book [1-151] written in German. The first chapter provides information on Berber from ancient onomastics; the second deals with historical phonology; the third presents a study of historical morphology. The fourth chapter gives sketches of Berber languages from Djerba (Qellāla) and Aith 'idhel. The second part [155-258] presents an English sketch of Siwi Berber, the easternmost Berber language. Following a couple dozen pages of background information on ethnography, history of study of the language and accounts of the author's years of personal experiences with the language, this section covers phonetics and etymology [179-195] and grammar [196-255], primarily noun and verb morphology, concluding with a short story.

UPCOMING MEETINGS ON AFRICAN LANGUAGES / LINGUISTICS

2005

Oct 6-8

INTERNATIONAL CONFERENCE ON FOCUS IN AFRICAN LANGUAGES. ZAS, Berlin, Germany. Contact: Focus in African Languages, ZAS, Jaegerstrasse 10-11, D-10117, Berlin, Germany; email <u>fial05@zas.gwz-berlin.de</u>). 1-page abstract submission deadline April 3, 2005.

Oct 9

INFORMATION STRUCTURE IN BANTU LANGUAGES. ZAS, Berlin, Germany. (co-organized with the London-Leiden-Berlin Bantu Grammar network). Contact: email <u>lutz@soas.ac.uk</u>, website <u>http://mercury.soas.ac.uk/users/lm5/bantu_project.htm</u>

Oct 12-14

CONFERENCE ON GUR LANGUAGES. University of Bayreuth, Germany. Theme: "Current questions in Gur Language Research: Between Tone and Text". Contact: conference webpage <u>http://www.uni-bayreuth.de/departments/afrikanistik/tagu/gurcon1.html</u>, email to <u>manfred.vonroncador@uni-bayreuth.de</u>

2006

April 6-9

ANNUAL CONFERENCE ON AFRICAN LINGUISTICS 37. University of Oregon. Abstract deadline: December 7, 2005. Contact: conference webpage <u>http://www.uoregon.edu/~acal2006</u>.

April 20-22

BANTU GRAMMAR: DESCRIPTION AND THEORY. SOAS, University of London. Abstract deadline: Dec. 19, 2005. 1 page A4 abstract 2 copies to Lutz Marten, Dept. of Africa, SOAS, Thornhaugh Street, Russell Square, London WC1H 0XG, England PDF or emailed to <u>BantuConference@soas.ac.uk</u>). Further information on the conference and abstract requirements: http://mercury.soas.ac.uk/users/lm5/BantuConference.htm.

July 10-12

WORLD CONGRESS ON AFRICAN LINGUISTICS, 5TH. Addis Ababa University, Addis Ababa, Ethiopia. Abstract acceptance dates: July 10-Dec 10, 2005. 1 page 12 pt. abstract as PDF or postscript attachment emailed to <u>afriling@dling.aau.edu.et</u>) or Word document on 3.5 in. floppy mailed to WOCAL 5, Department of Linguistics, Addis Ababa University, P.O.Box 1176, Addis Ababa, Ethiopia. Further information at the conference website <u>http://www.aau.edu.et/faculties/linguistics/wocal.htm</u>.

Note on Forthcoming Supplement

Volume 34 will include a supplement, *West African Linguistics: Papers in Honor of Russell G. Schuh* guest-edited by Paul Newman and Larry Hyman, approximately 250 pp. The volume will be available for \$10 to all subscribers of volume 34. The non-subscriber price will be \$20 plus shipping.
